ZnO nanowires and their application in field-effect transistors

by

Conor Patrick Burke-Govey

A thesis submitted to the Victoria University of Wellington in fulfilment of the requirements for the degree of Doctor of Philosophy in Physics.

Victoria University of Wellington 2018

Abstract

ZnO nanowires have shown great promise as a semiconducting material for a variety of different electronic applications at the nanoscale, and can be easily synthesised at low temperatures using the hydrothermal growth method. However, efforts to reliably produce field-effect transistors (FETs) using ZnO nanowires have been hampered by excessive charge carriers, requiring high temperature annealing (\geq 400°C) at the expense of the low-temperature synthesis before field dependence is achieved. This thesis presents hydrothermally synthesised ZnO nanowires which can effectively be used as FETs in dry and liquid environments without requiring any annealing or post-growth processing.

The role of polyethylenimine (PEI) in the hydrothermal growth of vertical ZnO nanowires is thoroughly investigated. PEI is a polymer used to increase the aspect ratio of ZnO nanowires, but the molecular weight of the polymer and interactions with other growth precursors are often overlooked. Using 4 mM of PEI($M_W = 1300$ g/mol) results in hierarchical nanowires, consisting of large primary nanowires which abruptly terminate in thinner secondary nanowires. The secondary nanowires, with lengths of up to 10 μ m and diameters below 50 nm, are synthesised during a PEI-mediated secondary growth phase, where Zn-PEI complexes continue to provide Zn²⁺ ions after the bulk of the precursors have been exhausted.

The PEI-mediated synthesis of hierarchical nanowires is used to fabricate FETs by laterally growing intersecting networks of nanowires from spaced pairs of ZnO/Ti films, which have been patterned on SiO₂/Si device substrates. All of these FETs show marked field dependence between $V_G = -10$ V to 10 V, despite being used without annealing. Typical on-off ratios are between $10^3 - 10^5$, with threshold voltages between -7.5 V to 5 V. This is a significant result, as the majority of ZnO nanowire FETs reported in the literature require high temperature annealing. Persistent photoconductivity measurements indicate that surface states on the nanowires contribute to the intrinsic field dependence of the devices.

Hierarchical nanowires are also synthesised by modular primary and secondary hydrothermal growths. FETs fabricated using these hierarchical nanowires show less field dependence than PEI-mediated hierarchical nanowires, with limited functionality when used in air. The best FET measured in air operates with an on-off ratio of 10^4 and a threshold voltage of ≈ 0 V. Devices which are field-independent in air can be reliably gated by measuring the FETs in a wet environment, using de-ionised water as a dielectric. A back-gated wet FET operates with an on-off ratio of 10^5 and a threshold voltage of ≈ 8 V. Top-gated wet FETs operate with on-off ratios within 10^3 - 10^4 , and threshold voltages within 0.4 - 0.9 V. These devices also have significantly low subthreshold swings, on the order of 80 mV/decade.

FETs are fabricated by contacting individual ZnO nanowires using electron-beam lithography, although only one vertical ZnO nanowire shows field dependence, with an on-off ratio of 10⁴ and a threshold voltage of -7 V. A PEI-mediated hierarchical nanowire is also contacted and shows field dependence, with an on-off ratio of 10² and a threshold voltage of -6 V. The poor on-off ratio is caused by high leakage currents of the device. The contacted nanowires undergo dissolution over time, disappearing from the substrates after 8 months, and also exhibit a conducting-to-insulating transition over 48 hours. This transition can be temporarily reversed by exposure to an electron beam. Neither of these effects are reported in the literature, and their causes are speculated on.

Finally, the thesis concludes with proposals for future work to further the advances made here.

Dedicated in memory of my mum, Jacquie Burke 30/01/1954 - 03/02/2014

For some we loved, the loveliest and the best that from his vintage rolling Time hath pressed have drunk their cup a round or two before and one by one crept silently to rest

> — Omar Khayyam, *The Rubaiyat of Omar Khayyam* (circa 1120 CE)

Acknowledgments

A PhD is not a walk in the park, and I was helped by many people at many stages through this project. I am long overdue to acknowledge them.

First and foremost, I am grateful to my primary supervisor, Natalie Plank, for making this entire project possible. She gave me the opportunity to do a PhD in nanoelectronics, which was a subject I had never even considered before joining the cleanroom at Victoria. I had an enjoyable and productive time during my PhD, which would not have been the case if it weren't for Natalie's supervision. I appreciate all of the insightful discussion about physics we had, and about life in general. Thank you for introducing me to this exciting area of research, and for all of your help and understanding throughout the project.

I am also very grateful to my second supervisor, Joe Trodahl, for his continued mentorship over the course of my entire academic career to date. If I hadn't worked under his tutelage during that formative summer between second and third year of undergrad, I likely wouldn't be involved in physics research today. Joe wields the critic's knife with an expertise and discernment honed over half a century of use, paring away the ugly and superfluous parts of an idea so that only the true and beautiful parts remain. He quickly taught me (intentionally or otherwise) to separate my ego from my thoughts, which is a skill I will use for the rest of my life. Thank you for everything, Joe.

I'd like to thank my previous and current labmates over the years. In particular, I would like to thank Hanyue (Hannah) Zheng, who was there to help me from the very start of my PhD, right until the very end. We shared late nights (and the occasional weekend) in the lab, enlightening discussions about physics and our experiments, and just life in general. You showed me the ropes of the cleanroom as a colleague and as a friend. I'd also like to thank Cameron Wood, who helped me get my feet on the ground and gave me a crash course in the hydrothermal method all those years ago. Thanks also goes to Leo Browning, Mohsen Maddah, and Selvan Murugathas. I extend my thanks to all of the past and present students and academics

in the UHV group, and the others who attended our group meetings. I always enjoyed our presentations and the discussions which followed. Special mention must go to Ben Ruck, Franck Natali, and Bart Ludbrook, with whom I had many fruitful discussions over a pint or three.

There are many people who worked behind the scenes to make this thesis possible. In particular, I'd like to thank David Flynn, for all of his help with SEM and for the odd (and desperately needed) coffee. I'd like to thank Alan Rennie and Nick and Manu, for keeping the lights on and the water running. Thanks must also go to Sarah Dadley, who has been invaluable in more ways than one!

I've had the pleasure of working with several collaborators over the course of my PhD. I'd like to thank Jerome Majimel and Uli Castanet for helping us with TEM measurements over the course of this project. I'd also like to thank Alex Nau for his internship, which contributed to the lateral hierarchical nanowire work. Thanks also goes to Martin Allen and Max Lynam for their help with EBL down at UC, along with Gary Turner and Helen Devereux. The EBL work at UNSW couldn't have been done without Rifat Ullah's help, to whom I am very grateful. Thanks for making my stay in Sydney so enjoyable. I'd also like to sincerely thank Adam Micolich for making the EBL work at UNSW possible, his input on our experiments, his expertise, his critical evaluation of our manuscripts, and also for being a bloody great bloke all-round. Ben Ruck helped read many draft manuscripts and was always keen to talk about experiments, which I'm very grateful for. He also directly contributed to the persistent photoconductivity measurements, alongside Harry Warring.

Many of my friends and fellow PhD students made my PhD an unforgettable experience, both on the clock and in our free time. I'd like to thank Brendan Darby, James McNulty, Eva Anton, Harry Warring, Stephanie Droste, Julia Schacht, Leah Graham, Peter Hauer, Ihab Sinno, Hamish Colenso, Leo Browning, and Mohsen Maddah, amongst the many others who all played a part. I will remember the moments we shared long after the minutiae of the project itself. A special mention goes to Felix Barber, whose friendship was never limited by the distance between us.

My sincere gratitude goes out to my partner, Hyojin Jung. Thank you for all of your love, your support, and your understanding during my PhD. You endured long, lonely nights and weekends when I was in the lab or busy otherwise. During the troglodyte phase of the writing process, you put up with my thesis beard and my scholarly haircut (or lack thereof.) You truly have the patience of a saint.

Finally, I would like to thank my family, close and extended. I would like to thank my sister and brother, Caitlin and Finn, and my mum and my dad, Jacquie and Ian. This project could never have started without your continued love and support, let alone see completion. In particular, I would like to thank my mum for her unwavering support and belief in me, her grace, and her optimism in the face of adversity. This thesis is dedicated to her memory.

Contents

| 1 | Intr | oductio | on and a second s | 1 |
|---|------|---------|---|----|
| | 1.1 | Motiv | ration | 1 |
| | 1.2 | Funda | amental properties of ZnO | 4 |
| | | 1.2.1 | Electronic properties | 5 |
| | 1.3 | Thesis | soutline | 10 |
| 2 | Hyd | lrother | mal growth | 13 |
| | 2.1 | Introd | luction | 13 |
| | 2.2 | Vapou | ır phase transport | 13 |
| | 2.3 | Hydro | othermal methods | 14 |
| | | 2.3.1 | Introduction | 14 |
| | | 2.3.2 | Hydrothermal synthesis equipment and method | 15 |
| | 2.4 | Chem | istry of the hydrothermal growth | 16 |
| | | 2.4.1 | Role of HMT | 17 |
| | | 2.4.2 | Polyethylenimine as a growth additive | 20 |
| | 2.5 | Hydro | othermal growth of ZnO nanowires | 24 |
| | | 2.5.1 | Homogeneous growth | 24 |
| | | 2.5.2 | Heterogeneous growth | 27 |
| | | 2.5.3 | Growth from seeded substrates | 27 |
| | 2.6 | Concl | usion | 30 |
| 3 | Fiel | d-effec | t transistors | 31 |
| | 3.1 | Introd | luction | 31 |
| | 3.2 | FET B | ackground | 31 |
| | | 3.2.1 | Mode of operation | 31 |
| | | 3.2.2 | Transfer characteristics and device parameters | 32 |
| | | 3.2.3 | Threshold voltage - V_{Th} | 33 |
| | | 3.2.4 | Transconductance - g_m | 34 |
| | | 3.2.5 | Subthreshold swing - SS | 35 |
| | | 3.2.6 | Mobility - μ_{eff} | 36 |
| | 3.3 | Hydro | othermal ZnO nanowire FETs in the literature | 37 |

| | | 3.3.1 | FETs fabricated with lateral NW arrays | 37 | | | |
|---|------|--|--|-----|--|--|--|
| | | 3.3.2 | FETs fabricated with individual ZnO nanowires | 40 | | | |
| | 3.4 | ZnO r | nanowire doping and annealing | 50 | | | |
| | | 3.4.1 | n-type ZnO nanowires | 50 | | | |
| | 3.5 | Hyste | resis and passivation in ZnO nanowire FETs | 53 | | | |
| | 3.6 | Concl | usion | 55 | | | |
| 4 | Exp | erimen | tal details | 57 | | | |
| | 4.1 | Introd | luction | 57 | | | |
| | 4.2 | Fabric | cation of substrates for lateral nanowire growth | 57 | | | |
| | | 4.2.1 | Photolithography | 58 | | | |
| | | 4.2.2 | Seed layer/electrode deposition | 60 | | | |
| | 4.3 | Fabric | cation of substrates for vertical nanowire growth | 62 | | | |
| | 4.4 | Electr | ical measurements | 62 | | | |
| | 4.5 | Scann | ing electron microscopy | 64 | | | |
| | 4.6 | Transi | mission electron microscopy | 64 | | | |
| 5 | Hyd | lrother | mal growth of vertical ZnO nanowires | 65 | | | |
| | 5.1 | Introd | luction | 65 | | | |
| | 5.2 | Exper | imental details | 66 | | | |
| | 5.3 | Effect | s of PEI molecular weight and concentration | 67 | | | |
| | | 5.3.1 | PEI(800) | 69 | | | |
| | | 5.3.2 | PEI(1300) | 73 | | | |
| | | 5.3.3 | PEI(2000) | 76 | | | |
| | 5.4 | Influe | nce of PEI on morphology | 79 | | | |
| | | 5.4.1 | ZnO thin film growth mechanism | 79 | | | |
| | | 5.4.2 | Hierarchical nanowire growth mechanism | 82 | | | |
| | 5.5 | Comp | parison of homogeneous nanowires | 86 | | | |
| | | 5.5.1 | Summary | 89 | | | |
| | 5.6 | Concl | usion | 89 | | | |
| 6 | PEI- | PEI-mediated growth of hierarchical ZnO nanowire FETs 91 | | | | | |
| | 6.1 | Motiv | ation and background | 92 | | | |
| | 6.2 | Hydro | othermal growth and hierarchical nanowire FET fabrication \ldots . | 93 | | | |
| | | 6.2.1 | Hydrothermal growth | 93 | | | |
| | | 6.2.2 | ZnO nanowire synthesis results | 94 | | | |
| | | 6.2.3 | Hierarchical nanowire device fabrication | 97 | | | |
| | | 6.2.4 | Hierarchical nanowire growth mechanism | 100 | | | |
| | | 6.2.5 | Hierarchical nanowire crystal structure | 101 | | | |
| | 6.3 | Electr | ical measurements | 102 | | | |

| | | 6.3.1 | PEI(2000) devices | 104 |
|---|------|---------|--|-----|
| | | 6.3.2 | PEI(1300) devices | 106 |
| | | 6.3.3 | PEI(800) devices | 109 |
| | | 6.3.4 | Discussion | 111 |
| | 6.4 | Persis | tent photoconductivity measurements | 111 |
| | | 6.4.1 | Transfer characteristics during persistent photoconductivity | 116 |
| | | 6.4.2 | Persistent photoconductivity of primary nanowire FETs | 118 |
| | 6.5 | Concl | usion | 119 |
| 7 | Moo | dular g | rowth of lateral hierarchical ZnO nanowire FETs | 121 |
| | 7.1 | Introd | luction | 121 |
| | 7.2 | Hiera | rchical nanowire synthesis | 122 |
| | | 7.2.1 | Theory and motivation | 122 |
| | | 7.2.2 | Experimental details | 123 |
| | | 7.2.3 | Initial results | 125 |
| | 7.3 | Optin | nisation for reproducibility and device fabrication | 126 |
| | | 7.3.1 | Hydrothermal growth details | 129 |
| | | 7.3.2 | Effect of preheat on the primary growth | 130 |
| | | 7.3.3 | Results and discussion | 133 |
| | 7.4 | Electr | ical measurements | 137 |
| | | 7.4.1 | Unpreheated primary growths and leakage through debris | 137 |
| | | 7.4.2 | Hierarchical nanowire device measurements | 138 |
| | | 7.4.3 | Liquid gating measurements | 141 |
| | | 7.4.4 | Context within literature | 149 |
| | 7.5 | Concl | usion | 150 |
| 8 | Indi | ividual | ly contacted ZnO nanowires | 153 |
| | 8.1 | Introd | luction | 153 |
| | 8.2 | Motiv | ation and background | 154 |
| | | 8.2.1 | Nanowire-metal contacts and Schottky barrier modelling | 155 |
| | | 8.2.2 | Batchwise fabrication of devices | 158 |
| | 8.3 | UNSV | V1 | 158 |
| | | 8.3.1 | Single nanowire device fabrication and yields | 158 |
| | | 8.3.2 | Electrical measurements | 161 |
| | 8.4 | UC1 . | | 165 |
| | | 8.4.1 | Single nanowire device fabrication and yields | 165 |
| | | 8.4.2 | Electrical measurements | 169 |
| | 8.5 | UC2 . | | 175 |
| | | 8.5.1 | Single nanowire device fabrication and yields | 175 |
| | | 8.5.2 | Electrical measurements | 178 |

| | | 8.5.3 | Conductivity decay and electron beam exposure | 183 |
|---|------|---------|---|-----|
| | 8.6 | Schott | ky barrier heights and ideality factors | 187 |
| | | 8.6.1 | Relation between contacts on a single nanowire | 187 |
| | | 8.6.2 | Relation between the SBH, ideality factor, and nanowire diameter | 187 |
| | | 8.6.3 | Nanowire growth conditions and Schottky contacts | 188 |
| | 8.7 | Materi | al degradation | 191 |
| | 8.8 | Summ | ary | 195 |
| 9 | Con | clusion | and future work | 197 |
| | 9.1 | Conclu | asion | 197 |
| | | 9.1.1 | Hydrothermal synthesis of vertical ZnO nanowires $\ldots \ldots \ldots$ | 197 |
| | | 9.1.2 | PEI-mediated growth of hierarchical ZnO nanowire FETs | 198 |
| | | 9.1.3 | Modular growth of hierarchical ZnO nanowire FETs | 198 |
| | | 9.1.4 | Individually contacted ZnO nanowires | 199 |
| | 9.2 | Future | work | 200 |
| | | 9.2.1 | PEI-mediated hierarchical nanowire growth over time | 200 |
| | | 9.2.2 | Hierarchical nanowire FETs as device platforms for biosensors . | 201 |
| | | 9.2.3 | ZnO nanowire sensitivity to electron beam exposure $\ldots \ldots$ | 201 |
| | | 9.2.4 | ZnO nanowire dissolution over time | 202 |
| A | Elec | tron be | am lithography procedure | 205 |
| | A.1 | Electro | on beam lithography | 205 |
| | | A.1.1 | Device substrate fabrication | 206 |
| | | A.1.2 | Nanowire transfer and imaging | 208 |
| | | A.1.3 | Electron beam lithography | 209 |
| | | A.1.4 | Development and metal deposition | 218 |

List of Figures

| 1.1 | Eight-year forecasts of the transparent display market | 2 |
|------|--|----|
| 1.2 | A graphical representation of future transistor development outlined by | |
| | the International Technology Roadmap for Semiconductors | 3 |
| 1.3 | The hexagonal wurtzite crystal structure of ZnO | 4 |
| 1.4 | Relationship between charge carrier concentration and temperature for | |
| | bulk ZnO | 7 |
| 1.5 | Relationship between electron mobility and temperature for bulk ZnO $$. | 8 |
| 1.6 | Monte Carlo simulations of electron drift velocity in ZnO and GaN \ldots | 9 |
| 2.1 | SEM images of hydrothermally synthesised ZnO nanowires published | |
| | by Andre-Verges and Vayssieres | 15 |
| 2.2 | Schematic of hydrothermal growth setup | 16 |
| 2.3 | Schematic showing the capping of polar and non-polar faces of ZnO | |
| | nanowires | 21 |
| 2.4 | A schematic of a branched PEI molecule | 21 |
| 2.5 | The Mannich reaction involving primary amines | 22 |
| 2.6 | Effects of PEI on ZnO nanowire growth chemistry | 23 |
| 2.7 | Images of bottles containing PEI(800) which have been left to preheat for | |
| | one hour | 26 |
| 2.8 | Vertical growth of ZnO nanowires from Au and ZnO thin films | 28 |
| 2.9 | Schematic of lateral ZnO nanowire growth from capped ZnO seeds | 29 |
| 2.10 | SEM images of laterally-grown ZnO nanowires depicting growth in a | |
| | 180° arc | 30 |
| 3.1 | Schematic of band bending via the field effect | 32 |
| 3.2 | Transfer characteristics of a standard FET | 33 |
| 3.3 | Threshold voltage extraction of a standard FET | 34 |
| 3.4 | Transconductance of a standard FET | 35 |
| 3.5 | Subthreshold swing of a standard FET | 36 |
| 3.6 | IV measurements of ZnO nanowire FETs - Park et al | 38 |
| 3.7 | IV measurements of ZnO nanowire FETs - Kalblein et al. | 41 |

| 3.8 | As-fabricated and annealed measurements of ZnO nanowire FETs - | |
|------|--|----|
| | Opoku et al. | 43 |
| 3.9 | Annealed IV measurements of ZnO nanowire FETs - Opoku et al | 44 |
| 3.10 | IV measurements of ZnO nanowire FETs - Florica et al | 46 |
| 3.11 | IV measurements of passivated ZnO nanowire FETs - Florica et al | 47 |
| 3.12 | Hydrogen diffusion and stability in ZnO nanowires | 51 |
| 3.13 | Transfer characteristics of ZnO nanowires annealed at different temper- | |
| | atures | 52 |
| 4.1 | The photolithographic process | 59 |
| 4.2 | Photolithographic mask used to fabricate substrates for lateral nanowire | |
| | growth | 60 |
| 4.3 | Optical and SEM images of lateral substrates for device fabrication | 61 |
| 5.1 | Cross-sectional and top-down SEM images of vertical ZnO nanowires | |
| | grown without PEI | 68 |
| 5.2 | Cross-sectional and top-down SEM images of vertical nanowires grown | |
| | with 8 mM and 6 mM of PEI(800) | 70 |
| 5.3 | Cross-sectional and top-down SEM images of vertical nanowires grown | |
| | with 4 mM and 2 mM of PEI(800) | 71 |
| 5.4 | Top-down SEM images of additional vertical nanowires grown with 4 | |
| | mM of PEI(800) | 72 |
| 5.5 | Cross-sectional and top-down SEM images of vertical nanowires grown | |
| | with 8 mM and 6 mM of PEI(1300) | 74 |
| 5.6 | Cross-sectional and top-down SEM images of vertical nanowires grown | |
| | with 4 mM and 2 mM of PEI(1300) | 75 |
| 5.7 | Cross-sectional and top-down SEM images of vertical nanowires grown | |
| | with 8 mM and 6 mM of PEI(2000) | 76 |
| 5.8 | Cross-sectional and top-down SEM images of vertical nanowires grown | |
| | with 4 mM and 2 mM of PEI(2000) | 78 |
| 5.9 | Schematic depiction of ZnO thin film growth | 80 |
| 5.10 | Partial film growth caused by 3 mM of PEI(2000) | 81 |
| 5.11 | Schematic of the PEI-mediated growth mechanism of vertical hierarchi- | |
| | cal nanowires | 83 |
| 5.12 | SEM images of localised hierarchical nanowire growth mediated | |
| | through PEI agglomerates | 85 |
| 5.13 | Comparison of the lengths of homogeneous vertical nanowires grown | |
| | with different concentrations of PEI | 87 |
| 5.14 | Comparison of the diameters and aspect ratios of homogeneous vertical | |
| | nanowires grown with different concentrations of PEI | 88 |

| 6.1 | Schematic of a lateral nanowire FET | 93 |
|------|--|-----|
| 6.2 | SEM images of hierarchical and regular lateral ZnO nanowires | 95 |
| 6.3 | Optical image demonstrating growth suppression of lateral ZnO | |
| | nanowires at 4 mM of PEI(800) | 96 |
| 6.4 | Schematic and SEM image of a PEI-mediated hierarchical nanowire FET | |
| | grown with 4 mM of PEI(1300) | 98 |
| 6.5 | Optical image of a PEI-mediated hierarchical nanowire FET grown with | |
| | 8 mM of PEI(800) | 99 |
| 6.6 | SEM images of PEI-mediated hierarchical nanowires grown with | |
| | PEI(2000), PEI(1300), and PEI(800) | 100 |
| 6.7 | Schematic of the PEI-mediated hierarchical nanowire growth mechanism | 101 |
| 6.8 | TEM images of primary and secondary nanowires from PEI-mediated | |
| | lateral hierarchical nanowires | 102 |
| 6.9 | TEM images of the primary/secondary nanowire interface of PEI- | |
| | mediated lateral hierarchical nanowires | 103 |
| 6.10 | Transfer characteristics of secondary- and primary-nanowire FETs fabri- | |
| | cated using 3 mM of PEI(2000) | 105 |
| 6.11 | Transfer characteristics of PEI-mediated hierarchical nanowire FETs syn- | |
| | thesised using 3 mM of PEI(2000) | 106 |
| 6.12 | Transfer characteristics of PEI-mediated hierarchical nanowire FETs syn- | |
| | thesised using 4 mM of PEI(1300) | 107 |
| 6.13 | Transfer characteristics of PEI-mediated hierarchical nanowire FETs syn- | |
| | thesised using 8 mM of PEI(800) | 109 |
| 6.14 | Transfer characteristics of a PEI(1300) hierarchical FET prior to evacuat- | |
| | ing the cryostat for persistent photoconductivity measurements | 113 |
| 6.15 | Transfer characteristics of a PEI(1300) hierarchical FET during the | |
| | pumping of the cryostat for persistent photoconductivity measurements | 114 |
| 6.16 | Persistent photoconductivity measurements of a hierarchical nanowire | |
| | FET fabricated using PEI(1300) | 115 |
| 6.17 | Transfer characteristics of a hierarchical nanowire FET fabricated using | |
| | PEI(1300) during persistent photoconductivity | 117 |
| 6.18 | Persistent photoconductivity measurements of a primary nanowire FET | 118 |
| 7.1 | Schematic showing modular growth of hierarchical lateral nanowires | 124 |
| 7.2 | Top-down SEM images of modular lateral hierarchical nanowires | 125 |
| 7.3 | Cross-sectional SEM images of modular lateral hierarchical nanowires . | 127 |
| 7.4 | TEM measurements of primary and secondary nanowires from modular | |
| | hierarchical nanowires | 128 |

| 7.5 | Flowchart showing the optimisation process of the modular hierarchical | |
|------|---|-----|
| | nanowire growth | 130 |
| 7.6 | Optical images showing surface-nucleated debris after primary | |
| | nanowire growth | 132 |
| 7.7 | Top-down SEM images of optimised modular hierarchical nanowires | 136 |
| 7.8 | Leakage currents for a modular hierarchical FET fabricated without a | |
| | preheated primary growth | 138 |
| 7.9 | Transfer characteristics of a field-dependent modular hierarchical | |
| | nanowire FET | 139 |
| 7.10 | Diode measurements of a modular hierarchical nanowire FET fabricated | |
| | using a 60 minute preheated primary growth | 140 |
| 7.11 | Transfer characteristics of a field-independent modular hierarchical | |
| | nanowire FET fabricated using a 60 minute preheated primary growth . | 141 |
| 7.12 | Schematic of a liquid gated hierarchical nanowire FET | 143 |
| 7.13 | Output characteristics of a field-independent modular hierarchical | |
| | nanowire FET | 144 |
| 7.14 | Transfer characteristics of a liquid top-gated modular hierarchical | |
| | nanowire FET | 145 |
| 7.15 | Transfer characteristics of a liquid back-gated modular hierarchical | |
| | nanowire FET | 146 |
| 7.16 | Wet back-gated transfer characteristics of the same FET | 147 |
| 7.17 | Further wet back-gated transfer characteristics of a modular hierarchical | |
| | nanowire FET | 148 |
| 7.18 | Output characteristics of the same wet back-gated FET | 149 |
| 8.1 | Schematic showing the conduction pathway for lateral FETs | 154 |
| 8.2 | Depiction of Schottky barrier formation at the metal/semiconductor in- | |
| | terface | 156 |
| 8.3 | SEM images of individually-contacted nanowires from UNSW S1 | 160 |
| 8.4 | Diode measurements of individually contacted nanowires from UNSW1 | |
| | S1 | 162 |
| 8.5 | Back-to-back Schottky barrier fits of UNSW1 S1 | 163 |
| 8.6 | SEM images of individually contacted nanowires from UC1 | 168 |
| 8.7 | Electrical measurements of UC1 S1 and S2 | 170 |
| 8.8 | Back-to-back Schottky barrier fits of UC1 S1 | 172 |
| 8.9 | Electrical measurements of UC1 S3 | 174 |
| 8.10 | SEM measurements of UC1 S1 and S2 | 177 |
| 8.11 | Diode measurements of UC2 S2 and S3 | 179 |
| 8.12 | Back-to-back Schottky barrier fits of UC2 | 180 |

| Transfer characteristics of UC2 - S2 | 181 |
|--|---|
| Transfer characteristics of UC2 - S1 | 182 |
| Electrical measurements of UC2 - S2 | 184 |
| Electrical measurements of UC2 - S4 | 186 |
| Schottky barrier heights and ideality factors of individually contacted | |
| nanowires | 188 |
| Schottky barrier heights and ideality factors of individually contacted | |
| nanowires plotted by radius | 188 |
| Schottky barrier heights and ideality factors of individually contacted | |
| nanowires plotted by nanowire growth recipe | 190 |
| SEM images showing disappearance of nanowires from UC1 during liftoff | 191 |
| SEM images showing disappearance of nanowires from UC2 - S2 after 8 | |
| months | 192 |
| EDS maps of dissolved UC2 nanowires and residue on substrate | 194 |
| Schematic of the mask used to fabricate EBL chips | 207 |
| SEM image of nanowires to be located during EBL | 210 |
| Schematic of the PMMA bilayer used in EBL | 211 |
| Process used to define the $(u,v) = (0,0)$ origin during EBL | 212 |
| Schematic showing angle correction during EBL | 213 |
| Schematic showing the location of the first writefield from the (u,v) ori- | |
| gin during EBL | 214 |
| Comparison between SEM pre-image and slowscan during EBL | 215 |
| Schematic depicting the use of the GDS editor | 216 |
| Schematic showing overlapping patterns defined using the GDS editor . | 217 |
| Optical image of EBL sample post-development | 219 |
| | Transfer characteristics of UC2 - S2 Transfer characteristics of UC2 - S1 Electrical measurements of UC2 - S2 Electrical measurements of UC2 - S4 Schottky barrier heights and ideality factors of individually contacted nanowires Schottky barrier heights and ideality factors of individually contacted nanowires plotted by radius Schottky barrier heights and ideality factors of individually contacted nanowires plotted by radius Schottky barrier heights and ideality factors of individually contacted nanowires plotted by nanowire growth recipe SEM images showing disappearance of nanowires from UC1 during liftoff SEM images showing disappearance of nanowires from UC2 - S2 after 8 months Schematic of the mask used to fabricate EBL chips Schematic of the mask used to fabricate EBL chips Schematic of the PMMA bilayer used in EBL Schematic of the PMMA bilayer used in EBL Schematic showing angle correction during EBL Schematic showing the location of the first writefield from the (u,v) origin during EBL Comparison between SEM pre-image and slowscan during EBL Schematic depicting the use of the GDS editor Schematic showing overlapping patterns defined using the GDS editor Schematic showing overlapping patterns defined using the GDS editor |

Chapter 1

Introduction

1.1 Motivation

The transistor was first conceptualised in the first quarter of the twentieth century, and was eventually realised on the 23^{rd} of December, 1947, at Bell Laboratories in the United States of America [1]. The importance of this invention was realised at the time, with John Bardeen, Walter Brattain, and William Shockley being awarded the 1956 Nobel Prize in Physics for their contributions to the project. Despite these accolades, the impact of the transistor must have been far greater than any of its inventors or contemporaries could have conceived of. What followed over the ensuing decades was an enormous paradigm shift called the Digital Revolution, which has arguably had the largest technological impact on humanity since the Industrial Revolution of the nineteenth century. Digital electronics have now become ubiquitous, available to almost all people of every society, culture, and socio-economic background.

Further development of the transistor lead to devices using different operating principles and semiconducting materials. Silicon emerged as the most common semiconductor used for transistor applications, and it now underpins the majority of modern electronics, and will continue to do so for some time. However, electronics are gradually trending towards applications which silicon is no longer suitable for, such as transparent electronics [2], blue/ultraviolet photonics [3], and high temperature applications [4]. Lucrative markets are already becoming established as these niche electronics mature from prototypes to consumer electronics. As an example, transparent LCD/OLED displays are predicted to be worth billions of dollars by the end of the decade [5] (shown in Figure 1.1.) These demands have renewed an interest in wide band-gap semiconductors over the last several decades, some of which are now mature technologies with commercial applications (such as gallium nitride.)



Eight-year Forecasts of Transparent Display by Frontplane Technology (\$ Million)

Figure 1.1: Eight-year forecasts of the transparent display markets. Figure reproduced from [5].

Silicon transistors are also facing pressure from the continual demands of scaling and miniaturising, as per Moore's law. The International Technology Roadmap for Semiconductors predicts that the current field-effect transistor (FET) architecture, the FinFET, will be reduced to its minimum size some time during 2020 [6]. New device architectures will then be required to fit more transistors on a single chip. The forefront of these new architectures are based around nanowires, with lateral all-around-gate nanowires and vertical all-around-gate nanowires predicted to find applications in commercial electronics from 2019 to 2030. This roadmap is graphically summarised in Figure 1.2(a), while Figure 1.2(b) shows specific ways of stacking nanowires to achieve more complex and high-density transistors.

These factors strongly motivate research into materials with wide bandgaps, high thermal stability, good optoelectronic properties, and ease of nanostructure synthesis. ZnO in particular has risen to the forefront of wide-bandgap semiconductor research over the past two decades due to its myriad of attractive properties. In addition to its intrinsic physical properties, ZnO nanostructures can be readily synthesized through a variety of means [9]. Perhaps the most promising synthesis route is the hydrothermal method, which is a low-cost, low-temperature wet chemical method for nanowire synthesis, with great potential for scaling compared to other contemporary nanowire synthesis methods [10]. Despite the potential of the synthesis route, hydrothermal ZnO nanowires are rarely used in field-effect transistor applications, primarily because of unintentionally high charge-carrier concentrations [11].



Figure 1.2: (a) A graphical representation of transistor development outlined by the International Technology Roadmap for Semiconductors [6]. Reproduced from [7]. (b) A series of different transistors of increasing complexity based on lateral and vertical nanowires. Adapted from [8].



Figure 1.3: A diagram showing the hexagonal wurtzite crystal structure of ZnO. Figure adapted from [14].

This thesis aims to explore the hydrothermal synthesis of ZnO nanowires and their applications in field-effect transistors, and solve some of the problems which prohibit their greater implementation. Attention is paid to single-nanowire FETs and FETs consisting of lateral arrays of nanowires, with a particular emphasis on scalability of fabrication. These results hopefully shed some light on the feasibility of hydrothermal ZnO nanowires in FET applications, and also offer a platform for further integration into sophisticated devices, such as biosensors.

1.2 Fundamental properties of ZnO

ZnO is a Group II-VI semiconductor which occurs in the hexagonal wurtzite crystal structure (shown in Figure 1.3), although both rock-salt and zincblende structures are possible if grown under high pressure [12]. It has a very wide direct bandgap of approximately 3.37 eV [13], which corresponds to a photon wavelength of \approx 368 nm, making it transparent to the visible spectrum and an excellent material for UV detection. The wide band gap also reduces thermal noise for high temperature applications. This is complemented by the high cohesive energy (7.52 eV) of the Zn-O bonds, which make the crystal structure extremely stable over high temperatures [12].

As with most II-VI semiconductors, ZnO forms mainly through ionic bonds. The O-plane and Zn-planes in the ZnO crystal structure are effectively ionised as O^{2-} and

Zn²⁺, giving rise to a permanent dipole moment [15]. This also gives ZnO a very strong piezoelectricity, which is useful for many different micro-electromechanical devices and applications, such as ambient energy scavenging and wearable electronics [16]. Piezoelectric effects can also be combined with FET applications, which gives rise to the piezotronic effect. This has opened up a new class of ZnO NW devices over the past decade, with future applications in a variety of sensors and other similar devices [17].

The large bandgap of ZnO is coupled with an extremely high free exciton binding energy of 60 meV [12], which is roughly 2.4 times higher than the energy $(k_BT \approx 25 \text{ eV})$ of room temperature thermal phonons. Free excitons are therefore theoretically stable at room temperature, which has important implications for devices such as solar cells and lasers [12]. These characteristics make ZnO a very attractive material for UV and optoelectronic applications.

ZnO also readily forms into a wide variety of nanostructures, such as nanowires, nanobelts, nanoflakes, nanourchins, and a plethora of other morphologies [18]. These are typically synthesized through high temperature vapour phase transport (VPT) methods, or low-temperature hydrothermal / solution phase techniques. The former techniques are the most widely used, despite being expensive and prohibitive to scale. On the contrary, hydrothermal methods are cheap and very easy to scale, but the nanowires which are produced can often have unintentionally high charge carrier concentrations [11].

1.2.1 Electronic properties

Electron donors in intrinsic ZnO

The electronic properties of bulk and nanostructured ZnO have been investigated in detail over several decades, although variability across different reported samples has made authoritative measurements difficult [19]. As such, the electronic properties of ZnO are still not well understood. ZnO is a natively n-type semiconductor at room temperature, caused by shallow electron donors which are seemingly unavoidable in the material. The exact origin of the electron donors is unclear and still a matter of study [20]. Temperature-dependent Hall-effect measurements suggest that these donors have ionisation energies between 30 to 70 meV [21].

Oxygen vacancies (V_O) and zinc interstitials (Zn_I) were initially proposed as shallow electron donors in ZnO, and are still often cited as being responsible for the n-type behaviour of as-grown ZnO. However, this assumption was seriously challenged around the turn of the millennium, with theoretical calculations by Kohan et al. [22] showing that both V_0 and Zn_I have high formation energies in n-type ZnO, and are also likely to be relatively deep donors. These calculations suggested that both defects would be relatively uncommon due to their formation energies, and if they were present in crystal ZnO, their ionisation energies would be too high to donate free electrons [19]. Conversely, other research has suggested that Zn_I ionises at lower energies than previously thought [23, 24], and could possibly donate shallow electrons in ZnO. However, the rather large formation energy of Zn_I restricts the number of defects present in ZnO, limiting its ability to affect the conductivity of the material.

The role of V_O and Zn_I as electron donors in ZnO was further challenged by theoretical research published by Van de Walle [25], which showed that incorporated hydrogen is a shallow donor in ZnO with a low enough formation energy to be abundant. As such, it was suggested that H is unintentionally incorporated in ZnO, where it becomes a dominant background donor and is responsible for the n-type behaviour. The role of H in ZnO has been tested in high-quality bulk ZnO, and has been experimentally confirmed to act as a shallow donor [26].

Nevertheless, there is still disagreement about the exact dopants which donate electrons in ZnO, and whether certain dopants are shallow or deep electron donors. Other research suggests that native defects may act as shallow donors by forming complexes [19], as many native defects are only considered as point defects in theoretical calculations. Furthermore, other structural defects such as stacking faults and line defects may affect the electrical properties of ZnO, and these have not been thoroughly investigated yet. As such, there is still a lot to learn about electron donors in ZnO. A discussion of the different possible dopants and their effects on ZnO nanowire FETs is included in Chapter 3.

Residual charge carrier concentrations in ZnO

Regardless of the exact nature of the donors, it is abundantly clear that ZnO is an n-type material. Due to the variability of electrical properties between reports in ZnO, it is difficult to find values of the charge carrier concentration which are representative. High-quality bulk intrinsic ZnO typically has a charge carrier concentration on the order of 1×10^{16} cm⁻³ in the best case [21], although more often is higher by an order of magnitude or more [27]. Figure 1.4 shows the measured charge carrier concentration of high-quality intrinsic bulk ZnO as a function of $10^3/T$ (K⁻¹), alongside its theoretical fit. There have been reports of lower charge carrier concentrations in thin films of ZnO, on the order of 1×10^{15} cm⁻³ in (0001) oriented ZnO [28] and 1×10^{14}

cm^{-3} in (1010) oriented ZnO [29].

The charge carrier concentration for nanostructured ZnO also shows a significant amount of variation, although tends to be significantly higher than high-quality bulk crystals. As-grown nanowires typically have charge carrier concentrations on the order of $10^{18} - 10^{19}$ cm⁻³ [30, 31]. High charge carrier concentrations pose a severe problem for FET applications, as they necessitate very large gate voltages to deplete the nanowire channel and achieve switching. In particular, ZnO is degenerately doped at $n \approx 4 \times 10^{18}$ cm⁻³, which renders most as-grown nanowires incompatible for FET use [11]. This value is explicitly calculated in Chapter 3, and the techniques used to reduce the charge carrier concentration for FET use are also described in detail with reference to electron donors present in the nanowires.



Figure 1.4: The relationship between the charge carrier concentration and temperature for high-quality bulk ZnO. Reproduced from [21]

Electron mobility in ZnO

The mobility of electrons in intrinsic ZnO has been reported several times for both bulk crystals and nanostructures, although reported values can differ considerably. The electron mobility in bulk crystals is most often determined by Hall-effect measurements, and is typically in the range of 200 cm² V⁻¹ s⁻¹ [12] for low fields at room temperature. Initial Monte Carlo simulations reported by Albrecht et al. [32] predicted that the room-temperature mobility of electrons in intrinsic ZnO should be approximately 300 cm² V⁻¹ s⁻¹ at room temperature, which has been corroborated by further theoretical calculations yielding values around 285 - 300 cm² V⁻¹ s⁻¹ [33, 34]. However, reported mobilities of bulk ZnO are almost always lower than this, with most mobilities around 150 cm² V⁻¹ s⁻¹ or lower [12].

One of the highest electron mobility reported in bulk ZnO is 205 cm² V⁻¹ s⁻¹ at room temperature [21], which agreed very well with non-Monte Carlo predictions. This data is shown in Figure 1.5. The theoretical model used to predict the electron mobility is based on solving the Boltzmann transport equation using Rode's method, and taking into consideration major sources of scattering, such as polar-optical-phonon scattering and Coulomb scattering. The peak value was approximately 2000 cm² V⁻¹ s⁻¹ at 50K.



Figure 1.5: The relationship between the electron mobility and temperature for highquality bulk ZnO. Reproduced from [21]

In comparison to bulk values, reported mobilities for ZnO nanostructures are typically below 100 cm² V⁻¹ s⁻¹, although particularly high-quality samples can sometimes show mobilities in the range of 1000+ cm² V⁻¹ s⁻¹ at room temperature [35, 36, 37, 15]. However, these are typically regarded as anomalous, and some authors have questioned these results [38]. The majority of hydrothermally-synthesised devices have mobilities under 50 cm² V⁻¹ s⁻¹; these results are reviewed and summarised in Chapter 3.

ZnO generally has a lower mobility than GaN due to its higher effective electron mass and larger optical phonon scattering parameter [13]. However, ZnO has been predicted to have a higher saturation velocity than GaN under high-field conditions. Albrecht et al. [32] theorised that the saturation velocity for bulk ZnO would be 3.2×10^7 cm s⁻¹ at 270 kV cm⁻¹. The predicted high-field drift velocities of ZnO and GaN as functions of temperature are shown in Figure 1.6. These results suggest that ZnO could outclass GaN in high-field conditions, which is extremely useful for high speed devices. Further theoretical publications have reinforced these claims, with predicted saturation velocities on the order of $1.5 - 2.2 \times 10^7$ cm s⁻¹ [33, 34]. Although there is comparatively little experimental research on high-field transport in ZnO, there are reports of saturation velocities on the order of 7.6×10^6 cm s⁻¹ [39].



Figure 1.6: Monte Carlo simulations of the drift velocities of ZnO and GaN in high-field conditions. ZnO has a higher saturation velocity than GaN. Reproduced from [32]

1.3 Thesis outline

This thesis consists of 9 chapters, with 4 results chapters devoted to three distinct research topics - namely, the hydrothermal growth of vertical nanowires, the hydrothermal growth of lateral arrays of nanowires used as FETs *in situ*, and the use of individual ZnO nanowires as FETs.

Chapter 2 describes the hydrothermal growth of ZnO nanowires. The current literature is discussed and referenced to explain the growth mechanisms and kinetics of ZnO nanowires. Homogeneous precipitation of ZnO is considered, and the heterogeneous nucleation from seeded substrates is discussed. These seeded substrates are used in the ensuing vertical and lateral growth chapters.

Chapter 3 reviews the current state of hydrothermally-synthesized ZnO nanowires in FET applications. The chapter begins with the fundamental theory behind FETs, and the relevant metrics and device parameters used to determine their quality. A review of the literature featuring hydrothermally-synthesized FETs then follows, with both single-nanowire and multi-nanowire devices discussed. The need for hightemperature annealing to achieve field dependence in both hydrothermal FETs and VPT FETs is also discussed. Theoretical papers predicting the expulsion of electron donors under high temperatures are considered in the context of field dependence and annealing.

Chapter 4 discusses the experimental methods used throughout this thesis. Topics include the fabrication of substrates for vertical and lateral nanowire growth and the electrical measurements of FETs.

Chapter 5 deals with the hydrothermal growth of vertical ZnO nanowires from thin films of ZnO. The impact of the molecular weight and concentration of polyethylenimine (PEI) (a typical additive in the hydrothermal growth) is explored and discussed. Vertical hierarchical nanowires are synthesised, which are caused by a PEI-mediated secondary growth phase driven by the release of Zn^{2+} ions through the Mannich reaction.

Chapter 6 presents FETs fabricated using lateral arrays of hierarchical nanowires which show field dependency without the need for any post-growth anneal or processing. The PEI-mediated hierarchical synthesis of ZnO nanowires is applied to lateral substrates, which works successfully. The growth mechanism of these hierarchical nanowires is related back to the hierarchical growth mechanism of vertical

nanowires presented in Chapter 5. FETs are fabricated by growing intersecting networks of the hierarchical nanowires, and all of the devices show field dependence. Persistent photoconductivity measurements indicate that the threshold voltage is strongly affected by the presence of the surface states, although they do not exclusively cause the observed field dependence of the nanowires.

Chapter 7 presents the modular two-pot synthesis of lateral arrays of hierarchical nanowires, and their use in FETs. The PEI-mediated growth technique of Chapter 6 is separated into two separate modular hydrothermal growths, and lateral hierarchical nanowires are successfully synthesised. The growth parameters of the two modular hydrothermal growths are investigated to optimise the yield of the hierarchical nanowires. Intersecting networks of the hierarchical nanowires are used as FETs. Dry measurements of the FETs are presented, which show limited intrinsic field dependence when used in air. Devices which are not field dependent as-is are wet-measured in de-ionised H_2O , and field dependence is achieved using back and top gates.

Chapter 8 presents the contacting of individual ZnO nanowires using electron beam lithography. Annealed and unannealed vertical nanowires are contacted in 3 total batches. All of the contacted nanowires show back-to-back Schottky contacts, which are modelled and fitted. The change in the Schottky contacts with exposure to UV light is investigated. The contacted nanowires show limited use as FETs, although two separate devices are fabricated. The nanowires show a drastic transition from conducting to insulating over 48 hours after liftoff. This transition can be temporarily reversed by exposing the nanowires to an electron beam. The nanowires disappear over a period of 8 months, most likely due to an interaction with the atmosphere. Accumulation of acidic carbon is considered.

Finally, **Chapter 9** is the conclusion of the thesis, summarising the work and laying ground for future research.

Chapter 2

Hydrothermal growth

2.1 Introduction

ZnO readily forms into a wide range of nanostructures due to the strong polarity of its crystal structure [18]. There are several different ways to synthesize these nanostructures, although most techniques can be broadly categorised into either vapour phase transport (VPT) techniques, or hydrothermal techniques [40]. This chapter focuses on the hydrothermal technique, as the hydrothermal growth is exclusively used throughout this thesis to synthesize ZnO nanowires. The chemistry of the hydrothermal growth is described in detail, along with some of the points of contention which still remain today. The growth of ZnO nanostructures is also described, and how homogeneous and heterogeneous nucleation affects the final product. Finally, nanowire growth from seeded substrates is discussed, which underpins the synthesis of the devices presented in this thesis.

2.2 Vapour phase transport

Vapour phase transport (VPT) synthesis techniques involve vaporised precursors which condense into ZnO nanowires. All of these techniques take place at high temperatures, and utilise expensive equipment. The most common VPT synthesis route is chemical vapour deposition (CVD). There are many variations of CVD procedures, including thermal evaporation [41], metal-organic [42], and carbon thermal reduction [43] amongst others. Despite the wide variation of procedures, most CVD techniques work on similar principles. The choice of substrate for CVD growth methods determines whether growth is possible at all, and if so what orientation the nanowires will take with respect to the substrate. Well-aligned vertical growth is achieved by exploiting lattice mismatches between the substrate and the ZnO nanowires, and has been demonstrated for a number of epitaxial substrates including both a-plane and

c-plane sapphire [44], Si [45], GaN and SiC [46], amongst others [38].

The necessity to exploit lattice mismatches between ZnO and the growth substrate, coupled with the high temperatures throughout the growth, greatly reduces the number of substrates that are compatible with VPT, and discourages direct growth onto device substrates or technologically relevant substrates. This thesis focuses on the hydrothermal method, which offers far greater flexibility due to its low cost and comparatively low temperature.

2.3 Hydrothermal methods

2.3.1 Introduction

The hydrothermal technique is one of the common methods of synthesizing ZnO nanostructures [10], including ZnO nanowires. The hydrothermal method is based on the hydrolysis of metal salts. Hydrothermal synthesis of ZnO nanowires takes place in water, with the precursors dissolved in the solution [40]. At least two precursors are typically used, with one precursor providing the zinc salt to be hydrolysed, and the other precursor providing the hydroxyl ions.

The hydrothermal synthesis of ZnO micro/nanostructures was first reported by Andre-Vergas et al. in 1990 [47], although the significance of this new synthesis method was somewhat overlooked. The authors synthesized ZnO microcrystals from aqueous solutions by combining zinc chlorides and zinc nitrates with hexamethylenetetramine (HMT). They investigated how the different variables of the synthesis process affected the reaction products, such as the temperature, the pH, the precursor concentrations, and the particular type of zinc salt used. This publication was eventually followed by Vayssieres in 2003, who hydrothermally synthesized ZnO nanowires on seeded substrates by mixing zinc nitrate hexahydrate with HMT in aqueous solution [48]. This publication was very well received, and the technique has since become the *de facto* standard for hydrothermal synthesis. Some of the ZnO microstructures and nanowires synthesized by Vayssieres and Andre-Verges et al. are displayed in Figure 2.1(a) and (b), respectively.

Although ZnO nanowires can be precipitated directly from the hydrothermal solution, they are typically grown on seeded substrates which are immersed within the solution. The growth solution is heated in a sealed vessel to relatively low temperatures (usually between 90 to 130 °C) at atmospheric pressure. This only requires a small amount of inexpensive and rudimentary equipment, such as a water bath or convection oven, making it very easy to scale up the hydrothermal synthesis of ZnO



Figure 2.1: ZnO nanowires hydrothermally grown via Zn(NO₃)₂ by (a) Vayssieres and (b) Verges. Adapted from references [48] and [47] respectively.

nanowires. In comparison, the expensive vacuum equipment and very high temperatures necessary for VPT synthesis make scaling expensive and difficult. The very low temperatures involved in the hydrothermal synthesis also makes the technique compatible with many vulnerable substrates, such as transparent and flexible materials, which are incompatible with high-temperature VPT methods.

2.3.2 Hydrothermal synthesis equipment and method

The simple hydrothermal set-up used throughout this thesis is displayed schematically in Figure 2.2. Individual 250 mL Schott bottles are filled with hydrothermal growth solution and were submerged in a Grant JBN12 water bath to raise their temperature to 95°C. The bottles can be left to pre-heat if necessary, before substrates are introduced into the growth solution. This allows the initial hydrothermal reactions to take place in the absence of a solid seed layer, restricting the ZnO growth to homogeneous nucleation. This can have a strong effect on the morphology of the final ZnO nanowires.

Seeded substrates are affixed to glass microscope slides using Kapton tape, and the slides are placed inside of the Schott bottles. Each slide can have multiple seeded substrates attached to it. Care must be taken to place the glass slides in the bottle so that the seeded substrates point "down", to ensure that homogeneously nucleated ZnO which precipitates out of solution doesn't fall on the surface of the substrates (indicated in Figure 2.2(b)). After the hydrothermal growth is complete, the slides are removed from the Schott bottles and placed in de-ionised H₂O. The substrates are then removed from the glass slides, individually rinsed in clean de-ionised H₂O, and



Figure 2.2: A simple schematic showing the equipment used to hydrothermally grow nanowires throughout this thesis.

dried with nitrogen at room temperature. The exact parameters and recipes of the hydrothermal growths used throughout this thesis are given in their relevant chapters.

2.4 Chemistry of the hydrothermal growth

As stated previously, the most common precursors used in the hydrothermal growth are zinc nitrate hexahydrate $[Zn(NO_3)_2]$ and hexamethylenetetramine (HMT) $[(CH_2)_6N_4]$. These precursors are dissolved using de-ionised H₂O as the solution. The $Zn(NO_3)_2$ provides Zn^{2+} ions, while the HMT provides the hydroxyl ions necessary for crystallisation of ZnO. The crystallisation of ZnO out of solution can be understood by the following reactions [10]:

$$(CH_2)_6N_4 + 6H_2O \longrightarrow 6HCHO + 4NH_3$$
(2.1)

$$\operatorname{Zn}(\operatorname{NO}_3)_2 \longrightarrow \operatorname{Zn}^{2+} + 2\operatorname{NO}_3^-$$
 (2.2)

$$NH_3 + H_2O \longleftrightarrow NH_4^+ + HO^-$$
 (2.3)

$$Zn^{2+} + 2HO^{-} \longrightarrow ZnO + H_2O$$
 (2.4)

$$\operatorname{Zn}^{2+} + 2\operatorname{HO}^{-} \longleftrightarrow \operatorname{Zn}(\operatorname{OH})_{2}$$
 (2.5)

$$Zn(OH)_2 \longleftrightarrow ZnO + H_2O$$
 (2.6)

HMT is highly soluble in water due to the inherent strain energy caused by its molecular structure [49], and readily hydrolyses into ammonia (4NH₃) and formaldehyde (HCHO) (as per Equation 2.1). Zinc nitrate hexahydrate is also very soluble and
rapidly decomposes, providing Zn^{2+} ions (Equation 2.2). The ammonia molecules produced by the decomposition of HMT lower the pH and react with the water to produce hydroxyl ions, as per Equation 2.3 [10]. These hydroxyl ions can combine with the free Zn^{2+} ions, directly crystallising out into ZnO (shown in Equation 2.4). ZnO can also crystallise indirectly through intermediate phases. The hydroxyl ions produced by the decomposition of HMT can complex with the Zn²⁺ ions to form Zn(OH)₂, shown in Equation 2.5 [10]. ZnO is finally derived by dehydrating the $Zn(OH)_2$, as per Equation 2.6. Dehydration of $Zn(OH)_2$ does not occur rapidly at room temperature, and requires a relatively small increase from room temperature. All of these reactions occur at a very small rate at room temperature, so the temperature of the growth solution has to be raised (or the concentration of the precursors increased) to thermodynamically increase the rate of reaction [10]. If the temperature of the growth solution is too low, ZnO nanowires will fail to crystallise out of solution. Increasing the temperature of the growth solution typically causes longer nanowires with higher aspect ratios and smaller diameters. This is due to an increased rate of reaction between the chemical precursors, and increased mobilities and diffusion lengths for the Zn and O ions in solution [10].

The chemical equations shown above only describe a somewhat simplified version of what really occurs. Many intermediary complexes can be formed through the combination of hydroxyl ions and zinc ions, forming complexes such as $[Zn(OH)]^+$ and $[Zn(OH)_4]^{2-}$ [50, 51]. These intermediaries undergo further reactions during the growth, and do eventually dehydrate in a similar way to $Zn(OH)_2$, although it isn't known how these intermediaries affect the hydrothermal growth. Some of these complexes can form very large molecules, such as $[Zn_{57}O_{27}(OH)_{56}]^{4+}$ [52]. In addition to possibly affecting the chemical reactions leading to ZnO nanowire growth, these large complexes may dehydrate over longer periods of time than smaller complexes, which may affect the rate of nanowire growth, or even act as a second, delayed period of nanowire growth.

2.4.1 Role of HMT

The role of the HMT is more complex than shown here, and its exact role in the growth chemistry is still under debate [53, 54, 55, 56, 57, 58]. Not only does the decomposition of HMT provide the hydroxyl ions necessary for the crystallisation of ZnO, the ammonia which is produced stabilises the Zn^{2+} ions as per Equation 2.7:

$$Zn^{2+} + 4NH_3 \longleftrightarrow [Zn(NH_3)_4]^{2+}$$
(2.7)

Free Zn²⁺ ions in the growth solution are consumed over the course of the reaction, crystallising directly or indirectly into ZnO. Similarly, the zinc-ammonia complex

shown in Equation 2.7 gradually decomposes over time in response, causing a stable concentration of Zn^{2+} in the growth solution and a relatively low level of supersaturation. This low level of supersaturation preferentially encourages heterogeneous crystal growth on the seeded substrates and discourages homogeneous crystal growth in the bulk of the solution, which yields longer ZnO nanowires on the seeded substrates [56, 59, 60, 61].

In addition to regulating the concentration of Zn^{2+} ions in the solution, the decomposition of HMT causes a controlled and sustained release of hydroxyl ions. The rate of HMT decomposition and hydroxyl production is determined by the pH of the solution, with an increased rate of production at low pH values, and a decreased rate of production at high pH values. This effectively buffers the pH of the solution and regulates the decomposition of HMT, ensuring the gradual and sustained release of hydroxyl ions [10, 49]. This effective bottleneck on the supply of hydroxyl ions has a very important effect on the nanowire growth, as a continual source of hydroxyl ions is necessary for nanowires with high aspect ratios. If the HMT decomposed very rapidly, the excessive OH⁻ ions would quickly precipitate the Zn^{2+} ions out of solution, exhausting all of the precursors over a very short period of time and making ZnO nanowire growth impossible [10, 49].

Several groups and publications have looked into the possibility that HMT binds to the non-polar side faces of wurtzite ZnO, encouraging anisotropic growth along the c-axis of the nanowire. The first of these publications argued from a phenomenological approach, observing that nanowire growth was restricted almost exclusively to the c-axis without a corresponding increase in the radius of the nanowire [56]. It was argued that HMT, as a non-polar chelating agent, was binding to the non-polar faces of the ZnO nanowires, encouraging further precursors to exclusively attach to the uncapped polar c-axis face. This possibility was specifically investigated in a separate publication using attenuated total reflection Fourier transform infra-red (ATR-FTIR) spectroscopy to gauge the molecules adsorped to the faces of hydrothermally synthesized ZnO nanoparticles[57]. They concluded that there was no presence of HMT on the surface of the nanoparticles, and that HMT merely acted as a source of hydroxyl ions. This was supported by growing nanowires in an aqueous mixture using Zn(NO₃)₂ as the zinc ion source, and a non-HMT pH buffer and KOH to maintain HMT-esque pH conditions.

This was followed by a further publication which explicitly investigated the role of HMT in the hydrothermal growth [58]. The authors hydrothermally grew ZnO nanowires using a wide range of $Zn(NO_3)_2$:HMT ratios and recorded how the diam-

eter and length of the nanowires varied. They concluded that the HMT has a direct influence on the axial growth of the nanowires, significantly reducing their diameter while also increasing their length. These observations were supported by Raman spectroscopy, which was used to gauge the presence of molecules on the surface by looking at stretching O-H and N-H modes. These measurements suggested that complicated physio-chemical interactions between the HMT and ZnO sidewalls occur during the hydrothermal growth. The role of HMT was further complicated by its interaction with the seeded growth substrate itself, which significantly affected the nucleation and overall density of ZnO nanowires from the seeds. These findings were also corroborated by another recently published paper, which reported the shape of ZnO nanowire cross-sections changing from a hexagon to a rounded cylinder with increasing HMT [62]. This effect may be caused by the preferential attachment of HMT to the side faces of the nanowire, as similar results have been observed with other capping agents (discussed in the next section.)

There are several other variables and factors which have been reported to affect the hydrothermal growth which aren't adequately understood or accounted for in the current model. For example, although the zinc salt is usually considered as solely providing the Zn²⁺ ions necessary for ZnO growth, there is evidence which suggests the counter-ions have a significant effect on the morphology of the grown nanowires. At least six different zinc salts have been investigated, with the resulting ZnO nanocrystals being star-shaped, flower-like, rod-like, or needle-like depending on the specific salt used [63]. Similarly, oxygen dissolved in the aqueous solution isn't considered in the common models for hydrothermal growth, and there is evidence that it may have a strong effect on the hydrothermal growth [10].

The various known and unknown factors that affect the hydrothermal growth can have a negative and frustrating impact on the reproducibility of the nanowires. Sometimes drastic measures are taken, with at least one recorded instance of ZnO nanowires being grown only strictly during the weekends and evenings, to reduce any possible impact of a populated building on the experiments' reproducibility [64]. In summary, the exact mechanism of HMT-mediated ZnO nanowire growth is clearly still a point of contention, and numerous experimental results run contrary to the relatively simple chemical equations which in theory govern the hydrothermal growth.

2.4.2 Polyethylenimine as a growth additive

Growth additives can be included with the regular Zn(NO₃)₂ and HMT precursors in the growth solution to affect the morphology of the ZnO nanowires. These typically work by selectively binding to specific faces of the ZnO crystal during the hydrothermal growth, prohibiting further growth in that direction. The binding of the additive to the nanowire face is usually mediated through electrostatic forces, which are either attracted to the polar (c-axis) or non-polar (side faces) of the ZnO nanowire [10]. Additives which bind to the polar face of the nanowire (such as citrate ions [65, 66, 67]) encourage growth in the radial direction, giving rise to low aspect-ratio nanowires or nanocrystals. In comparison, additives which bind to the side faces of the nanowire encourage growth in the c-axis direction, increasing the aspect ratio of the nanowires. The latter is preferable for many applications, and aspect-ratio enhancing additives have been widely studied and used throughout the literature [68, 69, 70, 71, 72, 73, 74]. The capping of polar and non-polar faces of ZnO nanowires is schematically depicted in Figure 2.3, and their influence on the morphology of the nanowire.

One of the most widely used additives to increase the aspect ratio of ZnO nanowires is polyethylenimine (PEI), and is used extensively in this thesis. PEI is a repeating non-polar polymer consisting of primary, secondary, and/or tertiary amine groups in a linear or branched arrangement. A schematic of a repeating PEI molecule is shown in Figure 2.4. It was initially argued that PEI increases the aspect ratio of the nanowires by binding to the side faces of the ZnO nanowire during growth, due to the affinity of the amine groups to the non-polar crystal faces in a basic environment [68]. However, it has been shown that the role of PEI in the hydrothermal growth is significantly more complicated than this, and has an effect on the fundamental chemical reactions which drive the hydrothermal growth [75].

PEI strongly chelates with Zn^{2+} ions in the growth solution [75] by forming NH₂-Zn complexes with the terminating amines of the PEI molecules. This introduces an alternative reaction pathway for the Zn^{2+} ions, which competes with and suppresses the homogeneous precipitation of ZnO out of solution. This also lowers the concentration of Zn^{2+} ions at any given time, which has an effect on the heterogeneous nucleation of ZnO. This interaction between PEI and the Zn^{2+} ions is expressed by Equation 2.8:

$$Zn^{2+} + PEI-NH_2 \longrightarrow PEI-NH_2-Zn^{2+}$$
 (2.8)

Critically, the Zn^{2+} ions complexed with PEI can be released back into the growth solution after some time, providing more reagents for inhomogeneous nucleation and nanowire growth. This release of Zn^{2+} ions from the PEI-Zn complexes is caused by the Mannich reaction. The Mannich reaction is defined generally as the amino alkylation of an acidic proton situated beside a carbonyl functional group, mediated



Figure 2.3: (a) A ZnO nanowire, with the polar c-axis highlighted. (b) Polar and nonpolar capping agents bind to specific faces of the ZnO nanowire, depending on the electrostatic interaction between the crystal and the capping agent. (c) Polar capping agents encourage non-axial growth, while non-polar capping agents encourage axial growth.



Figure 2.4: A schematic of a branched PEI molecule, showing the linear chain and branching amine groups. Figure adapted from [76]

through formaldehyde and a primary or secondary amine or ammonia. The chemistry involved in the generalised Mannich reaction lies beyond the scope of this thesis, and is considered here only in the framework of the hydrothermal growth of ZnO nanowires.

In this context, the Mannich reaction occurs between formaldehyde (HCHO) and the primary or secondary amines which terminate from branched PEI. The reaction between HCHO and the primary NH_2 amines is shown in Figure 2.5. The HCHO first interacts with the secondary amines, shown in Figure 2.5(a), forming the complex shown in (b). This complex then dehydrates into an imine group (N=CH₂), forming what is known as a Schiff base. Replacing the NH_2 amines with imine groups significantly reduces the chelating ability of PEI, releasing chelated Zn^{2+} ions back into the growth solution. This interaction is shown in Equation 2.9:



 $PEI-NH_2-Zn^{2+} + HCHO \longrightarrow Zn^{2+} + PEI-N=CH_2$ (2.9)

Figure 2.5: The Mannich reaction between NH_2 and HCHO. (a) Branched NH_2 interacts with HCHO, which forms an intermediary complex shown in (b). This dehydrates into the Schiff base, shown in (c). Figure reproduced from [77]

The formation of imine groups via the Mannich reaction also causes a change in colour to the growth solution. Imine groups absorb photons between 300 - 450 nm

through $\pi \to \pi *$ and $n \to \pi *$ transitions [75], which causes a yellow to orange colour in the growth solution which increases with time (an example is shown in Figure 2.7). This colour change is not observed in the hydrothermal growth solution when PEI has been omitted, which lends further evidence to the Mannich reaction taking place.

Including PEI into the hydrothermal growth solution also has further impacts on the hydrothermal growth chemistry, in addition to chelating and eventually releasing Zn^{2+} ions. PEI molecules can protonate with the NH₃ provided by the decomposition of HMT, forming NH₄⁺ and a PEI-NH₃⁺ complex. This in turn affects the intermediary states described by Equation 2.7. The protonation of NH₃ also provides OH⁻ groups into the hydrothermal growth solution, contributing to the formation of Zn complexes which in turn dehydrate into crystal ZnO. The extra production of OH⁻ ions through the protonation of NH₃ groups also offsets the decreased pH caused by the consumption of -NH₂ via the Mannich reaction.

The heterogeneous nucleation of ZnO through the PEI-influenced hydrothermal growth is depicted in Figure 2.6. This shows (1) the chelation between Zn^{2+} ions and PEI; (2) the protonation of the primary amine -NH₂ groups from PEI; (3) the decomposition of HMT into HCHO and NH₃; (4) the Mannich reaction between the primary amines of PEI and HCHO; (5 - 6) the heterogeneous nucleation of ZnO through the dehydration of Zn-OH complexes.



Figure 2.6: A diagram depicting the crystallisation of ZnO nanowires when PEI is included in the hydrothermal growth. Reproduced from [75]

In summary, the net effect of including PEI in the growth solution is an increase in nanowire aspect ratio and length, although the mechanism is much more complex than only encouraging growth in the c-axis direction through capping the side faces of the nanowire. The most significant effect of PEI is the strong chelation of Zn^{2+} ions out of the growth solution, which suppresses homogeneous precipitation in much the same way as the zinc-ammonia complexes shown in Equation 2.7. These Zn^{2+} ions are introduced back into the solution over time via the Mannich reaction, supplying more precursors for nanowire growth [75].

Further evidence for this reaction is shown by the gradual yellow-orange colour change to the growth solution, caused by the conversion of primary amine to imine groups in the PEI molecules. This differs significantly from hydrothermal growths where PEI is not included, in which the HCHO produced by the decomposition of HMT does not play a role in the growth chemistry. Critically, the ability to complex and reintroduce Zn^{2+} ions into the growth solution can have a significant effect on the final nanowires, which is demonstrated in Chapters 5 and 6 of this thesis.

2.5 Hydrothermal growth of ZnO nanowires

As discussed in the introduction (Chapter 1), ZnO almost exclusively forms in the hexagonal wurtzite structure . This crystal structure has a polar c-axis which terminates in either O atoms (the (0001) face) or Zn atoms (the $(000\bar{1})$ face). The other faces of the hexagonal wurtzite crystal are non-polar, such as the side faces. Generally speaking, polar facets have a high surface energy associated with them, while the non-polar facets have a low surface energy [10]. The hydrothermal growth of a ZnO crystal is driven by minimising the energy of the thermodynamic system [10], which means that precursors will preferentially migrate to the polar face and the crystal will grow in the c-axis direction. This self-assembly and natural anisotropy of crystal growth is one of the attractive features of using ZnO as a nanostructure.

2.5.1 Homogeneous growth

Homogeneous nucleation of ZnO occurs when all of the reactants involved in the nucleation are in a solution phase [78]. Zn^{2+} ions directly react with the hydroxyl ions provided by the decomposition of HMT, and either directly crystallise into ZnO or form complexes [10]. The initial complex serves as a scaffold for further Zn^{2+} ions and hydroxyl groups to attach to, eventually incorporating the ions and forming a crystal lattice through dehydration. This occurs without any external catalyst or initial point of nucleation. The crystal continues to grow in size over time as more ions are supplied from the growth solution, preferentially interacting with the less thermodynamically stable facets of the crystal. The reliable flux of constituent ions from the solution to the crystal occurs due to the supersaturation of precursors in the

growth solution [79], and abates when the system is thermodynamically stable or when the precursors are exhausted from the solution.

Homogeneous growth of ZnO nano/microwires is typically undesirable and unintentional. The reaction products can be isolated out from the growth solution and presumably used for device applications through drop-casting methods, etc., although the products are often very large in size, and have very poor aspect ratios. The hydrothermal method is often used to synthesise nanowires in predefined locations on pre-seeded substrates, and homogeneously nucleated ZnO poses a problem for device fabrication. Firstly, ZnO nanocrystals which precipitate out of solution can fall onto the substrate surface, potentially shorting electrodes or otherwise fouling the substrate. Secondly, the precursors that are consumed through homogeneous nucleation are no longer available for heterogeneous nanowire growth. This reduces the overall length of the ZnO nanowires (by reducing the total volume of ZnO), and in the past this has necessitated refreshing the growth solution [69, 73, 80].

There are several ways to prohibit and minimise the impact of homogeneous nucleation of ZnO during the hydrothermal growth. To minimise the risk of homogeneously grown ZnO from depositing on the substrate, the substrates are typically floated face-down on the surface of the growth solution over the course of the growth, or are attached to glass slides angled downwards. To reduce the homogeneous nucleation of ZnO in the first place, the chemistry of the hydrothermal growth can be tailored to reduce the level of supersaturation in the growth solution. This can be achieved by adding NH₃ (although the decomposition of HMT naturally introduces NH₃ into the solution), or including PEI in the growth solution. These effects can be very drastic and obvious. Figure 2.7 shows a series of Schott bottles containing hydrothermal growth solution which have been allowed to pre-heat for an hour. From left to right, the bottles contain 6 mM, 8 mM, 2 mM, and 3 mM of PEI(M_W = 800 g/mol). These different concentrations have significantly affected the homogeneous precipitation within the growth solution.

The turbidity of the growth solution clearly differs between the bottles, the bottles containing 2 mM and 3 mM of PEI being almost opaque on account of the precipitation within the growth solution. In comparison, the bottles containing 8 mM and 6 mM are completely transparent, and free of any precipitation. The various shades of yellow and orange of the growth solution are caused by the complexing of PEI with HCHO via the Mannich reaction [75].



Figure 2.7: A series of bottles which have been left to preheat for one hour. From left to right, the bottles contain 6 mM, 8 mM, 2 mM, and 3 mM of $PEI(M_W = 800 \text{ g/mol})$

2.5.2 Heterogeneous growth

In contrast to homogeneous nucleation, heterogeneous growth occurs when the initial point of nucleation is a crystal in the solid phase, while the reactants to the growth are in the liquid phase [40]. This is usually the type of nucleation and growth exploited in the hydrothermal synthesis of ZnO, as it allows for ZnO nanowire growth directly from pre-defined crystal seeds or catalysts [59]. The crystal seeds or catalysts may be suspended in the solution, although the more common technique (and the technique used exclusively throughout this thesis) is to heterogeneously grow the nanowires from seeded substrates [59]. As the reaction takes place, ions are incorporated into the existing crystal lattice, and the crystal grows larger. The constituent ions are supplied over time in much the same way as homogeneous growth. Whether homogeneous or heterogeneous growth is favoured depends on several factors, such as the supersaturation of precursors in the solution [56, 59, 60, 61].

Different ZnO crystal seeds have been used throughout the literature, such as ZnO nanocrystallite films [81, 82] or deposited thin ZnO films [83, 84, 85, 86]. Seedless growth can also be achieved using intermediary layers like Ti/Au [87], although seeded ZnO films are the more common substrates used. Seeded growth is one of the main benefits of the hydrothermal method, as it allows the synthesis of nanowires on a wide variety of different substrates. Because the growth conditions themselves are fairly benign (e.g. below 100 °C and in a slightly basic environment), if the substrate is compatible with seed layer deposition, then it is most likely compatible with the hydrothermal growth. To this end, nanowires have been hydrothermally grown on a wide variety of interesting and relatively vulnerable substrates, such as paper [88], flexible plastics [89], carbon fibres [90], and other fabrics [91].

2.5.3 Growth from seeded substrates

Vertical growth

ZnO nanowires tend to grow normally (i.e. at a 90° angle) from thin films of ZnO used as seed layers, although the orientation of the nanowires can show some variation depending on several factors of the ZnO film [83, 84, 85, 86]. Figure 2.8 shows ZnO nanowires grown from a thin layer of Au and an RF sputtered ZnO thin film on silicon substrates. In both cases the nanowires have grown normally from the substrate and their respective seed layer, although the orientation of the nanowires is significantly better from the ZnO thin film. This results in a vertical "forest" of ZnO nanowires, which can be isolated from the growth substrate and used for device fabrication individually. Vertical nanowire forests have also been successfully used in devices with



Figure 2.8: Top-down images of ZnO nanowires hydrothermally grown from thermally deposited Au (a) and RF sputtered ZnO (b) thin films.

several applications, such as power generation, lasing, and dye-sensitized solar cells [10].

Lateral growth

Hydrothermal growth from seeded substrates can also be used to grow lateral arrays of ZnO nanowires, and is one of the main advantages over VPT methods. Although it is possible to grow nanowires laterally using VPT/CVD growth [92], the results are often less than ideal, and the process is incompatible with a large number of substrates due to the high temperatures involved in the nanowire synthesis. The hydrothermal growth of lateral arrays of nanowires has been generalised to other substrates using sputtered seed layers of ZnO [93, 94, 95, 96, 97, 98]. As the nanowires grow normal to the face of the ZnO seed layer, it is necessary to fabricate a seed layer which is perpendicular to the substrate. This is most easily achieved by patterning and depositing a ZnO thin film with a capping layer. The capping layer prevents nanowire growth and nucleation from the top face of the seed, while the sidewall of the ZnO layer remains exposed to the growth solution, encouraging lateral growth. This is depicted schematically in Figure 2.9 .

Hydrothermal growth of lateral ZnO nanowires can also be achieved by combining electron beam lithography with single-crystal ZnO substrates orientated in the (2110) direction [99]. Extremely well-ordered arrays of ZnO nanowires are able to be grown in this way, and could be transferred to flexible substrates with no loss of order to the original array. The downside to this approach is the requirement for a single-crystal ZnO substrate, textured in a single direction.

In practice, hydrothermally-grown lateral nanowires often grow in a 180° arc



Figure 2.9: (a) A seeded substrate with a capped thin film of ZnO is submerged in hydrothermal growth solution. Zn^{2+} and O^{2-} ions nucleate on the sidewall of the thin film, which is exposed to the growth solution. (b) After growth, nanowires grow laterally from the sidewall.

from the seed layer. As the ZnO nucleates and grows at the seed layer sidewall, the crystal agglomerates into a large nucleus, partially delaminating the seed layer from the substrate. ZnO nanowires then grow from this nucleus and the underside of the seed layer, which causes the wide-angle brush orientation of the nanowires. Figure 2.10 shows a lateral array of ZnO nanowires. Figure 2.10(a) shows the lateral array from a bird's eye view. The SiO₂ substrate and capping layer have been labelled. ZnO nanowires are laterally growing from the exposed sidewalls of the ZnO seed, although they are clearly growing in a wide arc rather than purely flush to the substrate. Figure 2.10(b) shows a cross-section of the lateral array marked by the dashed box in (a). It is clear that the capping layer is preventing vertical nanowire growth, and that the ZnO seed layer has partially delaminated from the substrate. This partial delamination has caused the nanowires to grow in a 180° arc from the sidewall. This arc is shown in cross-section in the upper-right inset.



Figure 2.10: (a) A bird's eye view of a lateral array of ZnO nanowires. The capping layer has prevented vertical growth. The dashed box is shown in cross-section in (b). The seed layer has partially delaminated from the SiO_2 substrate, allowing the nanowires to grow in an arc. The large arc is shown in cross-section in the upper-right inset.

2.6 Conclusion

In conclusion, the hydrothermal growth is a powerful synthesis technique given its versatility. It can be combined with a wide variety of seeded substrates due to the low temperatures involved in the growth process. Zinc nitrate hexahydrate and HMT are the most common precursors used in the growth, although there is still some ambiguity regarding the exact growth mechanisms and chemical interactions which take place. PEI is often used as an additive, and adds another layer of complexity to the hydrothermal growth. Heterogeneous nucleation is favoured for most hydrothermal growth applications, with homogeneous nucleation intentionally suppressed.

Chapter 3

Field-effect transistors

3.1 Introduction

In this chapter, sufficient background on field-effect transistors is given to understand and contextualise the results presented in this thesis. The threshold voltage, on-off ratio, transconductance, and subthreshold swing device parameters are defined and explained. The literature regarding ZnO nanowire FETs is reviewed, with an emphasis on hydrothermally synthesised nanowires. The role of electron donors in ZnO nanowires is considered in the context of FET fabrication, and how they are addressed by high temperature annealing in the literature. The chapter concludes with a summary of the literature review, which motivates the need to reliably produce ZnO nanowire FETs which show field dependence without annealing.

3.2 FET Background

3.2.1 Mode of operation

Field-effect transistors (FETs) are a fundamental component in modern electronics, and are used in a wide variety of applications. An FET consists of a semiconducting channel, contacted by two electrodes (called the drain and source), and separated from a third electrode (the gate electrode) by an insulator. The FET can be thought of as an electronic switch, where the conductivity of the semiconducting material is modulated by applying an external electric field via a voltage difference between the gate and source electrodes. This modulation of conductivity is caused by the bending of the conduction and valence bands of the semiconductor [100]. Depending on the direction of the bending, charge carriers can be introduced or removed from the channel. This is known as the field effect.

Figure 3.1 shows a schematic of the conduction and valence bands of an n-type semiconductor as a positive voltage is applied to the gate electrode (shown to the right of the insulator.) When no gate voltage is applied, the charge carrier concentration in the semiconductor is low, as the Fermi level falls within the band gap of the material. When a positive gate voltage is applied, the conduction band of the material at the semiconductor/insulator interface is bent downwards below the Fermi level, filling those states with electrons and introducing charge carriers into the channel. This is known as an enhancement mode of operation, as a positive voltage is required to accumulate charge carriers. An alternative mode of operation is depletion mode, where the channel is populated with electrons at zero bias, and a negative bias is applied to bend the conduction band above the Fermi level.



Figure 3.1: Band bending of a semiconductor via the field effect. At zero bias, the channel is depleted. At a positive bias, the conduction band is bent below the Fermi level, populating the channel with electrons. Reproduced from [101].

3.2.2 Transfer characteristics and device parameters

One of the most important measurements used to assess an FET are the transfer characteristics of the device. These are measurements of the drain-source current, I_{DS} , versus the gate-source voltage, V_G , at a fixed drain-source voltage, V_{DS} . The gate voltage can be swept in both forward and backward directions. This can sometimes result in a discrepancy in I_{DS} , which is called a hysteresis. Figure 3.2 shows the transfer measurements of a standard n-type BS20 MOSFET on a logarithmic scale. The drain-source voltage is kept at a constant value of 0.1 V. The device is clearly off when V_G is below 0 V, which is called the off regime. Similarly, the FET is clearly on once V_G is above roughly 2.5 V. The ratio of these two currents is called the **on-off ratio** of the device, and is typically given as an order of magnitude. In this case, the on-off ratio is on the order of 10^{11} .



Figure 3.2: The transfer characteristics of a standard BS20 MOSFET. The device is driven at a drain-source voltage of 0.1 V

3.2.3 Threshold voltage - $V_{\rm Th}$

The value of the gate voltage which turns the device on or off is called the **threshold voltage** (V_{Th}) of the device. There are many different specific definitions of the threshold voltage, and several different techniques are used to extract the threshold voltage from multiple types of measurements [102]. In this thesis, all of the threshold voltages have been extracted using the linear extrapolation method. The transfer characteristics of the FET are plotted on a linear scale, and the linear portion of the drain-source current is extrapolated to intersect with the gate voltage axis. The point of intersection is then taken to be the threshold voltage.

The sign and magnitude of the threshold voltage is affected by the charge carrier concentration of the semiconducting channel. A high charge carrier concentration means that the device operates in depletion mode, requiring a negative threshold voltage to deplete the device of electrons. Higher charge carrier concentrations necessitate lower threshold voltages to switch between on and off states. Similarly, if the charge carrier concentration of the semiconductor is low, and the device is off at $V_G = 0$ V, then the device operates in enhancement mode, with a positive threshold voltage

inducing electrons into the channel through Schottky band bending. Lower charge carrier concentrations require higher threshold voltages to populate the channel with electrons. The threshold voltage of the device is also affected by other aspects of the channel, such as the surface states at the semiconductor/dielectric interface [100].

Figure 3.3 shows the transfer characteristics of the FET displayed in Figure 3.2, plotted on a linear scale and restricted to $V_G = 0$ to 5 V. The entire transfer characteristics on a linear scale are shown in the upper-left inset. A linear line (red) is extrapolated from the linear portion of the transfer characteristics, and intersects the V_G axis at \approx 2.2 V, which is taken to be the threshold voltage. This device operates in enhancement mode, as it is in the off regime at $V_G = 0$ V.



Figure 3.3: Extracting the threshold voltage of a standard BS20 MOSFET. The intercept between the linear extrapolation (red) and the V_G axis gives a threshold voltage of 2.2 V.

3.2.4 Transconductance - g_m

The transconductance of a device is defined as the change in drain-source current over a small change in gate voltage. More specifically, it is taken as the first derivative, $\frac{dI_{DS}}{dV_G}$, of the transfer curve. The transconductance is typically taken as the maximum value of the derivative. Figure 3.4 shows the transfer characteristics of Figure 3.2 plotted alongside its first derivative on a linear scale, plotted from V_G = 0 V. The maximum transconductance g_m is approximately 0.033 S.



Figure 3.4: The transfer characteristics of a standard BS20 MOSFET (black) plotted against its derivative (red) in order to find the transconductance. The maximum transconductance is approximately 0.033 S.

3.2.5 Subthreshold swing - SS

The subthreshold swing of a device is a measure of how fast the FET switches from the off state to the on state. This occurs over the subthreshold regime of the transfer characteristics, which is occurs at gate voltages slightly below the threshold voltage. The logarithm is taken of the transfer characteristics, and the subthreshold swing is then the inverse of the gradient of the subthreshold regime. It is typically given in mV/decade, where a decade refers to an order of magnitude of drain-source current. Approximately 60 mV/decade is considered to be the lowest subthreshold swing theoretically achievable by typical FETs [100].

Figure 3.5 shows the subthreshold regime and subthreshold swing of the FET measured in Figure 3.2. The logarithm has been taken of the transfer characteristics, which are plotted between $V_G = -0.5$ V and $V_G = 3.5$ V. The gradient of the linear portion of the subthreshold regime has been highlighted in red. The subthreshold swing of this particular device is approximately 180 mV/decade.



Figure 3.5: The subthreshold swing is inferred from the logarithm of the transfer characteristics. The rise and the run are highlighted in red, and the inverse of the gradient yields a subthreshold swing of $\approx 180 \text{ mV/dec}$.

3.2.6 Mobility - $\mu_{\rm eff}$

The mobility of an FET reflects how fast charge carriers can travel through the semiconducting material, and is adversely affected by electron scattering from various sources. ZnO is almost exclusively n-type, and mobility estimates are typically restricted to electrons. The mobility of an FET is typically measured by using the device in the linear region of the transfer characteristics, and is given as [103, 104, 105, 54, 106, 107]:

$$\mu = g_m \times \frac{\mathrm{L}^2}{\mathrm{V}_{\mathrm{DS}} \times \mathrm{C}} \tag{3.1}$$

Where g_m is the transconductance of the FET, L is the length of the channel, V_{DS} is the drain-source voltage, and C is the capacitance of the FET. Extractions of the FET mobility therefore rely on accurately knowing the exact channel length, and calculating the capacitance of the device. Most of the devices in this thesis use lateral networks of ZnO nanowires as FETs, which makes these calculations very difficult to estimate. However, mobilities of individual nanowires are considered in Chapter 8.

3.3 Hydrothermal ZnO nanowire FETs in the literature

ZnO nanowire and nanostructured FETs synthesised by vapour phase transport (VPT) have been investigated as early as 2003 [108], while hydrothermally synthesised nanowire FETs appear as early as 2005 [109]. In general, all VPT grown ZnO NW FETs use NWs that are grown at high temperatures and then removed *en masse* from the growth substrate, prior to being randomly dispersed onto the device substrate. The hydrothermal synthesis offers the advantage of directly growing laterally aligned ZnO NWs which are in the correct position for an FET device. However, one major drawback is the need to anneal the hydrothermally grown ZnO NWs to above 400°C in order to observe semiconducting behaviour [110]. The need for an anneal stage will be discussed, along with the subsequent changes to device performances and characteristics.

Despite the inherent advantages in using in-situ laterally-aligned NW arrays as the basis for FETs, there are only a limited number of research teams who have looked at the performance of laterally aligned hydrothermal ZnO nanowires for FET applications. These can be split into devices with a dense multi-nanowire array channel, and those with a single isolated ZnO NW. Both offer insight into the behaviour of the hydrothermally grown ZnO NWs and their suitability for FET devices. Importantly, both multi-nanowire and single-nanowire FETs offer a playing ground to explore the mechanisms of conduction in ZnO, with a nanowire system presumably rich in surface defects.

3.3.1 FETs fabricated with lateral NW arrays

Park et al.

Park et al. [93] have fabricated FETs using in-situ laterally-aligned ZnO nanowire arrays synthesised by the hydrothermal method. Standard photolithography and lift-off techniques are used to pattern two ZnO seeds, separated by a small distance, on an SiO₂(300 nm)/p-Si substrate. Each ZnO seed is individually capped with Ag to deter vertical NW growth and to act as a contact metal, similar to Figure 2.9 in Chapter 2. ZnO nanowires are grown using equimolar 10 mM Zn(NO₃)₂/HMT solution for 6 hours at 85°C. Lateral overlapping ZnO nanowires are grown from both ZnO seed side-walls, bridging the electrodes and forming a conduction channel. The resulting nanowires had an average diameter of 100 nm, and an average length of 5 μ m.

The nanowires are annealed in N_2 ambient for 2 minutes at 450 °C prior to being measured. Typical device characteristics indicated an n-type FET with an on-off ratio of 3×10^{-4} , a threshold voltage of \approx -35 V, a field-effect mobility of \approx 5.3 cm² V⁻¹ s⁻¹, and a transconductance of 250 nS. The FET does not display a saturation regime at high gate voltages, which is attributed to increased carrier scattering caused by junction effects and the large surface area of the active channels.

Improved FETs were fabricated by first laterally growing ZnO nanowires from a single ZnO seed side-wall, and subsequently metallizing the terminating NW ends with Ag using standard lithographic and lift-off techniques. These nanowires were also annealed in N₂ ambient for 2 minutes at 450°C prior to FET characterization. The resulting output characteristics are displayed in Figure 3.6, where (a) shows I_{DS} - V_{DS} curves at different values of V_G, and (b) shows transfer characteristics. The FET exhibits typical n-type behaviour, with an on-off ratio of 4×10^5 , a threshold voltage of \approx -15 V, a field effect mobility of 8.5 cm² V⁻¹ s⁻¹, and a transconductance of \approx 400 nS. Subthreshold swings were not provided, although are clearly >1 V/decade given the transfer characteristics. Similar to the original FETs, the drain-source current does not show a saturation regime, which is again attributed to surface defects. The improved device performance was attributed to the absence of the potential-energy barrier height caused by nanowire-nanowire junctions present in the channel of the original FET.



Figure 3.6: IV measurements from an improved FET fabricated by Park et al. (a) Drain current versus drain voltage curves at different gate voltages with 2 V steps (b) Logarithmic and linear (inset) scale transfer measurements taken at V_{DS} = 1.1V. Figure reproduced from [93].

Ko et al.

Ko et al. [111] have fabricated ZnO NW FETs on flexible substrates using as-grown NWs synthesised via the hydrothermal method. Gold nanoparticles were first

dispersed onto a dielectric/back-gated flexible substrate, forming a self-assembled monolayer. This monolayer was then imprinted with a PDMS mold which defined two separated electrodes, and subsequently heated to 140°C. This caused the gold nanoparticles to melt and coalesce into well-defined source and drain electrodes. The substrates were then spin-coated with a solution of ZnO quantum dots in an ethanol solution, which provide the seeds for nanowire growth.

Nanowires were grown by using an equimolar solution of 25 mM Zn(NO₃)₂/HMT at 65 - 95°C for 3 hours. The nanowires grown from these seeds were hexagonal wurtzite single crystals, verified by SEM and transmission electron microscopy (TEM). The average nanowire had a diameter between 150 - 300 nm, and a length between 1 - 3 μ m. The electrical characteristics and field dependence of the transistors were measured. The average FET was found to be n-type with an on-off ratio of 10⁴ - 10⁵, a threshold voltage of -5 V, and a transconductance of ~100 nS. The average subthreshold swing is provided as 8V/decade, which is a relatively high value. This may be due to the fact that the nanowires are unannealed. The field effect mobility could not be reliably estimated due to the uncertainty in the channel length of the transistors. Similar to Park et al. [93], the FET did not exhibit a clean saturation regime. This was attributed to the convoluted network of grain boundaries at the nanowire junctions, which prohibit carrier transport.

Pachauri et al.

Pachauri et al. [112] fabricated liquid-gated FETs using hydrothermally-synthesised nanowires grown in specific locations. Metal electrodes were first defined on an $SiO_2(600nm)/Si$ substrate using standard photolithographic techniques. These electrodes were then covered in ZnO nanoparticles through dielectrophoresis by applying an alternating current to the metal electrodes. This ensures that the nanoparticles adhere to the electrodes only, to allow for site-specific growth of the ZnO nanowires.

ZnO nanowires were then grown from the ZnO nanoparticles by submerging the samples into an equimolar mixture of $0.25 \text{ mM Zn}(NO_3)_2/HMT$ at 85 °C for up to 44 hours. Nanowires were grown both with and without a 4 hour preheat to the growth solution. The resulting ZnO nanowires were found to be too resistive to allow for electrical measurements, and were subjected to a rapid thermal anneal at 800°C for 30 seconds in ambient air, or a prolonged anneal at 300°C lasting a few hours. This typically decreased the resistance of the nanowires by several orders of magnitude.

Once the resistivity of the ZnO nanowires had been lowered, photolithographic

methods were used to define reservoirs and liquid channels for liquid gating measurements. Each reservoir was filled with de-ionised water as the dielectric liquid. All of the ZnO nanowire FETs showed n-type transport, with on-off ratios typically between 10⁴ to 10⁶. The subthreshold slopes for the FETs were typically between 100 - 200 mV/decade. The transconductance was measured on the order of 1 μ S, and the field-effect mobility on the order of 1 cm² V⁻¹ s⁻¹. Threshold voltages were not reported, although must have fallen within a V_G = -1 to +1 V range.

3.3.2 FETs fabricated with individual ZnO nanowires

Kalblein et al.

Kalblein et al. [113] fabricated FETs using individual ZnO nanowires grown using a variation of the hydrothermal method. A 1 cm×1 cm 250 μ m thick square of Zn foil was immersed in an aqueous solution of 480 mM sodium hydroxide and 95 mM ammonium persulfate at 150°C for 48 hours. The resulting nanowires had diameters ranging from 20 to 200 nm and aspect ratios as large as 500. The wires were sonicated in IPA and dispersed on an SiO₂ (200 nm)/Si global back-gated substrate. FETs were fabricated by contacting the ends of the nanowires using electron-beam lithography, and then employing the SiO₂/Si substrate as a dielectric and back-gate.

After depositing metal contacts, the as-grown nanowires were found to be ntype with a very high charge carrier concentration, and consequently unresponsive to gate voltages. Annealing processes were carried out in ambient air at 200°C, 400°C, and 600°C for 15 minute intervals, and the resulting field dependence was investigated. Figure 3.7(a) shows an SEM image of a single ZnO FET transistor, while (b) and (c) show output characteristics and transfer characteristics of an FET fabricated from an as-grown NW. The transfer characteristics of different FETs fabricated from NWs annealed at 200°C, 400°C, and 600°C are shown in (d), while (e) and (f) shows the transfer and output characteristics of an FET fabricated from a NW annealed at 600°C. The FETs annealed at 200°C and 400°C still exhibited metallic electrical behaviour and were not field dependent, while the FET annealed at 600°C exhibited a strong field dependence. This corroborates with theoretical calculations made by Bang et al. [114].

After annealing at 600°C, the FET exhibited an on-off ratio of 10⁷, a threshold voltage of \approx -10 V, a field effect mobility of 40 cm² V⁻¹ s⁻¹, and a subthreshold swing of 400 mV/decade. The transconductance was 0.2 μ S. Even on a conventional Si/SiO₂ substrate with a global back-gate, the hydrothermally grown nanowires exhibit excellent device characteristics that are on par with FETs constructed using single ZnO nanowires grown via high-temperature chemical vapour deposition (CVD) techniques





Figure 3.7: Electrical characteristics of ZnO nanowire FETs in the global back-gate configuration. (a) Atomic force microscopy image, (b) transfer characteristics, and (c) output characteristics of an as-grown ZnO nanowire with Al source and drain contacts. (d) Transfer characteristics of three ZnO nanowire FETs based on nanowires annealed at 200 °C (red), 400 °C (green), and 600 °C (blue) for a drain-source voltage VDS = 1 V. (e) Transfer and (f) output characteristics of a ZnO nanowire FET based on a nanowire annealed at 600 °C in air for 15 min. Figure reproduced from [113].

Having established optimum annealing conditions for the ZnO nanowires, topgated FETs were fabricated by selectively depositing Al contacts onto an annealed nanowire via electron-beam lithography. The dielectric for the top-gate was fabricated by covering the nanowire with a 2 nm thick self-assembled monolayer. After the source and drain contacts were fabricated, the nanowires were exposed to an O_2 plasma. This increases the number of surface hydroxyl groups, which in turn improves the quality of the alkylphosphonic acid self-assembled monolayer (SAM) [117]. After the plasma treatment, the sample was immersed in a solution of 1 mM of octadecylphosphonic acid, forming a self-assembled monolayer which serves as the dielectric. Au was then thermally deposited on top of the ensemble, forming the top gate of the FET. Subsequent electrical measurements showed the FET had an on-off ratio of 10⁷, a threshold voltage of \approx -1 V, and a transconductance of 1 μ S. The subthreshold slope was measured to be as low as 90 mV/decade. The field-effect mobility of the SAM FET was not reported. Kalblein et al.[118] fabricated a similar device in 2014 which coupled the selfassembled monolayer dielectric with a thin aluminium oxide layer. This oxide layer naturally formed from using Al as the contact metal for the top gate. This hybrid top-gate showed improved device performance compared with their previous monolayer-only dielectric, resulting in much lower leakage currents and higher operational voltages. Nanowires were synthesised using their previous method, and once again required annealing at 600°C to reduce the excessive charge carrier concentration.

FETs were fabricated as per the previous publication, with single-nanowire transistors showing on-off ratios of 10^7 , transconductances as high as 7 μ S, and subthreshold slopes of 150 mV/decade. Improvements were made to the FET architecture by arranging the source and drain electrodes in a honeycomb-like array over the nanowire. The best transistor fabricated in this architecture had an on-off ratio of 10^8 , a peak transconductance of 50 μ S, and a subthreshold slope of 100 mV/decade. The threshold voltage was not reported, although must fall within V_G = -1 to 1 V.

Law et al.

Law et al. [109] fabricated an FET using a single hydrothermally synthesised NW as part of their research into hydrothermally grown vertical nanowire arrays for solar cell applications. Although measurements of the FET were not part of the main manuscript, they were included in the supplementary information. ZnO nanowires were synthesised using a 25 mM equimolar $Zn(NO_3)_2/HMT$ aqueous solution, with an additional 5 - 7 mM of PEI to assist NW growth. Subsequent nanowires had aspect ratios up to and greater than 125.

Nanowires were harvested from the growth substrate and distributed onto the SiO₂(300 nm)/Si device substrate via an ethanol suspension, and then annealed at 400°C in air for 30 minutes. An FET was fabricated by selecting a nanowire approximately 8-10 μ m long and depositing Al contacts on the terminating ends using electron beam lithography. The FET had an on/off ratio of 10⁵, a threshold voltage of \approx -50 V, and a field-effect mobility of 1 - 5 cm² V⁻¹ s⁻¹. The transconductance and subthreshold swing of the FET was not reported. Despite the high-temperature annealing step, the very large threshold voltage shows how difficult it is to thoroughly deplete the ZnO nanowire.

Opoku et al.

Opoku et al. explored the relationship between high-temperature annealing and field dependence [54]. ZnO nanowire growth substrates were prepared by coating Si substrates with a thin film of ZnO using the sol-gel method. ZnO nanowires were then grown using equimolar 25 mM Zn(NO₃)₂/HMT along with 5 mM of PEI (M_W = 800 g/mol) at 95 - 100°C for 12 - 15 hours. This sequence was repeated up to 5 times to grow increasingly long nanowires, typically longer than 9 μ m, with diameters usually on the order of 100 nm in diameter.

Nanowires were transferred from the growth substrate to $SiO_2(170 \text{ nm})/Si$ device substrates by sonication and flow-directed assembly. After deposition, the substrates were softbaked at 200°C to ensure that no solvent remained on the substrates. FETs were then fabricated by contacting the nanowires using 200 nm Pt contacts defined via standard electron beam lithography and photolithography.

These FETs showed very little response to applied gate voltages due to an excessive charge carrier concentration, with an on-off ratio of ≈ 0.6 between $V_G = 0$ V and $V_G = -120$ V. These device characteristics were improved by annealing the FET at 550°C in ambient air for 30 minutes. After annealing, the nanowire FETs did show some form of field dependence, with an on-off ratio of 10⁴ and a threshold voltage of -26 V. Despite the 550°C anneal, the devices switched from on to off very gradually, and over a very broad gate voltage range. Measurements of the subthreshold slope were as high as 15 V / decade. The transconductance was measured to be $\approx 0.002 \ \mu$ S, while the field-effect mobility was ≈ 0.13 cm² V⁻¹ s⁻¹. Transfer characteristics of the as-fabricated FETs and post-anneal FETs are shown in Figure 3.8.



Figure 3.8: Transfer characteristics of (a) as-fabricated and (b) annealed FETs. Figure reproduced from [54].

These FETs were significantly improved upon by first annealing the nanowires af-



Figure 3.9: Transfer characteristics of FETs fabricated using nanowires annealed at 350 (black), 450 (red), and 550°C (green). Figure reproduced from [54].

ter transfer to the device substrate, and then contacting the nanowires using electron beam lithography. Al was also used as a contact metal instead of Pd due to its low work function. Nanowires were separately annealed at 350°C, 450°C, and 550°C to investigate between the annealing temperature and device performance. The nanowire annealed at 350°C effectively shows no field dependence, while nanowires annealed at 450°C and 550°C both show field dependence. These devices showed on-off ratios of 10⁵ and 10⁷ respectively, with threshold voltages of -5 and -2 V. The subthreshold swings were 550 - 580 mV/decade, and the field mobilities were 5 and 11 cm² V⁻¹ s⁻¹. Transfer characters of all three FETs are shown in Figure 3.9.

Despite the improved device characteristics achieved after annealing the nanowires, there is clearly some variation between devices fabricated using identical methods. Opoku et al. published a second paper [106] which used a nanowire FET fabricated as per their first publication. The nanowire was annealed at 450°C before Al source-drain contacts were defined and deposited on the nanowire ends. The FET did show clear field dependence, with an on-off ratio as high as 10⁸. The subthreshold swing of the device was $\approx 630 \text{ mV}/\text{decade}$, which is similar to the FETs fabricated in the first publication. However, the threshold voltage was approximately -26 V, which differs significantly from their previous FETs.

Wang et al.

Wang et al. [119] directly compared FETs manufactured with hydrothermally-grown single nanowires and single nanowires grown using high-temperature vapour-phase transport (VPT). The exact details of the hydrothermal synthesis are left unspecified, citing Gao et al. [120] in lieu of specific concentrations of precursors. Therefore the nanowires were most likely synthesised using 100 mL aqueous solutions of equimolar (0.02M) zinc chloride (ZnCl₂) and HMT, with an additional 1-5 mL of ammonia (NH₃).

Hydrothermally grown nanowires were transferred from the growth substrate to the SiO₂(300 nm)/p-Si device substrate via dispersion in an ethanol solution, and subsequently annealed at 550°C in ambient air for 4 hours. A nanowire with a diameter of 200 nm and an aspect ratio of \approx 17 was selected for FET fabrication. Electrodes were patterned on the ends of the single nanowire and a 200 nm thick layer of indium tin oxide (ITO) was deposited using a direct-current sputter coater. After a liftoff process to define the patterned electrodes, the entire sample was subjected to a second anneal at 400°C in ambient air for 30 minutes in order to improve the conductivity of the ITO electrodes.

Subsequent electrical measurements showed the hydrothermal NW FET was ntype, with an on-off ratio of 10^6 , a threshold voltage of -48.5 V, and a field effect mobility of 18.27 cm² V⁻¹ s⁻¹. Neither the transconductance or subthreshold swing of the hydrothermal NW FET were reported. The FET used for comparison features a VPT-grown single NW, the synthesis details of which are cited from the paper of Yang et al. [121]. The VPT FET is fabricated using the same electrode patterning process as used in the fabrication of the hydrothermal FET.

The VPT FET was characterized and found to have an on-off ratio of 10^5 , a threshold voltage of -14V, and a field effect mobility of 36.94 cm² V⁻¹ s⁻¹. The transconductance and subthreshold swing of the VPT FET were not reported. Therefore, while the hydrothermal NW FET had a greater on-off ratio, its threshold voltage and field effect mobility were worse than the VPT NW FET. Wang et al. conclude that the difference in device characteristics between the two different FETs arises from the different defect densities introduced into the ZnO nanowires during their respective synthesis process.

Florica et al.

Florica et al. [122] also compared FETs fabricated using both hydrothermal and high-temperature dry-oxidised nanowires. ZnO nanowires were synthesised hy-



Figure 3.10: Transfer characteristics of as-fabricated hydrothermal (a) and dry-oxidised (b) FETs. Figure adapted from [122].

drothermally using thin films of Ti/Au (10/90 nm) on Si as growth substrates. Growth solutions were prepared using equimolar 0.1 mM Zn(NO₃)₂/HMT at 90 °C inside a convection oven. The solution was preheated for 5 hours before the substrates were introduced. The growth took place over a total of 2 days before the growth substrates were removed and rinsed. Instead of regular VPT methods, the high-temperature nanowires were synthesised by oxidising a zinc foil in a convection oven at 500°C for 12 hours.

Both hydrothermal and high-temperature nanowires were isolated from their respective growth substrates by drop-casting the nanowires onto $SiO_2(50nm)/Si$ device substrates. Standard electron beam lithography and liftoff techniques were used to make Ti/Au (100/300 nm) source and drain contacts on the nanowires. The devices were measured after electrode fabrication, without any annealing or plasma treatments.

Both hydrothermal and dry nanowires showed field dependence when measured, with on-off ratios of approximately $10^4 - 10^5$, and threshold voltages of -4 V and +0.4 V, respectively. The field-effect mobilities were inferred from the transconductance to be 76 and 132 cm² V⁻¹ s⁻¹ respectively, although no value for the transconductance is supplied. The subthreshold swings were 342 mV/decade and 534 mV/decade, respectively. The transfer characteristics of the as-fabricated hydrothermal and dry nanowires are shown in Figure 3.10.

After the FETs were initially measured, the nanowires were covered in a layer of polymethylmethacrylate (PMMA), which acts as a passivation layer. The threshold voltages of the modified hydrothermal and dry FETs reportedly shift by $|\Delta V_{DS}| = 3$



Figure 3.11: Transfer characteristics of PMMA-passivated hydrothermal (a) and dryoxidised (b) FETs. Figure adapted from [122].

V and 1.2 V, respectively. The field-effect mobilities are reported as 87 cm² V⁻¹ s⁻¹ and 167 cm² V⁻¹ s⁻¹ post-passivation, respectively. The subthreshold swings are also improved, with post-passivation values of 269 mV/decade and 431 mV/decade, respectively. Transfer characteristics for both hydrothermal and dry nanowire FETs post-passivation are shown in Figure 3.11.

Summary of fabricated devices

The main FET characteristics of the papers presented above are summarized in Table 3.1.

Device geometry effects

All of the ZnO nanowire FETs which have been studied here are based on laterallyaligned nanowires, which ostensibly lie flush with the substrate. These nanowires are then used as FETs in conjunction with a top or bottom gate dielectric, which is usually considered to be flush and in contact with the nanowire. However, in some cases this is a simplification, as the nanowire can be separated from the gate dielectric by a small air gap. This can especially be a problem for FETs fabricated using arrays of lateral nanowires, where nanowires grow out of seeded sidewalls at angles.

ZnO nanowire FETs have previously been intentionally fabricated with air gaps between the channel and the dielectric, which provide information on how a lessthan-flush geometry can affect the device performance. In general, nanowire FETs with air gaps have higher threshold voltages than those lying flush with the substrate dielectric [123]. This is due to the depleting effect of adsorbed oxygen and hydroxyl groups on the nanowire surface; when the nanowire is separated from the dielectric by an air gap, oxygen groups are able to adsorp to entire nanowire surface, depleting them of electrons and requiring higher threshold voltages to populate the channel [123]. In comparison, surface states at the nanowire/dielectric interface can contain trapped positive charges, which locally affect the voltage and reduce the depletion width within the nanowire.

Devices with air gaps can also suffer from reduced gate effectiveness due to the additional distance between the nanowire and the gate electrode [124, 35], and epoxy or other filling layers can be required to reduce the impact of the air gap [124]. However, separating the nanowire from the dielectric layer has the advantage of reducing leakage current, meaning that low off-currents can be achieved in air-gap devices [35].

As such, the presence of air gaps in ZnO nanowire FETs can affect the device performance. Furthermore, if the FETs are fabricated by bridging lateral arrays of nanowires, then possible air gaps in the device geometry can be difficult to control, seeing as they are determined by the angle of nanowire growth from the seeded substrate. These imperfections can add a level of variation between fabricated transistors, which may be present in the devices reviewed in this chapter and presented in this thesis.

| Table 3.1 | : Summary of the synthesis and | device characte | ristics of the nar | nowire FETs pre | sented in this |
|-----------------|--|-----------------|---------------------------|-----------------------------|----------------|
| Authors | Precursors | # of NWs | Gate | V_{Th} (V) | |
| Park et al. | $10 \text{ mM Zn}(\text{NO}_3)_2/\text{HMT}$ | Many | SiO ₂ (300 nm) | ≈-35 | |
| Ko et al. | 25 mM Zn(NO ₃) ₂ /HMT | Many | SiO_2 | ц'n | |
| Pachauri et al. | $0.25 \text{ mM Zn}(NO_3)_2/HMT$ | Many | $DI H_2O$ | -1 $\leq V_{\rm Th} \leq 1$ | |
| Kalblain at al | 480 mM KOH | - | 7 nm C AM | $1 < V_{-} < 1$ | |
| | 95 mM ammonium persulfate | Ŧ | | т — ч.Т v — т- | |
| | 25 mM Zn(NO ₃) ₂ /HMT | - | CiO (300 nm) | L L | |
| Law el al. | 7 mM PEI | Т | | 00- | |
| | 25 mM Zn(NO ₃) ₂ /HMT | Ţ | C:O (170) | c | |
| Ороки ег аг. | 5 mM PEI | Т | | 7- | |
| Wone of al | 25 mM Zn(NO ₃) ₂ /HMT | - | CiO (300 nm) | и Х | |
| walig el al. | $1 - 5 \text{ mL NH}_3$ | Ţ | | 0.01- | |
| Florica et al. | $0.1 \text{ mM Zn}(NO_3)_2/HMT$ | 1 | $SiO_2(50 \text{ nm})$ | -4 | |
| | | | | | |

| Authors | On-off ratio | SS (mV/dec) | g_m (nS) | $\mu_{ m eff}$ | Anneal |
|-----------------|-----------------------------------|-------------|------------|----------------|--|
| Park et al. | $pprox\!4\!	imes\!10^5$ | \geq 2000 | 400 | 8.5 | $450~^\circ\text{C}$ in N_2 for 2 min |
| Ko et al. | $pprox 10^4$ | 8000 | 100 | | n/a |
| Pachauri et al. | 10 ⁴ - 10 ⁶ | 100 - 200 | 1000 | 1 | 800 °C in air for 30s |
| Kalblein et al. | 10 ⁷ | 06 | 1000 | | 600 °C in air for 15 min |
| Law et al. | 10^{5} | | | 1 - 5 | 400 °C in air for 30 min |
| Opoku et al. | 10^{7} | 580 | | 11 | 600 °C in air for 15 min |
| Wang et al. | 10 ⁶ | | | 18.27 | 550 °C in air for 4 hours |
| Florica et al. | 10^{4} | 342 | | 76 | n/a |

is chapter.

3.4 ZnO nanowire doping and annealing

3.4.1 n-type ZnO nanowires

It is a well-established fact that the vast majority of as-grown ZnO is doped with shallow donors, giving rise to a strongly n-type behaviour [125]. The high charge carrier concentrations can be beneficial in certain applications, such as those needing transparent conductors. For FET applications, it is desirable to use high-quality intrinsic ZnO with fewer charge carriers, as this gives rise to a stronger field dependence, a threshold voltage closer to 0 V, a high on-off ratio, and better device performance overall. As is evident in the literature being studied here, as-grown hydrothermally-synthesised ZnO NWs are not necessarily suitable for FET applications due to excessively high charge carrier concentrations.

It is difficult to determine the exact charge carrier concentration where ZnO nanowires become difficult to use in FETs. However, an order-of-magnitude approximation can be estimated by considering the doping concentration required to degenerately dope the nanowires. In this regime, the distance between electron donors is sufficiently small enough for their wavefunctions to overlap, forming a defect band which can overlap with the conduction band. This means that the Fermi level constantly lies within the defect band, even at very low temperatures. The charge carrier concentration required to achieve degenerate doping in a semiconductor can approximated by setting it equal to the effective density of states in the conduction band. This can be estimated as [38]:

$$n_{\rm eff} = 2 \left(\frac{2\pi m_{\rm eff} k_{\rm B} T}{h^2}\right)^{\frac{3}{2}}$$
(3.2)

where m_{eff} is the effective mass of the electron, k_B is Boltzmann's constant, T is the temperature, and h is Planck's constant. Taking m_{eff} to be approximately 0.3 m_0 [12] and T to be 293.15 K (corresponding to 20°C), the degenerately doped carrier concentration is calculated to be $\approx 4.2 \times 10^{18}$ cm⁻³.

The necessity of a high-temperature anneal to address the excessive doping in ZnO nanowire devices negates some of the inherent advantages of the hydrothermal synthesis technique, such as the ability to grow in-situ laterally-aligned nanowire arrays on vulnerable and delicate substrates that would be destroyed in the process of VPT synthesis methods. Therefore, there is a strong motivation to unequivocally determine the cause of the naturally n-type behaviour in ZnO nanowires.

As described in the introduction to this thesis, the origin of doping in ZnO has

been a matter of intense theoretical work over the past decade. Initially attributed to intrinsic point defects [125] such as Zn interstitials and O vacancies, other theoretical treatments by Janotti et al. [126] have suggested that intrinsic point defects cannot explain the n-type behaviour of ZnO. Building on this theoretical framework, hydrogen and its interactions with native point defects has been suggested as the main shallow donor leading to n-type behaviour [127, 25, 128, 114]. The exact nature of the dopants is still a point of contention however, with continued theoretical research suggesting that native defects are responsible for the n-type behaviour [129]. An in-depth discussion of the controversial n-type doping is outside of the scope of this thesis, and has been reviewed comprehensively in the past [130].

Part of the difficulty in determining what exactly causes the doping in ZnO is the fact that theoretical predictions of what annealing temperatures are required to purge suspected dopants from the bulk are not always mutually exclusive. Bang et al. [114] have used first principals calculations, in conjunction with kinetic Monte-Carlo simulations, to model the diffusion and stability of hydrogen in ZnO. Predictions based on the simulations are displayed in Figure 3.12, with Figure 3.12(a) displaying the relationship between hydrogen diffusion and anneal temperatures, and Figure 3.12(b) displaying the relationship between defect concentration and time at room temperature.



Figure 3.12: The results of kinetic Monte Carlo simulations for (a) the H diffusion at different annealing temperatures and (b) the evolution of injected H atoms at room temperature. Lines represent the average of simulations up to five, and points at the concentration of 10¹⁴ cm⁻³ correspond to nearly zero concentration in simulations. Figure reproduced from [114].

Systematic research into the annealing response of VPT-grown ZnO nanowires for FET applications has been conducted by Jin Jeon et al. [110]. VPT-grown ZnO nanowires were dispersed onto an $SiO_2(200nm)/p$ -Si device substrate. Several devices were made in both global back-gated geometries, and top-gated geometries utilising



Figure 3.13: Transfer curves measured at V_{DS} =1V from the bottom-gate FETs with ZnO NWs prepared under various conditions: un-annealed (black line), air-ambient annealed at 400°C for 15min (red line), air-ambient annealed at 600 °C for 15min (green line), and at 600°C for 30min (blue line). (b) Transfer curves measured at V_{DS} =1V from the top-gate FETs with ZnO nanowires that were annealed at 600°C. Figure reproduced from [110].

a 30 nm Al_2O_3 layer as the dielectric. Individual back-gated devices were annealed at 400°C for 15 minutes, 600°C for 15 minutes, and 600°C for 30 minutes, while individual top-gated devices were annealed at 600°C for 15 minutes, and 600°C for 30 minutes. An unannealed back-gated device was also measured to serve as a control.

After annealing, the individual devices had their transfer characteristics measured. The transfer characteristics of the different back-gated FETs are displayed in Figure 3.13(a). The unannealed nanowire exhibits very low field dependence, making it unfit for use in an FET. Significantly higher field dependence is observed in the back-gated device annealed at 400°C for 15 minutes, achieving an on-off ratio of 10^4 , but the associated threshold voltage (\approx -65 V) makes the device unsuitable for FET applications. The back-gated device annealed at 600°C for 15 minutes shows improved device characteristics, with an on-off ratio of 10^5 and a threshold voltage of -47 V. Finally, the back-gated device annealed at 600°C for 30 minutes shows an even more pronounced improvement in device characteristics, achieving an on-off ratio of 10^6 and a threshold voltage of -8 V.

The top-gated devices show a similar trend, although for any given annealing condition the top-gated devices exhibited better overall device characteristics. 15 minute and 30 minute anneals at 600°C yielded on-off ratios greater than 10⁶ and threshold voltages of -12 V and -3 V respectively. The transfer characteristics of the top-gated FETs are displayed in Figure 3.13(b).
While Jin Jeon et al. attribute this improvement of device characteristics with increasing annealing temperatures and times to a reduction of oxygen defects on the surface of the nanowires, the results of their study agree well with Bang et al.'s model of hydrogen defects and thermal stability. Bang et al.'s model also explains why Kalblein et al.'s FET does not exhibit a field dependence at 200°C and 400°C, and a pronounced field dependence at 600°C. Jin Jeon et al.'s back-gated FETs exhibit field dependence after being annealed at 400°C for 15 minutes, while Kalblein et al.'s FET does not exhibit a field dependence at 400°C. However, Jin Jeon et al. sweep their gate voltage to -70 V, and only observe depletion at -65V, while Kalblein et al. only sweep their gate voltage to -40V.

Although the vast majority of ZnO nanowire FETs exhibit field dependence only after annealing, there are a number of reported FETs fabricated with VPT synthesised ZnO nanowires that show field dependence before annealing [131, 132]. The reason why ZnO nanowires exhibit field dependence without the need of an anneal is currently unknown, and should warrant great attention. It may be that many of the VPT methods in the literature that take place at high temperatures actually incorporate the necessary anneal stage post-growth. For example, in a paper by Sohn et al. [131], NWs are grown at 900°C, and subsequent FETs utilizing said NWs require no annealing step for good device performance. However, in the above mentioned paper by Jeon et al. [110], the NW synthesis temperature is lower, at 760°C.

3.5 Hysteresis and passivation in ZnO nanowire FETs

Transfer characteristics of FETs are acquired by fixing the drain-source voltage at a given value while sweeping the gate voltage, as described in the beginning of this chapter. The gate voltage can be swept towards the positive or negative direction, although comprehensive transfer characteristic measurements sweep the gate voltage in both directions. These measurements can reveal a discrepancy in the drain-source current at a given gate voltage depending on the direction of the gate voltage sweep, which is known as hysteresis. Hysteresis has been observed in many FETs which use nanomaterials, including nanowire devices such as carbon nanotubes [133] and germanium nanowires [134], and 2D semiconducting materials, such as graphene [135] and molybdenum disulphide [136].

Hysteresis has also been recorded in ZnO nanowire FETs, although comprehensive transfer characteristic measurements of hydrothermally synthesised nanowires are rarely reported, and most measurements are taken by sweeping the gate voltage only in a single direction. For example, of the results reproduced in this chapter, only Kalblein et al [113] (shown in Figure 3.7) and Jin Jeon et al [110] (shown in Figure 3.13) show transfer characteristic measurements comprehensive enough to gauge the hysteresis of the fabricated devices. Nevertheless, the literature on hysteresis and passivation in ZnO FETs is worth examining in the context of this thesis, as some of the transistors fabricated in later chapters show hysteretic behaviour.

Hysteresis in nanoelectronics is typically attributed to charges trapped or otherwise present in the vicinity of the semiconducting channel, which alter the local voltage and the band bending of the device. Hysteresis is desirable for some applications, and several publications have investigated using hysteretic ZnO nanowires for memory storage. These applications benefit from increasing the hysteresis through introducing or increasing charge trapping sites such as Al₂O₃ layers deposited directly onto the nanowire [137], ferroelectric liquid crystals [138] or polymers [139], ferroelectric nanoparticles deposited on the nanowire surface [131], or SiO₂ back-gate dielectrics treated by high-pressure hydrogen annealing [140].

Although these measures all increase the hysteresis of the FETs, many ZnO nanowires show an inherent hysteresis that is typically attributed to intrinsic charge traps on the nanowire surface [141, 142], or the nanowire/dielectric interface [141, 143, 144]. The surface states of the nanowire are thought to be electron-trapping oxygen or hydroxyl groups which are chemisorbed to the nanowire surface [141, 142], while the local charges located at the nanowire/dielectric interface have been attributed to adsorbed oxygen on the surface [143], mobile protons present in the substrate [140], or mobile negative charge carriers [137]. A definitive single cause of hysteresis likely does not exist, and ZnO nanowire FETs may be affected by several different charge trapping mechanisms simultaneously.

Hysteretic behaviour can cause serious problems for FETs, as the device performance can significantly change over time in unpredictable ways. Hysteresis driven by interactions between the ZnO nanowire surface and measurement atmosphere can cause transient effects in FETs [145], and the hysteresis has been shown to depend on the range and rate of the gate voltage sweep [142, 146]. Passivation techniques can be used to reduce the hysteresis, which typically involves encapsulating the ZnO nanowires in an inert material, preventing reactions with the atmosphere at the nanowire surface [141, 147, 148]. However, passivation doesn't always prevent hysteresis, as charge traps can remain at the nanowire/substrate interface, and interactions between the passivation layer and atmosphere can drive further hysteresis [137, 147, 142, 146]. Further reduction in hysteresis can be achieved by annealing ZnO nanowire devices at high temperatures and in different atmospheres. The work by Kalblein et al [113] and Jin Jeon et al [110] reproduced in this chapter clearly demonstrates the reduction in hysteresis with high-temperature annealing. This is attributed to a passivation or removal of defect states, which is also the same mechanism relating field dependence to high-temperature annealing. Other research has combined passivation layers with annealing to minimise hysteresis [144] or annealing in specific atmospheres [143].

In summary, hysteresis has been documented in ZnO nanowire FETs, similar to other nanomaterials used for FET applications. Despite this, the exact causes and origins of the hysteresis vary, and the hysteresis of a given device is likely compounded by several types of charge traps located on the nanowire surface and nanowire/substrate interface. Passivation techniques involve encapsulating the nanowire to remove nanowire/atmosphere interactions, although this does not necessarily compensate for charge traps at the nanowire/substrate interface. High-temperature annealing can also be used to expel and passivate charge traps in the device, reducing the overall hysteresis.

3.6 Conclusion

In conclusion, the fundamental device parameters used to characterise FETs have been described. While there are several publications which use hydrothermally-synthesised ZnO nanowires as FETs, the majority of these devices require high-temperature annealing at or above 400°C in order to show field dependence. In some cases, the nanowires show poor device characteristics even after being annealed, requiring high threshold voltages (up to -50 V) to sufficiently deplete the channel, or very high subthreshold swings (up to 8 V/dec.) The need to anneal devices before field dependence is likely due to excessive charge carrier concentrations in the ZnO nanowires, which are caused by unintentionally incorporated dopants. The exact nature of these dopants is still uncertain, although theoretical models implicating interstitial hydrogen or oxygen defects on the nanowire surface both agree with the relationship between field dependence and annealing which is recorded in the literature.

As such, there is a strong motivation to reliably fabricate FETs which show field dependence without requiring high-temperature annealing. The necessity of the high-temperature anneal strongly restricts the choice of substrates that the devices can be fabricated on. ZnO is otherwise well-suited for emerging nanoelectronic applications which are based around flexible, transparent, and disposable substrates,

but their full potential will not be realised until this impasse has been addressed.

Chapter 4

Experimental details

4.1 Introduction

This chapter details the experimental methods and processes used throughout the work presented in this thesis. It aims to describe the various pieces of equipment and the way in which they are used sufficiently enough for the interested reader to reproduce the experimental results reported in this work.

4.2 Fabrication of substrates for lateral nanowire growth

In order to use ZnO nanowires as field-effect transistors, the nanowires must be in contact with (or very close to) a dielectric layer, which itself is in contact with a gate electrode. One common approach is to use a heavily doped silicon wafer as the device substrate, and to consequently use the entire substrate itself as a gate electrode. In order to electrically isolate the substrate from the nanowires and source/drain electrodes, the silicon is thermally oxidised to produce a thin layer of SiO₂ on the surface.

SiO₂ (100 nm) / p⁺-Si substrates (sourced from Silicon Quest International) were used as device substrates for lateral nanowire FETs. The wafers are 3" in diameter, and must be cleaved into smaller pieces to be used for device fabrication. As this cleaving process physically breaks the wafer into pieces, care must be taken not to ruin the thin SiO₂ dielectric on the wafer surface, or the device will leak and be unusable. To protect the wafer as much as possible, we first use a spin-coater to deposit a sacrificial layer of AZ1518 (Microchem) on the uncleaved wafers at 900 RPM for 60 seconds. AZ1518 is a photosensitive resin used for photolithography, although it is used here simply as a thin layer of material (on the order of 5 μ m) that is soluble in acetone. The wafer is then baked on a hotplate at 95°C for a few minutes to expunge the AZ1518 of excess solvent.

We then spin a layer of polydimethylsiloxane (PDMS) at 600 RPM for 60 seconds to add a further level of protection to the wafer. PDMS is a robust organic two-part polymer made by mixing an elastomer curing agent with a silicone base (Sylgard 184 Incor) at a 1:10 mass ratio. The thickness of the PDMS layer on the wafer is on the order of hundreds of microns. The wafers are placed on a hotplate at 95°C for 15 minutes after the PDMS is spun on, which speeds up the curing process.

Although the p⁺-Si wafers are highly doped and have a very low resistivity, it can be difficult to make good electrical contact to them as their back surfaces are unpolished. To ensure good contact to the back gate, the backs of the protected wafers are coated with Cr/Au (5/50 nm) via resistive resistive evaporation using our in-house Angstrom Engineering NexDep evaporator. The Cr serves as an adhesion layer for the Au, which is used as a contact metal because it doesn't corrode and has a high conductivity.

After the backs of the protected wafers have been metallised, they are cleaved into 1.1×1.1 cm squares using a diamond scribe and a metal ruler. The combined photoresist/PDMS layers confines the dust created when cleaving the wafer into pieces, leaving the SiO₂ surface largely free of particulates. After the wafer is cleaved, the individual squares are peeled away from the PDMS layer and placed in acetone, which dissolves the AZ1518 protective layer. The substrates are then individually sonicated in acetone and then IPA, which removes any residual contaminants or debris.

4.2.1 Photolithography

The fabrication of lateral ZnO nanowires relies on patterning electrodes/seed layers at specific locations on the substrate. We use optical photolithography to define our electrodes, which is a very common technique used in microfabrication. A photosensitive resin (known as a photoresist) is applied to the surface of the substrate via spin-coating. When the photoresist is exposed to ultraviolet (UV) light, it becomes soluble (or insoluble, depending on the specific photoresist,) to a chemical called a developer. In order to expose only specific parts of the photoresist to UV light, the UV light is filtered through a mask, which is typically a chrome stencil on a plate of soda lime glass.

After exposure, the substrates are washed in developer which dissolves and removes the unwanted parts of the photoresist. Once metal or other thin films have been



Figure 4.1: The photolithography process. (a) SiO_2 (100 nm)/Si substrates are spincoated with AZ1518 photoresist. (b) Portions of the photoresist are selectively exposed to UV light using a mask aligner. (c) The substrates are developed, which dissolves the exposed photoresist. (d) Material (in this case, ZnO/Ti) is deposited onto the patterned substrates. (e) The substrates are lifted off, removing excess material.

deposited on the photolithographically patterned samples, the remaining photoresist is removed by submerging the substrates in acetone (otherwise known as a liftoff.) This dissolves the remaining photoresist and removes the deposited material that sits on its surface. The deposited material remains adhered to the parts of the substrate which were defined in the initial exposure and development stage. The entire photolithography process is depicted schematically in Figure 4.1.

All of the photolithography undertaken in this thesis was completed using a Karl Suss MJB3 mask aligner, equipped with an USHIO SHP 350W lamp. The masks used for fabricating lateral ZnO nanowires were custom-made at the University of Canterbury, using a Heidelberg μ PG 101 mask writer. The mask .GDS files were designed using LayoutEditor. A schematic of the lateral nanowire mask is shown in Figure 4.2, with 30 - 50 μ m gaps between pairs of electrodes. We have also made use of a mask with 8 pairs of electrodes, with gaps varying from 10 to 80 μ m (not pictured.)

AZ1518 was spun on cleaned and cleaved wafers at 4000 RPM for 60 seconds. The substrates were then soft-baked at 95°Cs on a hotplate to remove residual solvent from the AZ1518 layer. AZ1518 is a positive-tone photoresist, meaning that it is insoluble in developer by default, and becomes soluble after UV exposure. The substrates are exposed for 10 seconds using the mask aligner, before being developed in AZ326 (Microchem), a common and widely-used developer which uses tetramethylammonium



Figure 4.2: A schematic of the mask used to photolithographically define substrates for lateral ZnO nanowire growth. The size of the channel gaps is indicated at the bottom of the schematic. The array of squares on each corner of the schematic are used for alignment purposes.

hydroxide (TMAH) as its active ingredient. The AZ326 was diluted at a 3:1 AZ326:DI H_2O ratio before being used. Substrates were developed for 40 seconds and rinsed twice in DI H_2O to remove any residual scum.

4.2.2 Seed layer/electrode deposition

After the substrates have been patterned using photolithography, thin films of ZnO and Ti are deposited on the substrates using radio-frequency (RF) sputter coating. Sputter coating is a process where a 'target', or source, is bombarded with energetic gas molecules. The gas molecules are ionised by applying a high-voltage bias between the target and a cathode, which forms a plasma. The plasma is localised and confined near the surface of the target by strong electric and magnetic fields generated by a magnetron, which sits underneath the target. The applied voltage between the target and the anode causes the ionised gas molecules to ballistically collide with the target, which consequently ejects atoms. These ejected atoms make their way to the chuck via random walks, eventually depositing on the surface of the substrate.

The cleanroom at VUW is equipped with an HHV Auto500 sputter coater, coupled with an Advanced Energy Cesar 136 600W RF generator. The Auto500 has 3 separate sources, which allows us to deposit three separate materials without venting the chamber. Instrument grade argon (BOC, Ar >99.99%) is used as the processing gas for all of the sputter coating referenced in this thesis. The Auto500 chamber is



Figure 4.3: (a) A top-down optical image of a lateral substrate. The ZnO/Ti electrodes are clearly defined. (b) A cross-section SEM image of the lateral substrate.

evacuated with an initial roughing pump coupled with a turbomolecular pump.

After the substrates have been patterned using photolithography, the Auto500 sputter coater is used to deposit 100 nm of ZnO (target: 99.99%, Plasmaterials), which serves as the seed layer for nanowire nucleation during the hydrothermal growth. The initial ZnO deposition is followed by a capping layer of Ti (target: 99.99%, Plasmaterials), which can be deposited without venting the chamber due to the multiple sources in the Auto500. The Ti layer covers the surface of the ZnO seed, leaving only the sidewall of the ZnO layer exposed once the excess material is lifted off. It also serves as a contact material for source/drain electrodes. The role of the ZnO seed layer in the hydrothermal growth process is discussed in detail in Chapter 2.

After the Ti layer has been deposited, the substrates are lifted off in a bath of acetone to dissolve the photoresist pattern. This removes the excess material deposited on the photoresist surface, leaving ZnO/Ti on the areas of the substrate where the photoresist was dissolved away during the development step. Sometimes the liftoff process leaves small amounts of residual material at the edges of the patterns, which are removed using brief sonication in acetone or IPA. The substrates are then ready for hydrothermal growth.

Figure 4.3(a) shows an optical microscope image of a substrate post-liftoff. Figure 4.3(b) shows a cross-sectional image taken using SEM of a substrate used for lateral nanowire growth. The Si, SiO_2 , ZnO, and Ti layers can be individually be seen on the substrate.

4.3 Fabrication of substrates for vertical nanowire growth

The fabrication of substrates for vertical nanowire growth is significantly simpler than fabricating substrates for lateral nanowire growth. A p^+ -Si substrate (Silicon Quest international) is cleaved into 1.1 cm × 1.1 cm squares, which are consequently cleaned via sonication in acetone and IPA. Because the substrate isn't used for any electrical measurements, the wafers do not require a dielectric layer of SiO₂ on the surface, or any photoresist/PDMS protective layers. Similarly, the exact dopants of the Si wafer do not matter, although we have used p^+ -Si wafers out of convenience.

After the substrates have been cleaved, 100 nm of ZnO is deposited via sputter coating, as described in Section 4.2.2 of this chapter. The ZnO film uniformly covers the substrates, as we don't require any photolithographic patterning of the seed layer. After the deposition, substrates coated with the ZnO seed layers are removed from the sputter coater and are ready for use.

4.4 Electrical measurements

Electrical measurements of all the devices contained in this thesis are performed using an Agilent 4156C parameter analyser in conjunction with a Rucker and Kolls probe station. Titanium tri-axial probes are used to make contact to the photolithographically defined pads. Data is retrieved from the parameter analyser using a laptop with an in-house developed Python script. All transfer characteristic measurements involving gate voltage sweeps go from negative to positive to negative, unless otherwise stated. The gate voltage is typically changed in 10 mV increments. Device parameters are extracted from transfer characteristics taken on the upscan or downscan depending on the mode of FET operation. Similarly, the drain-source voltage in output characteristic or diode measurements is swept from negative to positive values. These are also typically measured in 10 mV increments, although diode measurements in Chapter 8 are performed using 1 mV increments.

The rate at which the voltage is swept is determined by the parameter analyser, using a combination of the measurement mode and the magnitude of the drain-source current being measured. For reference, all electrical measurements presented in this thesis are taken in the "Long" measurement mode. The time taken to acquire a measurement of the drain-source current at a specific voltage ranges from 2 seconds (when I_{DS} is in the range of 10 pA) to 0.02 seconds (when I_{DS} is in the range of 10 nA

to 100 mA.) Details of exactly how the integration time is determined can be found in the Agilent 4155C/4156C User's Guide [149].

Similarly, the uncertainties in the applied voltages and measured currents depend on the current/voltage regime used by the parameter analyser. Applied voltages within +/- 2 V have an output resolution of 100 μ V, while applied voltages within +/- 20 V have an output resolution of 1 mV, which applies to the majority of measurements presented in this thesis. Applied voltages within +/- 40 V and +/- 100 V have a 2 mV and 5 mV output resolution, respectively.

Uncertainties in the currents measured by the parameter analyser depend both on the magnitude of the current and the integration time, and the relationship is nontrivial. Uncertainties in the measured currents presented in thesis are shown in Table 4.1. These uncertainties only represent the measurement resolution of the parameter analyser, and do not account for other uncertainties which may be introduced by other components of the measurement set-up (such as the capacitive effects of the tri-axial cables, probe/contact effects, etc.) A full and detailed relationship between the measurement resolution, the current range, and the integration time can be found in the Agilent 4155C/4156C User's Guide [149]

| | * | |
|-------------------|------------------------|--|
| Measurement range | Measurement resolution | |
| 10 pA | 1 fA | |
| 100 pA | 1 fA | |
| 1 nA | 10 fA | |
| 10 nA | 10 fA | |
| 100 nA | 100 fA | |
| $1\mu\mathrm{A}$ | 1 pA | |

Table 4.1: Current measurement resolution of the 4156C parameter analyser

Similarly, uncertainties in FET device characteristics extracted from electrical measurements - such as the threshold voltage, transconductance, subthreshold-swing, and electron mobility - are often overlooked in literature, and uncertainty values are very rarely provided. Uncertainties of device parameters are first determined by uncertainties in the electrical measurements, and then compounded by uncertainties in their processing.

For example, threshold voltages are determined by taking the intercept of the linear portion of the drain-source current, and this intercept can be influenced by uncertainties in the drain-source current itself. Similarly, the transconductance is defined

as the maximum value of the derivative of the transfer characteristic measurements. Small amounts of noise in the transfer characteristics can be clearly identified as spurious data, but these small spikes can have very large derivatives. This leads to a certain level of subjectiveness in the definition of the transconductance, as spurious values may have to be discarded by hand. These problems lead to a certain level of uncertainty in the device parameters which is very difficult to accurately quantify.

As such, no specific uncertainties in the device parameters are provided in this thesis, which is in line with published literature in the field. However, some level of uncertainties in these values must be acknowledged. In lieu of specific errors for each value presented in this thesis, it would be prudent to consider the values as having an uncertainty on the order of $\pm -5\%$.

4.5 Scanning electron microscopy

All scanning electron microscopy (SEM) images that are presented in this thesis are taken from a JEOL 6500F at Victoria University. The exact accelerating voltage and other specific parameters vary between images, and are not presented here.

4.6 Transmission electron microscopy

Transmission electron microscopy measurements are presented in Chapters 6 and 7. These measurements were performed by Dr. Jerome Majimel and Uli Castanet at ICMBC Bordeaux as part of a collaboration. Primary and secondary ZnO nanowires were collected from growth substrates by using a scalpel, and then suspended in alcohol by ultrasonication. Drops of the suspension were then deposited onto a copper grid covered with a carbon film. This grid was finally air-dried for 15 minutes. TEM and high resolution TEM observations were performed using a JEOL 2200 FS equipped with a field emission gun, operating at 200 kV and with a point resolution of 0.23 nm. High-resolution transmission electron microscopy micrographs were acquired with a Gatan Ultrascan CCD 2k - 2k and digital diffractograms were calculated using the Gatan Digital Micrograph program. In order to be representative and statistically meaningful, many images from several regions of various samples were recorded, and the most characteristic results are presented here. Due to the thickness of the nanorods, high-resolution TEM observations were almost exclusively performed on the nanorod terminating ends. Only Electron Diffraction (ED) patterns have been recorded all along the nanorods.

Chapter 5

Hydrothermal growth of vertical ZnO nanowires

5.1 Introduction

ZnO nanowires can be easily and reproducibly grown en masse from seeded substrates using the hydrothermal method [48]. The simplicity and scalability of this synthesis route offers significant advantages over comparatively difficult or expensive nanowire synthesis techniques [10]. As described in Chapter 2, nanowires grow approximately normal to the plane of the seeded substrate, giving rise to dense vertical arrays of nanowires from flat substrates. Vertical arrays of ZnO nanowires have already been used in a wide variety of opto-electronic applications, such as solar cells and piezoelectronic power generators [10, 18, 109, 150, 151, 121, 152, 132]. Vertical nanowires can also be removed from the growth substrate, and then used individually for device fabrication.

Whether used individually or as arrays, the nanowire morphology can have a significant impact on device performance. High aspect ratios (the length of the nanowire divided by its width) are often desirable, which places an emphasis on synthesising longer ZnO nanowires without a corresponding increase in diameter. To this end, polyethylenimine (PEI) is a polymer commonly used as an aspect-ratio enhancer in the hydrothermal growth [68, 69, 70, 71, 72, 73, 74], as outlined in Chapter 2. While there are published reports which focus on optimising the PEI concentration to give the longest nanowires [68, 69, 74], there has been comparatively little research on how the molecular weight of the polymer might affect the hydrothermal synthesis.

In this chapter, the effects of using different concentrations and molecular weights of PEI ($M_W = 800$, 1300, and 2000 g/mol) on the morphology of the nanowires is

investigated. These results are contrasted with a control sample, grown without any PEI. The inclusion of PEI can drastically alter the final hydrothermal product, either etching the growth substrate, depositing a homogeneous thin film of ZnO, growing hierarchical ZnO nanowires, or resulting in homogeneous nanowires. Growth mechanisms for ZnO thin films and for hierarchical ZnO nanowires are put forward. Finally, the length, diameter, and aspect ratios of the homogeneous nanowires are compared. All of the PEI-assisted nanowires are longer than the control nanowires, and also have better aspect ratios. The longest nanowires are grown using 8 mM of PEI($M_W = 800 \text{ g/mol}$), while the highest aspect-ratio nanowires are grown using 6 mM of PEI($M_W = 800 \text{ g/mol}$). These results give valuable information on optimising vertical ZnO nanowire growth for the fabrication of individual field-effect transistors, and the proposed growth mechanism for the hierarchical nanowires is important for the results presented in later chapters.

5.2 Experimental details

Growth substrates were first fabricated by cleaning and cleaving p^+ -Si wafers (sourced from Silicon Quest International) into 1 cm × 1 cm squares. After cleaving, the growth substrates were covered in a 100 nm uniform thin film of ZnO via RF sputtering, to serve as the seed layer for hydrothermal growth of vertical nanowires. The substrates were annealed at 500 °C for 60 minutes in ambient atmosphere using a hotplate. Annealing RF-sputtered thin films of ZnO improves their crystallinity, leading to longer nanowires after the hydrothermal growth [153, 154, 155, 156]. Thorough experimental details for the substrate fabrication are provided in Chapter 4.

In order to independently determine how the concentration and molecular weight of included PEI affects the hydrothermally grown nanowires, all of the other hydrothermal growth variables were kept constant. To this end, each hydrothermal growth was prepared by first making a control solution using 25 mM of $Zn(NO_3)_2$ (Sigma Aldrich, 98% purity) and hexamethylenetetramine (HMT) (Sigma Aldrich, 99% purity) mixed with 250 mL de-ionised (DI) water ($\geq 18.2 \text{ M} \Omega \cdot \text{cm}$). These control solutions were then mixed with PEI to complete the hydrothermal growth solution. Three different varieties of PEI were used, with molecular weights of 800, 1300, and 2000 g/mol, which are referred to as PEI(800), PEI(1300), and PEI(2000), respectively. Each PEI variety was used at a concentration of 2, 4, 6, and 8 mM. Three separate growth substrates were included in each hydrothermal growth to ensure that the results were reproducible.

After the hydrothermal growth solutions were thoroughly mixed together in

borosilicate glass bottles, they were placed in a waterbath held at 95 °C to preheat for one hour. After the preheat, growth substrates were placed into each hydrothermal growth solution and allowed to grow for 19 hours at 95 °C. The general theory and experimental techniques underpinning the hydrothermal growth of ZnO nanowires is described in detail in Chapter 2. After the 19 hour growth was complete, the substrates were removed from the growth solution and rinsed thoroughly in DI water (\geq 18.2 M $\Omega \cdot$ cm) before being dried with nitrogen at room temperature.

Finally, the hydrothermally grown ZnO nanowires were imaged using a JEOL6500F SEM, also described in Chapter 4. Images were taken at three different locations on each sample to ensure that the imaged nanowires were representative. These locations were intentionally chosen to span the entire length of each sample in cross-section, in order to account for any possible localised variation in the nanowire morphology. The length and diameter of the imaged nanowires were measured using ImageJ. Selection bias was avoided by measuring the dimensions of every nanowire in each SEM image. The uncertainties in these measurements are compounded by the uncertainties introduced by the JEOL6500F, the resolution of the SEM images (as ImageJ measures dimensions on a pixel-by-pixel basis), and human error in using ImageJ. The uncertainties in the nanowire diameters are taken to be +/-1 nm, while the uncertainties in the nanowire lengths are taken to be +/-10 nm.

The three separate substrates included within each hydrothermal growth were individually compared using SEM, and in each instance their morphologies were found to be in agreement with one another. This lends credence to the overall reproducibility of the results presented within this chapter. For the sake of brevity, only one sample for each growth condition is presented here.

5.3 Effects of PEI molecular weight and concentration

Vertical ZnO nanowires were first hydrothermally grown without any PEI included in the precursors, to serve as a control. These are shown from a top-down and side-on perspective in Figure 5.1. The nanowires uniformly cover the surface of the substrate, and are relatively normal to its surface. The median diameter of the nanowires is 50 nm, and their median length is 1.46 μ m. As stated in the introduction to this thesis, ZnO almost exclusively forms in the hexagonal wurtzite structure, unless grown under very high pressures [12]. This causes the profiles of the nanowires to be hexagonal in cross-section, which is consistent with the literature. The nanowires have grown preferentially and anisotropically in the c-axis direction.

In comparison to the predictable outcome of the control hydrothermal growth, the

68 CHAPTER 5. HYDROTHERMAL GROWTH OF VERTICAL ZNO NANOWIRES



Figure 5.1: Cross-section (a) and top-down (b) SEM images of ZnO nanowires hydrothermally grown without any PEI.

inclusion of PEI in growth precursors can produce a wide variety of different outcomes, depending on the molecular weight and concentration of PEI used. The morphology of the resulting nanowires (or nanocrystals) can vary quite significantly from the regular nanowires grown without PEI, or nanowire growth can be suppressed altogether. Specifically, the results of the PEI-mediated hydrothermal growth can be categorised into the following:

- **Homogeneous nanowires**: Regular nanowires which grow uniformly across the surface of the substrate.
- Homogeneous hierarchical nanowires: Hierarchical nanowires which grow uniformly across the substrate surface. These are composed of large, primary nanowires which quickly transition much thinner, secondary nanowires
- Thin ZnO film: Rather than growing nanowires, a thin film of ZnO forms over the entire substrate
- **Growth supression**: ZnO nanowire growth is strongly suppressed over the entirety of the substrate, but the seed layer remains intact
- **Substrate etching**: The PEI removes the ZnO seed layer and etches into the silicon substrate

A summary table of the different PEI molecular weights/concentrations and their outcomes are presented in Table 5.1. Examples of these morphologies are given in the following subsections, which describe the effects of the PEI on the hydrothermal growth in detail.

| | PEI(800) | PEI(1300) | PEI(2000) |
|------|-----------------------|------------------------|-----------------------|
| 8 mM | Homogeneous nanowires | Substrate etching | Substrate etching |
| 6 mM | Homogeneous nanowires | Thin film deposition | Substrate etching |
| 4 mM | Growth suppression | Hierarchical nanowires | Thin film deposition |
| 2 mM | Homogeneous nanowires | Homogeneous nanowires | Homogeneous nanowires |

Table 5.1: Summary of results depending on concentration and molecular weight of PEI used in growth.

5.3.1 PEI(800)

Hydrothermal growths were carried out using 8, 6, 4, and 2 mM of PEI(800) included with the standard Zn(NO₃)₂/HMT precursors. SEM images of the nanowires grown using 8 mM and 6 mM of PEI(800) are shown in Figure 5.2 from both a top-down and cross-sectional perspective. The highest concentration of PEI(800) used (8 mM) results in long and thin homogeneous ZnO nanowires, which grow uniformly over the substrate surface. The nanowires are roughly normal to the surface of the substrate, as seen in Figure 5.2(a), although they show some bowing and bending due to their high aspect ratio. The top-down perspective shown in Figure 5.2(b) suggests that the nanowires partially fall into one another due to their length, forming upright bundles. The median length of these nanowires is 10.5 μ m, while the median diameter is 89 nm.

Using 6 mM of PEI(800) also results in homogeneous nanowires which more or less uniformly cover the substrate surface, although there are small areas where nanowire growth has been suppressed. The top-down and cross-section images shown in Figure 5.2(c) and (d) respectively suggest that the 6 mM PEI(800) nanowires also collapse into one another and form bundles; the top-down perspective in particular suggests that this happens to a greater extent than the 8 mM PEI(800) nanowires, despite the fact that the aspect ratio of the 6 mM nanowires is lower. The median length of the 6 mM PEI(800) nanowires is 7.98 μ m, and the median diameter is 51 nm.

Nanowires grown using 4 mM and 2 mM of PEI(800) are shown in Figure 5.3. Unlike the previous two results, including 4 mM of PEI(800) with the regular precursors in the growth solution strongly suppresses all nanowire growth across the entire sample. Figure 5.3(a) and (b) show cross-sectional and top-down views of the substrate after growth. The seed layer is still visible, but is almost totally devoid of all nanowires. Despite this, there are small, localised patches of the substrate where nanowire growth has taken place, although the growth has been heavily suppressed. An example of such a patch is visible in Figure 5.3(b). The nanowires which grow from these patches are typically 100 to 200 nm in length, with diameters on the order of tens of nanometres. Rather than growing normal to the seed layer, these small nanowires



Figure 5.2: (a) & (b): Cross-sectional and top-down SEM images of ZnO nanowires grown using 8 mM of PEI(800). (c) & (d): Cross-sectional and top-down SEM images of ZnO nanowires grown using 6 mM of PEI(800).



Figure 5.3: (a) & (b): Cross-sectional and top-down SEM images of the sample grown using 4 mM of PEI(800). Nanowire growth has been considerably suppressed. (c) & (d): Cross-sectional and top-down SEM images of ZnO nanowires grown using 2 mM of PEI(800)

lie at random orientations to the substrate.

Finally, nanowires grown using 2 mM of PEI(800) are shown in 5.2(c) and (d). This concentration of PEI results in the growth of homogeneous nanowires normal to the ZnO seed layer, which are uniform across the entire substrate. These nanowires closely resemble the control sample grown without any PEI, although they have better aspect ratios and more desirable dimensions. The nanowires have some variation in their diameter, with the thinner nanowires partially collapsing and forming bundles. The median length of the 2 mM PEI(800) sample is 2.6 μ m, and the median diameter is 58 nm.

Although the heavy suppression of nanowire growth at 4 mM of PEI(800) is incongruous with the other results presented here, the results are demonstrably reproducible. Figure 5.4 shows another substrate grown under identical conditions in a sep-



Figure 5.4: Another sample grown using 4 mM of PEI(800). Nanowire growth has been considerably suppressed. The inset shows a lower magnification image of the same sample.

arate growth bottle. Despite the fact that the seed layer is intact, this sample also shows the strong suppression of nanowire growth, suggesting that the growth suppression is a genuine phenomenon at this particular concentration.

5.3.2 PEI(1300)

SEM images of nanowires grown using 8 mM and 6 mM of PEI(1300) are shown in Figure 5.5. The sample grown using 8 mM of PEI(1300), shown in Figure 5.5(a) and (b), shows significant etching from the excessive concentration of PEI. The ZnO seed layer has been totally etched away from the Si substrate, which has in turn been anisotropically etched in the [111] direction. This concentration of PEI is clearly unusable for ZnO nanowire growth.

The sample grown using 6 mM of PEI(1300) is shown in Figure 5.5(c) and (d). Instead of growing ZnO nanostructures, this concentration of PEI causes the growth of a thin film of ZnO uniformly over the seed layer. The original seed layer can be seen as a thin layer at the very bottom of the ZnO film. The thin film is uniform and homogeneous, with little texture to the surface of the film.

Figure 5.6 shows SEM images of the samples grown using 4 mM and 2 mM of PEI(1300). Hierarchical ZnO nanowires are grown when 4 mM of PEI(1300) is included with the regular hydrothermal growth solution. These hierarchical nanowires are shown in cross-section and from a top-down perspective in Figure 5.6(a) and (b). The primary portion of these nanowires resemble regular nanowires synthesised using the hydrothermal growth. However, unlike regular hydrothermally-synthesised nanowires, the tips of these primary nanowires terminate in very fine secondary nanowires which have grown in the c-axis direction. These secondary nanowires can be up to tens of microns long and have diameters smaller than 50 nm. The secondary nanowires which intersects the vertical array of primary nanowires which grow from the substrate.

When 2 mM of PEI(1300) is included in the growth solution, regular homogeneous nanowires grow from the substrate surface, similar to those grown with 2 mM of PEI(800). These nanowires are shown from a cross-sectional and top-down perspective in Figure 5.6(c) and (d). The nanowires have uniformly grown across the substrate surface at a 90° angle. The median length of the nanowires is approximately 2.6 μ m, while their median diameter is roughly 50 nm.



Figure 5.5: (a) & (b): Cross-sectional and top-down SEM images of the sample grown using 8 mM of PEI(1300). The growth substrate has been significantly etched during growth. (c) & (d): Cross-sectional and top-down SEM images of the sample grown using 6 mM of PEI(1300). A homogeneous thin film of ZnO has grown from the seed layer, covering the substrate.



Figure 5.6: (a) & (b): Cross-sectional and top-down SEM images of the nanowires grown using 4 mM of PEI(1300). These ZnO nanowires are hierarchical, with distinct primary and secondary components. (c) & (d): Cross-sectional and top-down SEM images of ZnO nanowires grown using 2 mM of PEI(1300).

76 CHAPTER 5. HYDROTHERMAL GROWTH OF VERTICAL ZNO NANOWIRES



Figure 5.7: (a) & (b): Cross-sectional and top-down SEM images of the nanowires grown using 8 mM of PEI(2000). The PEI concentration has etched the Si growth substrate in the [111] direction. (c) & (d): Cross-sectional and top-down SEM images of ZnO nanowires grown using 6 mM of PEI(2000). This concentration has also etched the Si growth substrate.

5.3.3 PEI(2000)

Samples grown using 8 mM and 6 mM of PEI(2000) are shown in Figure 5.7. When 8 mM of PEI(2000) is included in the hydrothermal growth, the ZnO seed layer originally deposited on the substrates is completely removed and the Si substrate itself is heavily etched. These results are shown in Figure 5.7(a) and (b). The anisotropic etching of the Si substrate in the [111] direction mirrors the results obtained from including 8 mM of PEI(1300), previously shown in 5.5(a) and (b). Unlike the case for PEI(1300), substrate etching is still observed at a lower concentration of 6 mM of PEI(2000), due to the higher molecular weight of the PEI. This is shown from a top-down and cross-sectional perspective in Figure 5.7(c) and (d).

Figure 5.8 shows samples grown using 4 mM and 2 mM of PEI(2000). When 4

mM of PEI(2000) is included in the growth solution, a thin film of ZnO is deposited uniformly across the surface of the substrate, as in Figure 5.8(a) and (b). This result is similar to the thin film obtained by including 6 mM of PEI(1300), which is shown in Figure 5.5(c) and (d). As is the case with the 6 mM PEI(1300) thin film, the 4 mM PEI(2000) thin film doesn't show any kind of texture in its cross-section. However, unlike the film grown using PEI(1300), the 4 mM PEI(2000) thin film does have a textured surface. This texture consists of small circular nodules with diameters on the order of tens of nanometres which are flush with the thin film surface. Small ZnO nanospheres have grown from a portion of the nodules which texture the film surface. These nanospheres protrude from the surface of the film, with diameters comparable to the circular nodules. As these nanospheres sit atop the surface of the film, they have clearly nucleated at the end of the hydrothermal growth, after the bulk of the thin film has been deposited.

Finally, including 2 mM of PEI(2000) into the hydrothermal growth solution results in a homogeneous array of ZnO nanowires which grow vertically from the sputtered ZnO seed layer. These nanowires are shown from top-down and in cross-section in Figure 5.8(c) and (d). The nanowires grow uniformly across the surface of the substrate, similar to the control sample grown without any PEI. The nanowires have an median length of approximately 3.8 μ m, and an median diameter of 45 nm. The nanowires have grown normal to the substrate, although their high aspect ratio has caused them to partially collapse into one another, forming upright bundles. When seen in cross-section, it is apparent that the nanowires have some variation in their length, although their diameters are fairly uniform.



Figure 5.8: (a) & (b): Cross-sectional and top-down SEM images of the sample grown using 4 mM of PEI(2000). A homogeneous thin film of ZnO has grown from the seed layer. (c) & (d):Cross-sectional and top-down SEM images of ZnO nanowires grown using 2 mM of PEI(2000).

5.4 Influence of PEI on morphology

As shown in the previous section, including PEI with the standard growth precursors can cause the morphology of the hydrothermal growth products to significantly deviate from homogeneous nanowires. Specifically, we observe thin films of ZnO when 6 mM of PEI(1300) or 4 mM of PEI(2000) is included in the growth solution, and hierarchical ZnO nanowires when 4 mM of PEI(1300) is included. These unusual morphologies are not seen when standard growth precursors are used in the absence of PEI, and the control sample shows the predictable homogeneous nanowires one would expect from a hydrothermal growth.

5.4.1 ZnO thin film growth mechanism

Hydrothermally synthesised thin films of ZnO have been reported in a number of publications over the past few decades [157, 158, 159, 160, 161, 162, 163, 164]. Thin films typically arise when ZnO nanowires grow towards one another in non-axial directions, coalescing into large agglomerates and eventually forming an uninterrupted film. This process is depicted schematically in Figure 5.9. Nanowires can nucleate and grow spatially separated, eventually coalescing as the gaps between neighbouring nanowires grow smaller and smaller, as in Figure 5.9(a). Alternatively, the nanowires can nucleate in very close proximity and grow side-by-side, forming a continuous film from the beginning of the nanowire growth, as in Figure 5.9(b). A transition from regular nanowire growth to coalescence and thin film formation usually occurs when axial growth is suppressed or restricted in favour of non-axial growth. Although PEI usually encourages axial growth and restricts non-axial growth, we have synthesised clear and reproducible thin films of ZnO when using 6 mM of PEI(1300) or 4 mM of PEI(2000).

Further evidence for this growth mechanism is shown in Figure 5.10. This nanowire sample is grown at a slightly reduced concentration of 3 mM of PEI(2000), along with the regular 25 mM of standard $Zn(NO_3)_2/HMT$. This sample has grown inhomogeneously, with growth totally suppressed at some areas of the growth substrate. Other portions of the substrate have given rise to broad, flat-faced nanowires which are orientated roughly normal to the ZnO seed layer. These nanowires have grown in very close proximity to one another, partially overlapping one another and forming a rough and uneven thin film. The structure of the uneven film is highlighted by imaging the sample in cross-section, shown in the inset to Figure 5.10. Thin film formation has progressed one step further in some areas of the substrate, where nanowires have coalesced into extremely flat and even thin films. These thin films are untextured and homogeneous, and are directly comparable to the thin films synthesised with 4 mM of PEI(2000) and 6 mM of PEI(1300), as shown in Figures 5.8 and 5.5, respectively. This strongly suggests that a similar coalescence mechanism is responsible for the growth



Figure 5.9: A schematic depiction of the hydrothermal growth of a thin film of ZnO. Thin films can either start (a) as separated ZnO nanowires which coalesce, or (b) as adjacent nanowires which grow vertically.

of the thin films studied here, although it is not clear exactly how the PEI mediates this non-axial growth.



Figure 5.10: A sample grown using 3 mM of PEI(2000). Large adjacent ZnO nanowires have grown vertically, forming a rough thin film (shown cross-sectionally, inset). Portions of the substrate have gone on to grow homogeneous flat films, as depicted in the middle of the image.

5.4.2 Hierarchical nanowire growth mechanism

When 4 mM of PEI(1300) is included in the hydrothermal growth solution, hierarchical nanowires grow vertically from the ZnO seed layer. These hierarchical nanowires consist of comparatively large primary nanowires which abruptly transition into significantly thinner and longer secondary nanowires. The c-axis of the primary and secondary nanowires are aligned, and both share the same axial direction. Arrays of hierarchical nanowires have previously garnered significant interest due to their potential applications in a wide variety of devices [165, 166, 167, 168, 169, 170, 171, 172, 173, 174], although most of the hierarchical nanowires are not co-aligned along the c-axis. Hierarchical nanowires are especially useful in situations where a high surface area is required, such as dye-sensitised solar cells, or where a large degree of interconnectivity between nanowires is required [165, 166, 168, 169, 170, 171, 172, 174].

However, most of the previous reports on the synthesis of hierarchical nanowires use two separate and distinct growth steps, where the primary nanowires are grown in the initial step, and the secondary nanowires are grown in a separate second step. In comparison, the hierarchical nanowires shown here are grown in a single hydrothermal growth. Due to the distinct differences between the primary and secondary nanowires, and the fact that the secondary nanowires clearly grow after the primary nanowires have finished growing, we can conclude that the single hydrothermal growth consists of two distinct growth phases, both of which must be mediated through PEI.

As outlined in Chapter 2, PEI plays a complex role in the hydrothermal growth. In addition to possibly binding to the non-polar facets of ZnO nanowires during the hydrothermal growth, PEI strongly chelates with Zn^{2+} ions in solution [75]. The chelated Zn^{2+} ions ions are gradually released back into the growth solution by PEI-Zn complexes interacting with formaldehyde via a Mannich reaction [75]. These reactions provide another source of Zn^{2+} ions for continued nanowire growth which is otherwise absent when PEI is not included in the growth solution [75]. The formation of hierarchical nanowires suggests that this PEI-mediated source of Zn^{2+} allows for further nanowire growth after other Zn^{2+} sources have been exhausted, which allows for the growth of secondary nanowires.

The hierarchical growth process is depicted schematically in Figure 5.11. Zn^{2+} ions, initially provided by the $Zn(NO_3)_2$ precursor, directly and indirectly crystallise into primary nanowires, as shown in Figure 5.11(a). Once these initial precursors have been consumed, the PEI-chelated Zn^{2+} ions are the only source of Zn^{2+} available.



Figure 5.11: A schematic depiction of the growth mechanism of hierarchical ZnO nanowires. (a) The primary nanowires are grown through direct and indirect ZnO crystallisation from numerous Zn^{2+} ion sources. (b) After the primary ZnO nanowires have grown, a comparatively slow PEI-mediated Zn^{2+} ion source causes further growth of the nanowires. The difference in Zn^{2+} concentration causes the secondary nanowires to have a much smaller diameter than the primary nanowires.

These ions in turn directly and indirectly crystallise into secondary nanowires, depicted in Figure 5.11(b). The difference in diameters between the primary and secondary nanowires can be explained by the significantly reduced concentration of Zn^{2+} in the growth solution once the bulk of ions have been consumed. The chemical interactions between the precursors within the growth solution are thoroughly explained in Chapter 2, in addition to the direct and indirect routes of ZnO crystallisation.

Further evidence for the PEI-mediated growth mechanism of secondary nanowires can be seen in cases where the PEI has been inhomogeneously distributed throughout the hydrothermal growth solution. Although every effort is made to thoroughly mix the PEI into the growth solution, agglomerates and clumps can form as the polymer settles in the bottle. These agglomerates can then deposit on the substrate or vertical nanowires during the growth. In some cases, the deposited PEI agglomerates lead to highly localised secondary nanowire growth, causing portions of otherwise homogeneous nanowires to grow as hierarchical nanowires. Examples of localised hierarchical nanowire growth are shown in Figure 5.12. Figure 5.12(a) and (b) show a portion of a sample grown using 2 mM of PEI(800). An amorphous agglomerate of PEI rests on the surface of the nanowire array indicated by an arrow, in the centre of the image. Secondary nanowires have grown from the tips of the regular nanowires which are nearby the agglomerate, as shown in Figure 5.12(b). Part of the PEI agglomerate is visible in the upper-right corner of the image. Likewise, Figure 5.12(c) and (d) show localised secondary nanowire growth from a sample grown using 5 mM of PEI(800). The PEI agglomerate is located near the top of the image, and indicated with an arrow. Figure 5.12(d) shows the edge of the localised secondary nanowires, where there is a clear transition from hierarchical and regular vertical nanowires. These samples strongly suggest that PEI-chelated Zn^{2+} ions reintroduced into the growth solution are responsible for secondary nanowire growth.



Figure 5.12: Localised secondary nanowire growth caused by agglomerations of PEI. (a) & (b) show a sample grown using 2 mM of PEI(800), while (c) & (d) show a sample grown using 5 mM of PEI(800). The hierarchical nanowires caused by the agglomerated PEI are clearly demarcated from the rest of the nanowires, which are homogeneous.

5.5 Comparison of homogeneous nanowires

As seen in the control sample shown in Figure 5.1, homogeneous nanowires grow when PEI is omitted from the hydrothermal growth solution. Homogeneous nanowires are also grown when 2 mM of PEI(2000) or PEI(1300) is included in the growth solution, or when PEI(800) is included over a range of concentrations. Generally speaking, longer, thinner nanowires are preferable to shorter, wider nanowires, and PEI is primarily used in the hydrothermal growth to increase their aspect ratio. To compare the effects of PEI on the length and diameter of homogeneous nanowires, SEM images of nanowires grown at different concentrations and molecular weights were taken and measured using ImageJ.

Figure 5.13 shows a box and whisker plot of the measured lengths of all of the homogeneous nanowires grown in this chapter, including the control nanowires grown without PEI. This graph shows the lengths of the individual nanowires measured (marked as diamonds on the graph), and the median length, represented by the centre line of each box. The upper and lower bounds of each box represent the upper and lower quartiles of the measurements, indicating the variation observed across the measured nanowires. The values of the median lengths are shown in the upper-right inset.

Homogeneous nanowires grown using 8 mM of PEI(800) (black) are the longest on median, with a median length of 10.5 μ m. However, they also show the largest variation in length, with some nanowires being several microns shorter than the median. Nanowires grown using 6 mM of PEI(800) (red) are slightly shorter on median, with a median length of 7.9 μ m. Their variation is comparable to nanowires grown using 8 mM of PEI(800), although the standard deviation in length is slightly smaller. In comparison, nanowires grown with 2 mM of PEI(800) (blue) are significantly shorter, with a median length of 2.6 μ m. The variation in length is slightly reduced compared to nanowires grown using higher concentrations of PEI(800), with the majority of measured nanowires falling within +/- 0.5 μ m of the median.

Nanowires grown using 2 mM of PEI(1300) (magenta) are directly comparable in length to those grown with 2 mM of PEI(800), also having a median length of 2.6 μ m. However, they show a higher degree of variation in their length, with some nanowires differing from the median by +/- 1 μ m. Nanowires grown with 2 mM of PEI(2000) (green) are measurably longer on median, with a median length of 3.8 μ m. These nanowires also show a higher degree of variation in their length, with some nanowires differing from the median by +/- 2 μ m. However, this variation is still



Figure 5.13: Lengths of homogeneous nanowires. Individual measurements are shown as coloured symbols. The median of each recipe is indicated by the centre line of the box graph, while the upper and lower lines of the box indicate the upper and lower quartiles. The median length of each recipe is shown in the upper-right of the graph.

significantly lower than the nanowires grown using 8 mM and 6 mM of PEI(800). Finally, the control sample grown without PEI (purple) has the shortest nanowires, with a median length of 1.6 μ m. These nanowires also have the smallest degree of variation in length out of all those studied here, with the large majority of nanowires falling within +/- 0.25 μ m of the median.

Figure 5.14 shows the measured diameters of the homogeneous nanowires studied here, following the same convention as Figure 5.13. The median values of the diameters are shown in the upper-right inset, along with the median aspect-ratios (defined as the median length of each nanowire divided by the median width.) Nanowires grown with 8 mM of PEI(800) (black) have a median diameter of 89 nm, and the majority of the nanowires fall within +/- 25 nm of this value. In comparison, nanowires grown using 6 mM of PEI(800) (red) are narrower on median, with a median diameter of 52 nm. The nanowire diameters show a considerably asymmetric variation about the median, with the upper quartile reaching 126 nm. The majority of the thinner nanowires fall within 25 nm of the median.



Figure 5.14: Diameters of homogeneous nanowires. Individual measurements are shown as coloured symbols. The median of each recipe is indicated by the centre line of the box graph, while the upper and lower lines of the box indicate the upper and lower quartiles. The median diameter of each recipe is shown in the upper-right of the graph, along with the median aspect ratio (taken as the median length divided by the median diameter.)

Nanowires grown using 2 mM of PEI(800) (blue) have a slightly wider diameter on median than those grown using 6 mM, with a median diameter of 58 nm. The variation in diameter is fairly restricted, with most nanowires falling within +/- 25 nm of the median. The diameters of the nanowires grown using 2 mM of PEI(1300) (magenta) and PEI(2000) (green) are very similar, with median diameters of 55 and 45 nm, respectively. These are comparable to the median diameter of the control sample (purple), which is 50 nm. All of the nanowires grown using 2 mM of PEI have a similar degree of variation in diameter to the control sample, despite the different variations in their length.
5.5.1 Summary

The above results indicate that all varieties of PEI produce nanowires with higher aspect ratios than regular nanowires grown without, provided the concentration of PEI causes homogeneous nanowires to grow in the first place. Different concentrations and molecular weights of PEI can be chosen to suit the requirements of the nanowire morphology. For example, in the hypothetical case where the length of the nanowires is the most important variable, using 8 mM of PEI(800) in the nanowire growth is clearly ideal. However, if a high aspect ratio is prioritised, then growing nanowires using 6 mM of PEI(800) is preferred. If a minimal diameter is preferred over both length and aspect ratio, then 2 mM of PEI(2000) is the ideal concentration and molecular weight of PEI to include in the growth.

Variations in the diameters and lengths of the nanowires may be important as well, depending on their applications. Nanowires grown at 8 mM and 6 mM of PEI(800) clearly show the most variation in both length and diameter, with 8 mM of PEI(800) causing the most variation in the length of the nanowire, and 6 mM of PEI(800) causing the most variation in diameter. Conversely, minimal variation in the diameters and lengths of the nanowires are seen when using low concentrations of PEI, or omitting PEI entirely. All of these variables must be carefully weighed and considered when choosing what PEI to use, depending on the requirements and applications of the final ZnO nanowires.

5.6 Conclusion

In conclusion, the molecular weight of the PEI used in the nanowire growth process is of critical importance, despite often being overlooked. Different molecular weights of PEI can produce drastically different results, even when used at identical concentrations. Excessively high concentrations of high molecular weight varieties of PEI can etch and destroy the growth substrate, and are unsuitable for nanowire growth. Specifically, 8 mM and 6 mM of PEI(2000) etches the Si growth substrates in the [111] direction, as does 6 mM of PEI(1300).

Different molecular weights and concentrations of PEI can also result in irregular nanostructures, which differ significantly from regular homogeneous nanowires grown without PEI. Homogeneous thin films of ZnO are grown at 4 mM of PEI(2000) and 6 mM of PEI(1300), which are caused by large nanowires coalescing together. Hierarchical nanowires are grown at 4 mM of PEI(1300), which have distinct primary and secondary nanowire components. The primary nanowires resemble regular homogeneous nanowires, transitioning into ultra-thin secondary nanowires. These secondary nanowires are grown through a PEI-mediated source of Zn^{2+} ions which are still present in the growth after the bulk of ions have been consumed.

Homogeneous nanowires are grown at 8 mM, 6 mM, and 2 mM of PEI(800), and at 2 mM of PEI(1300) and PEI(2000). All of these nanowires have higher aspect ratios than nanowires grown without any PEI. The ideal concentration of PEI depends on the requirements of the nanowires. The longest on average nanowires are grown using 8 mM of PEI(800), with a median length of 10.5 μ m. The thinnest on average nanowires are grown using 2 mM of PEI(2000), with a median diameter of 45 nm. Finally, the highest aspect-ratio nanowires are grown using 6 mM of PEI(800), with a median aspect ratio of 153.

Chapter 6

PEI-mediated growth of hierarchical ZnO nanowire FETs

While hydrothermal ZnO nanowires have been used in field-effect transistors (which are reviewed extensively in Chapter 3), they are typically field-independent when used as-is due to excessive charge carrier concentrations caused by unintentionally incorporated electron donors. This can be addressed by annealing the nanowires at high temperatures (typically \geq 400°C) [113, 114, 118, 54, 106, 11, 93] at the expense of losing the advantage of low-temperature synthesis. With future electronics trending towards the use of delicate substrates [175], low-temperature processes will be necessary for fabricating devices at the wafer scale. Further development of ZnO nanowire FETs will therefore benefit from being able to synthesize field-dependent nanowires in situ, whilst also taking full advantage of low-temperature processing and device fabrication [118].

As we have seen in Chapter 5, the hydrothermal growth can be significantly affected by including different concentrations and molarities of PEI. Motivated by these previous results, lateral ZnO nanowires are grown using a range of different concentrations and molecular weights of PEI. A portion of these recipes yield distinctly hierarchical primary/secondary ZnO nanowires, similar to the hierarchical morphology reported in Chapter 5. Significantly, these secondary nanowires are field dependent in their as-grown state, and do not require post-growth annealing or processing before use. This is in direct contrast to the primary nanowires, which are unresponsive to applied fields. Several FETs are fabricated by growing hierarchical nanowires show good device performance, with on-off ratios of 10³ to 10⁵ and threshold voltages between -7.5 V to 5 V. Transmission electron microscopy shows the secondary nanowires hierarchically nucleate from the larger ZnO nanorods, with both sharing a common c-axis. Persistent photoconductivity measurements suggest

that depleting surface states on the nanowires play a significant role in determining the threshold voltage of the devices.

6.1 Motivation and background

In order to use ZnO nanowires in field-effect transistors, they must first be synthesised on a growth substrate, and then electrically contacted on an appropriate FET device substrate. While the hydrothermal synthesis of ZnO nanowires is rapid and scalable, making electrical contact to the nanowires poses a challenge. One method of electrically contacting nanowires is to initially grow them on a dedicated substrate, transfer them to a separate device substrate, and then perform electron beam lithography to pattern a resist for metallisation, as described in Chapter 8. However, this contacting process is resource-intensive and laborious, and cannot be effectively scaled up to higher throughputs.

A solution to this problem is to use a ZnO nanowire growth substrate which also doubles as the FET device substrate. This is most easily accomplished by synthesising intersecting lateral networks of ZnO nanowires from patterned seed layers, as described in Section 2.5.3 in Chapter 2. The patterned seed layers consist of thin ZnO films which are capped with Ti. Only the sidewall of the ZnO layer is exposed during the hydrothermal growth due to the Ti capping layer, which restricts the nanowires to grow laterally from the sidewall and parallel with the substrate. The ZnO/Ti layers are patterned in pairs, which are separated by a small gap on the order of tens of microns. Lateral ZnO nanowires intersect in the middle of this gap, which forms the FET channel. The Ti capping layers can then be electrically contacted as drain and source electrodes, and current passed through the intersecting wires. The entire SiO₂/Si substrate can then be used as a global back gate to switch the FET on and off. A thin layer of Cr/Au (\approx 5/50 nm) is typically applied to the back of the substrate to improve electrical contact to the Si bulk. The architecture of the final device is shown in Figure 6.1.

Nanowires can be hydrothermally grown with or without using polyethylenimine (PEI), as described in Chapter 2. In Chapter 5, it was shown that varying the molecular weight and concentration of PEI in the growth solution can drastically affect the nanowire morphology. Three different molecular weights are of PEI are used ($M_W = 2000 \text{ g/mol}$, 1300 g/mol, and 800 g/mol) at a variety of concentrations to grow lateral nanowires, and to assess their viability as FETs.



Figure 6.1: A schematic of an FET with an intersecting network of ZnO nanowires as the channel material. The Ti capping layer ensures that the nanowires grow laterally, and also act as a source and drain. The SiO_2/p -Si substrate can then be used as a dielectric and back gate, completing the FET.

6.2 Hydrothermal growth and hierarchical nanowire FET fabrication

Standard photolithographic processing was used to define sets of pairs of electrodes on $1.2 \text{ cm} \times 1.2 \text{ cm} \text{SiO}_2(100 \text{ nm})/\text{p}^+$ -Si wafers, sourced from Silicon Quest International. The inter-electrode spacing of the electrode pairs ranged from 10 μ m to 80 μ m, in 10 μ m increments. After the photoresist was patterned and developed, 100 nm of ZnO was deposited via RF sputtering, followed by a further 60 nm of Ti. The initial ZnO layer serves as a seed for hydrothermal growth. The layer of Ti caps the top of the ZnO seed layer, which prohibits vertical nanowire growth and provides an electrical contact for device measurements, while the sidewall of the ZnO seed layer remains uncovered. This geometry allows for the lateral growth of ZnO nanowires, described in Chapter 2. Photolithographic and RF sputtering processes are described in detail in Chapter 4.

6.2.1 Hydrothermal growth

All of the growth solutions were prepared using an equimolar mix of 25 mM of $Zn(NO_3)_2$ (Sigma Aldrich, 98% purity) and 25mM of HMT (Sigma Aldrich, 99% purity) in 250 mL of de-ionized (DI) water ($\geq 18.2 \text{ M}\Omega \cdot \text{cm}$), and a variable amount of PEI. PEI was used in concentrations between 2 - 12 mM with molecular weights of 800, 1300, and 2000 g/mol, similar to the range of variables explored in Chapter 5. These varieties of PEI are referred to as PEI(800), PEI(1300), and PEI(2000), respectively.

Each molecular weight of PEI was included in the hydrothermal growth solution at an initial concentration of 2 mM, and was incrementally increased in subsequent hydrothermal growths by 1 mM amounts. The concentration of PEI in subsequent hydrothermal growths was increased until the ZnO seed layer was etched away. In the case of PEI(800), the concentration of PEI was increased by 2 mM increments after reaching 6 mM.

After the precursors were thoroughly mixed, borosilicate glass bottles containing the growth solutions were put in a water bath and left to preheat for 1 hour at a constant temperature of 95°C. After the preheat, the growth/device substrates were submerged in the growth solutions for 19 hours at a constant 95°C before being removed. The samples were then rinsed with DI water ($\geq 18.2 \text{ M}\Omega \cdot \text{cm}$) and then dried with nitrogen at room temperature. The general theory and experimental techniques underpinning hydrothermal growth of ZnO nanowires are described in detail in Chapter 2.

6.2.2 ZnO nanowire synthesis results

The concentration and molecular weight of PEI used in the hydrothermal growth has a significant effect on the synthesised lateral ZnO nanowires, and can result in a range of distinctly different nanowire morphologies. This is similar to the PEI-mediated growth of vertical nanowires presented in Chapter 5, which also have variable morphologies based on the concentration and PEI used in the hydrothermal growth. The final morphology of the lateral nanowires can be categorised as:

- Nanowires Regular ZnO nanowires which grow laterally from the patterned seed layers. These are typically 1 10 μ m in length and 100 nm 1 μ m in diameter
- Hierarchical nanowires Hierarchical nanowires which grow laterally from the patterned seed layers. These consist of large primary nanowires which abruptly terminate in long, thin secondary nanowires. The primary nanowires have dimensions on the order of the regular, non-hierarchical nanowires, while the secondary nanowires are typically 5 10 μ m long. These are morphologically very similar to the hierarchical nanowires synthesised in Chapter 5.
- Nanowire growth suppression Nanowire growth is suppressed from the ZnO seed layers, which remain intact after the hydrothermal growth. Nanowire growth suppression typically occurs at high concentrations of PEI.
- Substrate etching Etching of the ZnO seed layers or the SiO₂/Si substrate from excessive PEI concentration.

The influence of the PEI concentration and molecular weight on the morphology of the ZnO nanowires is summarised in Table 6.1.

| Table 6.1: A summary of the lateral nanowire morphologies which result from a giver |
|---|
| PEI concentration and molecular weight |

| | PEI(800) | PEI(1300) | PEI(2000) |
|-------|-----------------------|-------------------|-------------------|
| 2mM | Nanowires | Nanowires | Hierarchical NWs |
| 3 mM | Nanowires | Hierarchical NWs | Hierarchical NWs |
| 4 mM | NW growth suppression | Hierarchical NWs | Nanowires |
| 5 mM | Nanowires | Nanowires | Substrate etching |
| 6 mM | Nanowires | Substrate etching | |
| 8 mM | Hierarchical NWs | | |
| 10 mM | NW growth suppression | | |
| 12 mM | Substrate etching | | |

The range of possible nanowire morphologies is similar to those obtained from the PEI-mediated growth of vertical nanowires, which is reported in Chapter 5. Figure 6.2 shows SEM images of (a) hierarchical lateral nanowires and (b) regular lateral ZnO nanowires. The lateral hierarchical nanowires strongly resemble the vertical hierarchical nanowires grown in Chapter 5, although their orientation differs due to the geometry of the ZnO sidewall. Hierarchical nanowires are grown when using 4 mM of PEI(1300), which is also the same concentration/molecular weight used to grow vertical hierarchical nanowires. Lateral hierarchical nanowires are also grown when using 8mM of PEI(800), 3 mM of PEI(1300), and 2 - 3 mM of PEI(2000). In comparison, the regular nanowires are very similar in dimension to the primary portion of the hierarchical nanowires, although lack the hierarchically-nucleated secondary nanowires.



Figure 6.2: SEM images of (a) hierarchical lateral ZnO nanowires and (b) regular lateral nanowires. The seed-layer is not shown in these images.

It is also worth noting that the growth of lateral ZnO nanowires is suppressed when using 4 mM of PEI(800), which also occurs during the vertical growth of nanowires. Figure 6.3 shows an optical image of a lateral substrate grown with 4 mM of PEI(800). The distance between the two electrodes is 50 μ m. The source and drain ZnO/Ti electrodes still clearly remain on the substrate, but minimal nanowires have nucleated and grown from the sidewalls. Growth suppression of nanowires typically occurs as the concentration of PEI becomes excessive, chelating the Zn ions out of solution and preventing homogeneous/heterogeneous ZnO nucleation. When the concentration of PEI is increased further, the ZnO seed layers are etched away from the substrate, and the substrate itself is etched. In comparison, the growth suppression at 4 mM of PEI(800) occurs at a much lower concentration of PEI. Although the mechanism behind the growth suppression at this concentration is currently not known, it is clearly a reproducible and real phenomenon.





The most significant result of the PEI concentration/molecular weight experiments is the distinct hierarchical nanowire morphology which occurs under the right growth conditions. The concentration and molecular weight of PEI used in the hydrothermal growth has a definite effect on the dimensions of the regular lateral nanowires, just as it has an effect on the dimensions of the vertical nanowires. However, FETs fabricated using these lateral nanowires are shown to be field-independent, while FETs fabricated using the hierarchical nanowires show pronounced field dependence. As such, an indepth analysis of the relationship between the PEI used and the dimensions of the regular lateral nanowires is omitted here, and the remainder of this chapter focuses on the fabrication and electrical characterisation of the hierarchical nanowire FETs.

6.2.3 Hierarchical nanowire device fabrication

Eight pairs of ZnO/Ti electrodes are deposited on each growth/device substrate, and are separated by 10 μ m to 80 μ m channel distances. When hierarchical nanowires are synthesised from the ZnO seed layers, the shorter electrode channels are bridged by large primary nanorods which meet in the middle of the channel gap. The range of channel distances ensures that a portion of the longer channels are bridged by the secondary nanowires, which grow hierarchically from the primary nanowires. The longest channel gaps are typically left unbridged after the nanowire growth.

Figure 6.4 shows a schematic and SEM image of a pair of electrodes bridged by the secondary nanowire portion of the hierarchical nanowires after the hydrothermal growth process. Lateral arrays of secondary nanowires have grown horizontally along the surface of the substrate from the opposing electrodes, meeting and intersecting in the middle of the channel to bridge the electrodes and form a conduction pathway. These secondary nanowires nucleate hierarchically from the primary nanowires which grow directly from the ZnO sidewall, as shown in Figure 6.4 (b). Figure 6.4 (c) and (d) show corresponding SEM images of an actual device, which is grown with 4 mM of PEI(1300). The primary nanowires are from 150 to 300 nm in diameter and 10 - 15 μ m in length. They project radially in a 3 dimensional brush-like structure from the exposed ZnO sidewall. Secondary nanowires, with diameters from 10 to 50 nm and lengths of up to 10 μ m, emerge from the tips of the primary nanowires, as shown in Figure 6.4 (d).

Although the secondary nanowires are only 10 to 50 nm in diameter, it's possible to image their diffraction patterns against the SiO₂ substrate using regular optical microscopy. Figure 6.5 shows an optical image of an FET bridged with hierarchical nanowires from a top-down perspective. These hierarchical nanowires are grown using 8 mM of PEI(800). The two large electrodes ZnO/Ti are labelled as 'S' and 'D' (for 'source' and 'drain'.) Large primary nanowires grow laterally from the exposed ZnO sidewall, fringing the electrodes. The inset of Figure 6.5 shows a magnified view of the channel gap. The secondary nanowires can clearly be discerned despite their diameter – optical microscopy is therefore an efficient way to quickly determine whether the hydrothermally-grown nanowires are regular nanowires or whether they're hierarchical. The large image of the FET is taken at 10x magnification, while the magnified inset is taken at 100x.

SEM images of several secondary nanowire FETs are shown in Figure 6.6. Figure 6.6(a) and (b) shows an FET fabricated using 3 mM of PEI(2000). The channel gap of



Figure 6.4: (a) A schematic of the hierarchical nanowire FET, showing the seed layers/electrodes and bridging nanowires. (b) A close-up of the primary/secondary nanowires. (c) An SEM image of an actual device. (d) A close-up of the channel of the device, showing the primary/secondary nanowires.



Figure 6.5: Optical image of a PEI-mediated hierarchical nanowire FET grown with 8 mM of PEI(800). Both primary and secondary nanowires can be discerned. The source and drain electrodes are marked 'S' and 'D' respectively. The gap between the electrodes is 80 μ m.

the FET is too wide for the primary nanowires to intersect in the middle, while the secondary nanowires readily bridge the gap, highlighted in Figure 6.6(b). Figure 6.6(c) and (d) show two different FET devices grown using 4 mM of PEI(1300). Although the length of the secondary nanowires is similar between the two devices, they have nucleated and grown from the primary nanowires in distinctly different densities. Figure 6.6(e) and (f) show two FETs fabricated using 8 mM of PEI(800). This recipe yields large numbers of secondary nanowires, creating very dense and complicated intersecting networks.



Figure 6.6: SEM images of hierarchical nanowire FETs synthesised using: (a),(b) 3 mM of PEI(2000), (c),(d) 4 mM of PEI(1300), (e),(f) 8 mM of PEI(800).

6.2.4 Hierarchical nanowire growth mechanism

Although the hydrothermal growth process only involves a single growth step, two distinct nanostructures grow hierarchically from the seed layers. As noted in Section

6.2.2, the lateral hierarchical nanowires grown here strongly resemble the vertical hierarchical nanowires grown in Chapter 5. Both vertical and lateral hierarchical nanowire growths use PEI at similar or identical concentrations, and neither growth methods yield hierarchical nanowires when PEI is omitted. It is therefore very likely that the lateral and vertical hierarchical nanowire growth is driven by the same PEI-mediated mechanism, described in Section 5.4.2 in Chapter 5.

The growth of lateral hierarchical nanowires is shown in Figure 6.7. The primary nanowires are first grown during an initial growth phase, shown in Figure 6.7(a). During this phase, the Zn²⁺ ions are provided exclusively by the decomposition of Zn(NO₃)₂ precursors, which directly and indirectly crystallise into the primary nanowires. A portion of these Zn²⁺ ions form PEI-Zn complexes through chelation. As the hydrothermal growth continues, the concentration of Zn²⁺ precursor in the solution decreases as it crystallises out. Eventually the only source of Zn²⁺ ions are the remaining Zn-PEI complexes. These Zn-PEI complexes gradually release the Zn²⁺ ions back into solution via a Mannich reaction with formaldehyde in solution [75]. This drives the second growth phase, shown in Section 5.4.2(b). As the concentration of Zn²⁺ in solution is comparatively lower than during the primary growth phase, the released ions crystallise into the much thinner secondary nanowires, which nucleate and grow from the terminating ends of the primary nanowires.



Figure 6.7: A schematic of the PEI-mediated hierarchical nanowire growth mechanism. Primary and secondary nanowires are grown in two distinct phases.

6.2.5 Hierarchical nanowire crystal structure

High resolution transmission electron microscopy (HRTEM) was performed to investigate the crystal structure of the hierarchical nanowires. These measurements were undertaken by Dr. Jerome Majimel at ICMCB Bordeaux - the specific parameters used are detailed in Chapter 4. The hierarchical nanowires were grown using 4 mM of PEI(1300). These measurements show that the primary and secondary nanowires are monocrystalline, and both are orientated along the [001] c-axis. Figure 6.8(a) shows the hexagonal wurtzite structure of the primary nanowires, with corresponding electron diffractogram shown inset. A representative HRTEM measurement of the secondary nanowire is shown in 6.8(b), confirming the hexagonal wurtzite crystal structure and c-axis orientation of the secondary nanowires.



Figure 6.8: TEM images of (a) primary and (b) secondary nanowires from hierarchical nanowires.

HRTEM images of the primary/secondary nanowire interfaces also confirm that the secondary nanowires nucleate and grow epitaxially from the primary nanowires. Figure 6.9(a) shows a secondary nanowire growing from the side face of a primary nanowire, with the interface between the two highlighted. Figure 6.9(b) shows an HRTEM image of the highlighted interface, displaying the common c-axis orientation between the two. Although the primary and secondary nanowires have grown epitaxially in relation to each other, there are small areas at the interface where the crystal orientation slightly deviates from the common c-axis. Figure 6.9(c) highlights these deviations, which appear as contrast variations in the micrographs.

6.3 Electrical measurements

The architecture of the growth/device substrates used in the hierarchical nanowire FET fabrication allows for bridged ZnO/Ti electrodes to be used as an FET, as described in Section 6.1. In these measurements, the ZnO/Ti electrodes are used as the transistor source and drain, while the SiO_2/Si substrate is used as a dielectric and global



Figure 6.9: (a) Secondary nanowires hierarchically nucleating from primary nanowires.(b) TEM images of the highlighted primary/secondary nanowire interface. Arrows highlight small deviations in the shared c-axis, which are shown in (c).

backgate. The measurements of the hierarchical nanowire FETs in this section are organised by the molecular weight of the PEI used in their synthesis. The experimental techniques used to electrically measure the devices are described in Chapter 4, while the device characteristics which are measured are described in Chapter 3. All of the transfer characteristics in this section are obtained by sweeping the gate voltage from the negative to the positive direction, and then back in the negative direction.

6.3.1 **PEI(2000) devices**

As noted in Table 6.1, hierarchical nanowires are synthesised when 3 mM of PEI(2000) is included in the growth. Because the spacing of the ZnO/Ti electrodes ranges from 10 μ m to 80 μ m, some devices are bridged by intersecting secondary nanowires, while other devices are bridged by intersecting primary nanowires. Both of these were electrically measured to contrast the difference in field dependence between the two. The devices were measured immediately after the hydrothermal growth, with no annealing or post-growth processing.

Figure 6.10 shows drain-source current vs. gate voltage measurements (transfer characteristics) for FETs bridged with secondary nanowires (black, circles) and with primary nanowires (red, squares) grown using 3 mM of PEI(2000). Both devices have been fabricated on the same device substrate, and are driven at a drain-source voltage of 1 V. Despite the fact that the devices are unannealed, the secondary nanowire FET shows clear field dependence between $V_G = -10$ V and 10 V. The FET operates in enhancement mode, with a threshold voltage of 2 V and an on-off current ratio of 10^5 . The transconductance of the device is 3.7×10^{-8} S, and the subthreshold swing is 547 mV/decade. The mobility of the device is very difficult to calculate, as the channel length is uncertain given the multiple wires which make up the channel. Furthermore the capacitance is difficult to estimate, which is also necessary to accurately determine the device mobility.

In contrast, the primary nanowire FET shows very little field dependence over the same range of gate voltages. Although the primary and secondary nanowires have different sizes (which may affect their response to an applied gate voltage), the difficulty in gating the primary nanowire FET is consistent with other examples of as-grown ZnO nanowires [113, 114, 118, 54, 11], even with diameters directly comparable to the secondary nanowires [113, 118].

Although the field dependence of the as-grown secondary nanowires is surprising, it is consistent across all of the secondary nanowire devices measured. The transfer characteristics from two additional FETs fabricated using 3 mM of PEI(2000) are shown



Figure 6.10: Transfer characteristics of secondary-nanowire (black, circles) and primary-nanowire FETs (red, squares) fabricated using 3 mM of PEI(2000). Both devices are driven at $V_{\rm DS}$ = 1 V.

in Figure 6.11. Both of these devices are driven at $V_{DS} = 0.1$ V. In contrast to the previous FET, both of these devices operate in depletion mode. The FET measured in Figure 6.11(a) has a threshold voltage of \approx -2.5 V and an on-off ratio of 10⁵. The transconductance of the device is 4.38×10^{-8} S, and the subthreshold swing is 191 mV/decade. The FET measured in Figure 6.11(b) has a threshold voltage of \approx -6.5 V and an on-off ratio of 10⁵. The transconductance is 6.44×10^{-8} S, and the subthreshold swing is 376 mV/decade. The device characteristics of the FETs presented in Figures 6.10 and 6.11 are summarised in Table 6.2. They are listed as FET 1, (a), and (b), respectively.



Figure 6.11: Further transfer characteristics of PEI-mediated hierarchical nanowires synthesised using 3 mM of PEI(2000). Both devices are driven at $V_{DS} = 0.1$ V.

 Table 6.2:
 Summary of the device characteristics of 3mM PEI(2000) hierarchical nanowire FETs

| FET | V_{th} | $I_{on/off}$ | g _m (S) (max.) | Subthreshold swing (mV/dec.) (min.) |
|-----|----------|--------------|---------------------------|-------------------------------------|
| 1 | 2 | 10^{5} | $3.7	imes10^{-8}$ | 547 |
| (a) | -2.5 | 10^{5} | $4.38	imes10^{-8}$ | 191 |
| (b) | -6.5 | 104 | $6.44	imes10^{-8}$ | 376 |

6.3.2 **PEI(1300)** devices

All of the secondary-nanowire FETs fabricated using PEI(1300) also show well-defined field dependence and device characteristics, similar to those grown with PEI(2000). In total, 4 separate devices are measured here. Figure 6.12 shows the transfer characteristics of four FETs bridged by secondary nanowires grown using 4 mM of PEI(1300). All of the FETs clearly respond to applied gate voltages and have clear on and off regimes. The devices are driven at a constant drain-source voltage of 1 V. As before, all of the devices here exhibit n-type behaviour. The two devices shown in 6.12(a) and 6.12(d)

operate in enhancement mode, with threshold voltages of 4 V and 5 V respectively, and with on-off ratios of 10^4 and 10^3 . The other two devices (shown in 6.12(b) and 6.12(c)) operate in enhancement mode, with threshold voltages of -4 V and -7.5 V, respectively. Both of these devices have an on-off ratio of 10^4 .



Figure 6.12: Transfer characteristics of hierarchical nanowire FETs synthesised using 4 mM of PEI(1300). The devices are measured at $V_{\rm DS}$ = 1 V.

The device characteristics for the FETs displayed in Figure 6.12, including their transconductance and subthreshold swing values, are summarised in Table 6.3. The transconductances of the devices are comparable to those fabricated using PEI(2000), and are on the order of $10^{-8} - 10^{-9}$ S. The subthreshold swings are also comparable, although the FET shown in Figure 6.12(c) has a slightly higher subthreshold swing than other FETs, which is on the order of 1V/decade.

 Table 6.3: Summary of the device characteristics of 4 mM PEI(1300) hierarchical nanowire FETs

| FET | V_{th} | $I_{on/off}$ | g_m (max.) (S) | Subthreshold swing (mV/dec.) (min.) |
|-----|----------|-----------------|---------------------|-------------------------------------|
| (a) | 4 | 104 | $3.37	imes10^{-9}$ | 561 |
| (b) | -4 | 10^{4} | $1.3	imes10^{-8}$ | 719 |
| (c) | -7.5 | 10^{4} | $5.7 	imes 10^{-8}$ | 1083 |
| (d) | 5 | 10 ³ | $9.55	imes10^{-9}$ | 336 |

6.3.3 **PEI(800)** devices

Figure 6.13 shows the transfer characteristics of three FETs bridged by secondary nanowires grown using 8 mM of PEI(800). As before, all of the devices show n-type behaviour and well-defined on and off regimes. All of the devices are driven at a constant drain-source voltage of 1 V. The FET in Figure 6.13(a) operates in enhancement mode, with a threshold voltage of 1 V and an on-off ratio of 10⁴. In contrast, the two FETs shown in Figure 6.13(b) and (c) operate in depletion mode, with threshold voltage of -2.5 V and -5 V respectively. Both have on-off ratios of $\approx 10^3$. The devices which operate in depletion mode have a significantly increased hysteresis compared to those operating in enhancement mode, as is the case for the PEI(1300) FETs displayed in Figure 6.12.



Figure 6.13: Transfer characteristics of hierarchical nanowire FETs synthesised using 8 mM of PEI(800). The devices are measured at $V_{DS} = 1$ V.

The device characteristics of the FETs shown in Figure 6.13 are summarised in Table 6.4. The minimum subthreshold swing values of these devices is significantly larger than all of the FETs fabricated with PEI(2000), and almost all of the FETs fabricated with PEI(1300) (bar FET (c), detailed in Table 6.3.) This is attributed to the increased number of nanowire-nanowire junctions present in the PEI(800) samples, which has a detrimental effect on the switching ability of the FET. These samples are represented

by the SEM images depicted in Figure 6.6.

Table 6.4: Summary of the device characteristics of 8 mM PEI(800) hierarchical nanowire FETs

| FET | V_{th} | $I_{on/off}$ | g _m (max.) (S) | Subthreshold Swing (mV/dec.) (min.) |
|-----|----------|-----------------|---------------------------|-------------------------------------|
| a | 1 | 104 | $1.1	imes 10^{-8}$ | 1149 |
| b | -2.5 | 10 ³ | $5.57	imes10^{-8}$ | 2061 |
| с | -5 | 10 ³ | $2.76	imes10^{-8}$ | 1219 |

6.3.4 Discussion

All of the hierarchical nanowire FETs that have been measured show obvious field dependence, and are responsive to applied gate voltages without any high-temperature annealing or post-processing. This is in stark contrast to the majority of reported ZnO nanowire FETs [113, 114, 118, 54, 11] (which are discussed in detail in Chapter 3), and hierarchical nanowire FETs which are fabricated using intersecting primary nanowires. Furthermore, some of the FETs presented here operate in enhancement mode, meaning they are depleted at zero gate voltage and must be turned on by populating the nanowires with charge carriers. In comparison, field-independent ZnO nanowires are typically highly conductive at zero gate voltage, and cannot be depleted even with very strong negative gate biases [113, 118].

As the secondary nanowires presented here have a lower electron concentration in their as-grown state than usual, there must either be fewer intrinsic electron donors to begin with, or the intrinsic electron donors are compensated for by electron acceptors. Because the nanowires have a high surface to volume ratio, it is possible to capture electrons from the bulk through surface traps. It is well-known that O- and OH- groups from the atmosphere preferentially adsorb on ZnO nanowire surfaces, depleting the nanowire by compensating intrinsic electron donors [176, 177, 178]. There is some variation in the densities of the depleting surface states between nanowires, which can even occur between nanowires retrieved from the same growth batch [179]. This variation can affect the charge carrier density of the nanowires, which may be responsible for the variation of device characteristics between the nanowire FETs fabricated here. As discussed in Section 3.5 of the thesis, surface states also strongly affect the hysteresis of nanowire FETs [180, 181], which may be responsible for the device hysteresis and variation thereof.

6.4 Persistent photoconductivity measurements

It has been well documented that the surface states which deplete ZnO nanowires can be desorbed using UV radiation, releasing trapped electrons back into the nanowire bulk [182, 183, 184, 185]. Although there is still some contention over the exact mechanism [183], it is generally accepted that photogenerated holes from electron-hole pairs recombine with surface-trapped electrons, which then causes the surface species to desorb [178, 182, 185, 96]. Thus, if a nanowire is exposed to UV radiation while being kept under vacuum, desorped surface species are pumped away before they can readsorp, giving rise to a persistent nanowire photoconductivity. The significance of the surface traps can then be determined by the magnitude of the persistent photoconductivity.

To properly gauge the impact of the nanowires' adsorbed surface states, a hierarchical nanowire FET which operates in enhancement mode (i.e. with a positive threshold voltage) was selected for persistent photoconductivity measurements. As the FET is depleted at zero gate voltage, it is assumed that the channel is fully depleted due to the surface states. The FET measured was fabricated using 4 mM of PEI(1300).

The FET was connected to a printed circuit board using copper wire and silver paste, and consequently measured inside of a cryostat. This cryostat was used solely as a vacuum chamber, and all measurements were taken at room temperature. The cryostat is fitted with a CaF₂ window, which allows for illumination of the sample with UV light. UV light was provided by 395 nm LEDs at an intensity of 8.8 mW cm⁻². The intensity of this UV source was measured before and after each exposure to ensure that there was no fluctuation in power. The pressure of the chamber was measured during pumping, and decreased from an initial pressure of 5.56×10^{-4} mbar to 1.2×10^{-4} mbar over a 35 hour period. All electrical measurements were acquired using the standard Agilent 4156C parameter analyser used elsewhere in this thesis.

Before the cryostat was pumped down and the FET illuminated with UV, transfer characteristics of the FET in atmosphere were recorded as a control. The transfer characteristics are displayed in Figure 6.14. The FET shows well-defined on-off switching in atmosphere, with a threshold voltage of \approx 2 V. The rise in the drain-source current (shown in black) with negative gate voltages is explained by the leakage current (plotted in red), likely introduced by noise and unintentional conduction pathways in the cryostat system. The absolute value of the leakage current has been plotted so that it can be shown on a logarithmic scale. The FET is driven with a V_{DS} value of 5 V, which is significantly higher than the other FETs shown in this chapter.



Figure 6.14: Transfer characteristics of the PEI(1300) hierarchical FET prior to evacuating the cryostat. The device is driven at $V_{DS} = 5$ V.

Further transfer characteristics of the same FET were taken in vacuum while the cryostat was actively pumped, but prior to UV exposure. The transfer characteristics of the FET before (black) and during (red) pumping are displayed in Figure 6.15. The measurement taken before pumping is the same transfer characteristics shown in Figure 6.14. The transfer characteristics of the FET are largely unchanged during the pumping process, with an identical threshold voltage of ≈ 2 V. However, the hysteresis of the device has increased as the gate voltage is swept in both directions. As noted before, nanowire surface states play a large role in determining the hysteresis of nanowire FETs [180, 181], which may change when the chamber is evacuated. The increased hysteresis of the device may be caused by these changes to the surface states.



Figure 6.15: Transfer curves before (black) and during (red) pumping of the cryostat. The FET is driven at $V_{DS} = 5$ V.

After the FET transfer curves were recorded before and during pumping, photoconductivity measurements were performed by illuminating the sample with the 395 nm UV source. Figure 6.16 shows the persistent photoconductivity of the FET as it is exposed to UV radiation under vacuum. The FET is driven at a drain-source voltage of 1 V with no applied gate voltage, and shows an initial dark current of 10^{-12} A. After the UV source is switched on, the drain current rapidly increases by 5 orders of magnitude. Although the wavelength of UV used (395 nm) falls outside of the bandgap of bulk ZnO, we clearly see a photoresponse corresponding to electron-hole pair production and surface species desorption, which agrees with other reports using similar wavelengths of light [186, 187].

The photoconductivity is very persistent while the vacuum is maintained, remaining stable 24 hours after the UV source is switched off. Similar to other ZnO nanowire persistent photoconductivity experiments in vacuum [182], there is no immediate drop in the current after switching off the UV light. This suggests that the photocarrier contribution to the photocurrent is minor. The contribution of the photocarriers to the photocurrent is very difficult to determine due to the indeterminate path length of the FET, but a similar order-of-magnitude calculation reported in the literature suggests that it may be as low as $\approx 10^{-4}$ nA [182]. The lack of a sharp decrease in photoconductivity when turning off the UV illumination also corroborates with other persistent photoconductivity experiments in vacuum, which are even performed using UV that falls within the bandgap of bulk ZnO [182]. After venting to atmosphere, the drain current decreases by almost two orders of magnitude over ten hours. Although the FET does not return to its initial dark current over this timescale, other nanowires in vacuum have shown similarly long decay times previously [182].



Figure 6.16: The persistent photoconductivity of the PEI(1300) hierarchical nanowire FET before, during, and after UV exposure.

6.4.1 Transfer characteristics during persistent photoconductivity

The persistent photoconductivity shows that the nanowire electron concentration rapidly increases by orders of magnitude as the depleting surface states are removed from the nanowire surface. The depleting surface states therefore act as a competing defect mechanism to the electron donors still present in the bulk of the nanowire. To determine how the depleting surface states affect the field dependence of the FET, the cryostat was re-pumped after venting and the FET was re-exposed to the UV source. The transfer characteristics of the FET were then measured while the FET was in the persistent photoconductivity state (i.e. after the UV source had been switched off, but before the cryostat was vented.)

Figure 6.17 shows the transfer curves of the FET both prior to pumping (black) and during persistent photoconductivity (red). Removing the depleting surface states has an enormous effect on the behaviour of the device. As the secondary nanowires have been stripped of the depleting surface states, there is no competing mechanism to account for the intrinsic electron donors. As such, the FET is now filled with charge carriers at zero gate voltage, causing the device to switch from operating in enhancement mode to depletion mode. The threshold voltage of the device has changed from \approx 2 V (prior to UV exposure) to \approx -13 V (during persistent photoconductivity.) While the FET is measured during persistent photoconductivity, the on-current increases by almost two orders of magnitude, and the hysteresis also increases significantly.

As discussed in the overview of hysteresis and passivation in ZnO nanowires (Section 3.5), surface states at the nanowire/atmosphere interface and the nanowire/substrate interface can influence the hysteresis of the FET. The persistent photoconductivity measurements show an increase in hysteresis, despite the fact that absorped charge traps are being expunged from the nanowire surface. This may suggest that the nanowire/substrate interface states act in opposition to the surface charge traps, and the balance is tipped with their removal. Another possibility is that the UV light releases trapped charges on the SiO₂ surface, which are now mobile. In any case, it is likely that the surface states have an impact on the hysteresis, although it is very difficult to determine exactly what causes the increase under persistent photoconductivity conditions.

Despite the fact that the threshold voltage has significantly shifted in the negative direction, the hierarchical nanowire FET still shows clear field dependence. Furthermore, the threshold voltage of \approx -13 V is not particularly low when compared to other hydrothermal nanowire FETs in the literature (summarised in Table 3.1 in Chapter 3), even when they have been annealed and have comparable diameters to the hierar-



Figure 6.17: Transfer characteristics of the hierarchical nanowire FET taken before pumping (black) and during persistent photoconductivity (red).

chical nanowires [113, 118]. This suggests that while the surface states play a role in determining the threshold voltage of the device, they may not determine the field dependency of the nanowires.

6.4.2 Persistent photoconductivity of primary nanowire FETs

FETs which are bridged with the primary nanowire portion of the hierarchical nanowires show no field dependence, as demonstrated in Figure 6.10. To gauge the impact of the depleting surface states on these devices, a primary nanowire bridged device was connected to the cryostat and persistent photoconductivity measurements were performed. Figure 6.18 shows the persistent photoconductivity of the primary nanowire FET. The FET exhibits a stable dark current before the UV is switched on, after which the drain current rapidly increases by an order of magnitude. This photoconductivity is largely persistent after the UV source is turned off, although a small decrease is observed over several hours with further small decreases once the vacuum is opened. While both the primary and secondary nanowire FETs show a persistent response to UV illumination, the marked difference in their magnitude suggests that depleting surface states play a much larger role in determining the charge carrier concentration of the nanowires.



Figure 6.18: Persistent photoconductivity measurements of a primary nanowire FET. The device is driven at $V_{DS} = 0.1$ V.

6.5 Conclusion

In conclusion, a PEI-mediated hydrothermal growth is shown to reliably fabricate lateral hierarchical nanowires using 3 different molecular weights of PEI. These hierarchical nanowires consist of large primary nanowires, which abruptly terminate in significantly thinner, longer secondary nanowires. Both of the primary and secondary nanowires are monocrystalline, and share an epitaxial common c-axis at their interface. The growth substrates are designed to also function as device substrates, which allow for direct measurements of the primary and secondary nanowires as FETs. FETs which are bridged by the secondary nanowire portion of the hierarchical nanowires show clear field dependence, despite being measured immediately after growth without any annealing or post-growth processing. In contrast, FETs bridged with the as-grown primary nanowires show no field dependence, which is consistent with other reports of ZnO nanowire FETs in the literature.

The secondary nanowire FETs show good device performance, with on-off ratios of 10^3 to 10^5 and threshold voltages between -7.5 V to 5 V. Persistent photoconductivity measurements show that the secondary nanowire surfaces are covered in depleting surface states, which play a strong role in determining the threshold voltage of the FETs. These depleting surface states act as a competing defect mechanism to intrinsic electron donors, which also plays a role in determining the field dependence of the nanowires.

Chapter 7

Modular growth of lateral hierarchical ZnO nanowire FETs

7.1 Introduction

One of the main struggles of creating ZnO nanowire field-effect transistors is to synthesise nanowires which exhibit field dependence [11]. In Chapter 6, hierarchical nanowires which showed clear field dependence were successfully synthesised. These hierarchical nanowires did not require any post-growth annealing or other processing, which is significantly different from most ZnO nanowires [11]. The mechanisms behind the growth of hierarchical ZnO nanowires (discussed in Chapters 6 and 5) are attributed to an initial growth phase driven by regular interactions of hydrothermal precursors, which is then followed by a distinctly different second growth phase mediated exclusively through PEI-Zn²⁺ complexes. These complexes decay through a Mannich reaction, releasing Zn²⁺ ions back into solution [75]. These two growth phases take place during a single hydrothermal growth, which allows for the rapid fabrication of field-dependent ZnO nanowire FETs.

Instead of relying on a PEI-mediated growth process over a single hydrothermal growth, it would be ideal to separate the two distinct growth phases into two modular hydrothermal growths, neither of which necessarily rely on PEI. The modular synthesis of hierarchical nanowires is achieved in this chapter, with the first growth responsible for the growth of primary nanowires, and the second growth responsible for the growth of secondary nanowires. These hydrothermal growths initially contain PEI, although the optimum growth recipes for maximum hierarchical nanowire yield neglect PEI entirely. The primary modular hydrothermal growth is preheated for 20 or 60 minutes, and lasts for 6 hours. These optimised recipes yield hierarchical

nanowires 50% of the time.

Electrical measurements of the twice-grown hierarchical nanowires show that they are relatively field-independent when measured as-is, unlike the PEI-mediated hierarchical nanowires grown in Chapter 6. Field dependence is achieved by measuring in a liquid environment using de-ionised H_2O as a liquid gate dielectric, in conjunction with a reference electrode or a back gate electrode. Liquid-gated hierarchical nanowire FETs show good device characteristics when used in conjunction with a reference electrode, with threshold voltages as low as 0.45 V and subthreshold swings as low as 82 mV/decade. A liquid back-gated FET also shows reliable field dependence, with an on-off ratio as high as 10^5 , and a threshold voltage of 8 V. The implementation of a liquid gate may allow the FETs to be used as sensors for biological applications.

7.2 Hierarchical nanowire synthesis

7.2.1 Theory and motivation

In Chapter 5, vertical hierarchical ZnO nanowires were grown by including 4 mM of PEI(1300) in the hydrothermal growth solution. These nanowires were composed of a 'primary' nanowire which nucleated directly from the substrate, and a comparatively finer 'secondary' nanowire, which nucleated from the terminating end of the primary nanowire. The proposed hierarchical nanowire growth mechanism involved two distinct growth phases. The first growth phase was responsible for synthesising the primary nanowires, and was driven by direct and indirect crystallisation of ZnO from Zn^{2+} ions. In turn, the secondary nanowires were synthesised during the second growth phase, which was driven exclusively by the crystallisation of Zn^{2+} ions supplied in the growth solution by a PEI-mediated Mannich reaction [75]. Likewise, PEI-mediated lateral hierarchical ZnO nanowires were grown in Chapter 6, and were also attributed to a combination of primary and secondary growth phases, with the second growth phase driven by the crystallisation of Zn²⁺ ions.

In both instances, the vertical and lateral hierarchical ZnO nanowires are grown over the course of a single hydrothermal growth, with complex interactions between the PEI and regular precursors causing the two distinct growth phases. Although coupling the two growth phases in a single hydrothermal growth has some advantages, it would also be useful to separate the two growth phases into two distinct hydrothermal growths. This would enable a greater level of control over each respective growth phase, thereby offering more control over the growth of the primary and secondary nanowires. Furthermore, if the two growth phases could be separated, PEI could possibly be omitted entirely, which could possibly lead to a more facile and predictable hierarchical nanowire synthesis route.

The synthesis of lateral hierarchical nanowires through two separate hydrothermal growths is shown schematically in Figure 7.1. Lateral growth substrates (shown in Figure 7.1(a)) first undergo a hydrothermal growth resulting in primary nanowires which extend from the exposed ZnO sidewall, as shown in Figure 7.1(b). The samples then undergo a second hydrothermal growth, causing secondary nanowires to hierarchically nucleate from the tips of the primary nanowires, as in Figure 7.1(c). In order to achieve a significant difference in diameter between the primary and secondary nanowires, the second hydrothermal growth must use a drastically reduced precursor concentration [64, 97, 188, 48]. This occurs naturally for hierarchical nanowires grown using PEI in a single hydrothermal growth, as the PEI-mediated Zn^{2+} ion source driving the second growth phase occurs only after the bulk of the original precursors have been exhausted. If the initial hydrothermal growth is simply repeated twice, the growth solution is effectively refreshed, which is a well-established technique to grow longer nanowires while maintaining or increasing their diameter [10, 69, 189].

7.2.2 Experimental details

To initially test the viability of a two-growth approach to synthesizing lateral hierarchical nanowires, growth substrates were fabricated by patterning SiO₂ (100 nm)/p⁺-Si substrates with ZnO/Ti (80/50 nm) electrodes. Each substrate had 8 electrodes, with electrode gaps ranging from 10 μ m to 80 μ m in 10 μ m increments. Primary nanowires were hydrothermally grown from these substrates in a hydrothermal growth solution containing 25 mM of Zn(NO₃)₂/HMT, along with 3 mM of PEI (M_W = 2000 g/mol). The hydrothermal growth solution was preheated for 20 minutes at 95 °C before the patterned growth substrates were placed into the growth solution. The patterned growth substrates were kept in the growth solution for 19 hours before being removed, and were consequently rinsed with de-ionised water and dried with N₂ at room temperature.

After the first hydrothermal growth was completed, the secondary hydrothermal growth solution was prepared using 2.5 mM of $Zn(NO_3)_2/HMT$ and 0.3 mM of PEI ($M_W = 2000 \text{ g/mol}$). The concentration of these precursors is 10% of the concentration used in the primary nanowire growth. The second hydrothermal growth solution was preheated for 3 hours at 95 °C before the primary nanowires were introduced. After 19 hours, the now twice-grown substrates were removed from the second



Figure 7.1: (a) The growth substrates used in the fabrication of twice-grown hierarchical nanowires. ZnO seed layers sit atop an SiO₂/Si ensemble, and are in turn capped with Ti. This only leaves the sidewall of the ZnO seed layer exposed. ZnO nanowires nucleate and grow from the exposed ZnO sidewall during the first hydrothermal growth (b). A second hydrothermal growth follows, resulting in the growth of secondary nanowires from the primary nanowires, as shown in (c).


Figure 7.2: (a) A top-down view of an interelectrode channel showing hierarchical nanowires growing from the ZnO seed layer. Large primary nanowires, synthesised during the first hydrothermal growth, grow from the sidewall of the ZnO seed layer. These primary nanowires terminate in much thinner secondary nanowires, which are synthesised during the second hydrothermal growth. The inset to the figure shows a broader picture of the two electrodes. (b) A closer view of the primary/secondary nanowires. The secondary nanowires lie flush with the SiO₂ substrate.

growth solution, rinsed with de-ionised water, and dried with N₂ at room temperature.

The fabrication process of the substrates is described in detail in Chapter 4, while the hydrothermal growth and roles of the various precursors are discussed at length in Chapter 2.

7.2.3 Initial results

Figure 7.2 shows an SEM image of one of the twice-grown samples from a top-down perspective. Large primary nanowires, synthesised during the primary growth, nucle-ate laterally in a 180° arc from the ZnO sidewalls, as shown in the inset to Figure 7.2(a). It is believed that partially etching the ZnO layer under the Ti capping layer can restrict this effect for optimised devices, but this has not performed on these samples. These primary nanowires have lengths of 15 to 20 μ m and diameters of 0.5 - 1 μ m. Secondary nanowires, 20 - 50 nm in diameter and up to 10 μ m in length, hierarchically nucleate from the tips and surfaces of the primary nanowires. A portion of these secondary nanowires lie flush along the dielectric of the substrate, clearly shown in Figure 7.2(b), capable of bridging pairs of electrodes.

Figure 7.3 shows the sample from a cross-sectional perspective. The cross-section of a single channel is shown in Figure 7.3(a). At this scale, the two arcs of primary

nanowires growing from the ZnO sidewalls (indicated by arrows) are clearly visible, while the ZnO/Ti seed layer is too thin to be clearly discerned. A close-up of a single primary growth arc is shown in Figure 7.3(b), showing the 180° arc the primary nanowires make. Figure 7.3(c) shows the centre of the arc, where the primary nanowires have nucleated from. The primary nanowires share a common ZnO agglomerate that has grown from the seed layer during the primary growth. Secondary nanowires can be seen hanging down from the primary nanowires. A close-up of the seed layer and common ZnO agglomerate is shown in Figure 7.3(d). Nucleation of ZnO at the sidewall during the primary growth has forced the ZnO seed layer to partially delaminate and curl in on itself, which allows the primary nanowires to grow in an arc rather than confined strictly to the surface of the substrate.

Hierarchical nanowires were isolated from growth substrate and placed on carbon grids to investigate their crystal structure. Transmission electron microscopy (TEM) measurements were carried out by Dr. Jerome Majimel and Uli Castanet. Figure 7.4 shows TEM images of the twice-grown hierarchical nanowires. The primary nanowires, shown in Figure 7.4(a) and (b), are coated in an agglomerate of ZnO nanoparticles from the second hydrothermal growth. These nanoparticles have different crystal orientations in relation to the primary nanowires, as evidenced by the high-resolution image shown in Figure 7.4(b). These nanoparticles give the primary nanowires a feathery texture, rather than the smooth texture of a single crystal that would be expected from a primary-only nanowire growth. The interface between a primary and secondary nanowire is shown in Figure 7.4(c). The secondary nanowire consists of the same ZnO nanoparticles that coat the primary nanowires. These nanoparticles have self-assembled and orientated into a nanowire structure. The individual crystal orientations of the nanoparticles are shown in Figure 7.4(d). From these measurements, it is clear that the secondary nanowires are polycrystalline, rather than single crystals, and also have a similar texture to the primary nanowires.

7.3 Optimisation for reproducibility and device fabrication

The initial results in the previous section clearly demonstrate the feasibility of growing hierarchical nanowires through two separate hydrothermal growths. This is in contrast to the hierarchical nanowires synthesised in Chapters 5 and 6, where the primary and secondary growth phases occur over the course of a single hydrothermal growth. In these instances, the primary nanowire growth is driven by a multitude of Zn^{2+} ion complexes directly and indirectly crystallising, while the second growth phase is driven exclusively through the gradual release of PEI-chelated Zn^{2+} ions via a



Figure 7.3: (a) A cross-section of a ZnO electrode after undergoing two consecutive hydrothermal growths. The primary nanowires are indicated by arrows. (b) A closer view of a single edge of the ZnO seed layer. The primary nanowires clearly grow in a 180° arc. (c) A higher magnification image of the centre of the primary growth arc. The primary nanowires all collectively share an agglomeration of ZnO as their nucleation point, which has grown over the course of the primary growth. (d) A high-magnification image of the interface between the common ZnO agglomerate and the seed layer. The seed layer has partially delaminated during the primary growth, exposing the underside of the layer to the hydrothermal growth solution.



Figure 7.4: (a),(b) Transmission electron microscopy images of primary nanowires, showing nanoparticle accumulation on the surface. (c),(d) The primary/secondary nanowire interface.

Mannich reaction [75]. However, while the primary and secondary growth phases are assigned to separate hydrothermal growths in Section 7.2, PEI was included in both of these hydrothermal growths. This naturally raises the question of whether PEI is necessary in either or both of the hydrothermal growths for the reproducible synthesis of secondary nanowires. As the two hydrothermal growths are distinct and separate, we can elucidate the role of PEI by including or excluding it in each respective growth, and noting whether there is any effect on the reproducibility or morphology of the secondary nanowires.

Similarly, the separation of the primary and secondary growth phases allows us to vary the parameters of each hydrothermal growth individually, and investigate the effect on the reproducibility of the hierarchical nanowire synthesis. In this section, the preheat time of the primary growth is systematically varied, along with the inclusion of PEI in both the first and second hydrothermal growth. This ultimately allows for the reliable synthesis of hierarchical nanowires and the fabrication of electronic devices.

7.3.1 Hydrothermal growth details

Primary hydrothermal growths were prepared using 25 mM of HMT/Zn(NO₃)₂ as basic precursors, as in Section 7.2. To test the influence of PEI on the primary growth, 3 mM of PEI(M_W = 2000 g/mol) was included with the standard precursors for those grown with PEI, and was omitted for those grown without PEI. The primary hydrothermal growth bottles were preheated for 0 minutes, 20 minutes, or 60 minutes, both with and without PEI. Four growth substrates (identical to those described in Section 7.2) were placed into each of these primary hydrothermal growth bottles after their respective preheat, and were removed after 6 hours. This comparatively short growth time was chosen to decrease the likelihood of unintentionally bridging pairs of electrodes with primary nanowires by reducing their overall length. The short growth time also ensures that secondary nanowires don't grow through a PEI-mediated second growth phase during the primary growth, as in Chapters 5 and 6. Three identical bottles of each preheated hydrothermal growth (with and without PEI) were prepared to test for the overall reproducibility of the hierarchical nanowire synthesis.

After the primary hydrothermal growth was completed, the growth substrates were removed and rinsed with DI water. Secondary hydrothermal growth solutions were prepared using 2.5 mM of HMT/Zn(NO₃)₂, both with and without 0.3 mM of PEI($M_W = 2000 \text{ g/mol}$). These secondary hydrothermal growth bottles were all preheated for 3 hours. Each individual growth substrate from a given primary growth was placed into its own secondary growth bottle. As each primary growth



Figure 7.5: The growth flow of a particular fixed set of variables for the primary growth. Four substrates are included in a single primary growth bottle, which are then individually placed into secondary growth bottles.

bottle contained four separate growth substrates, four secondary growth bottles were prepared for each primary growth bottle. Two of these secondary growth bottles contained PEI along with the standard precursors, while the other two secondary growth bottles omitted PEI. This allowed for comparisons of growths both with and without PEI in the primary growth, and with and without PEI in the secondary growth. The individual second-grown substrates are removed from the hydrothermal growth solution after 19 hours, rinsed in DI, and dried with N₂ at room temperature. The process flow of the primary and secondary growth is shown in Figure 7.5 for clarity.

7.3.2 Effect of preheat on the primary growth

As the primary hydrothermal growth solution heats up after being placed in the waterbath, the dissolved precursors are thermodynamically driven to interact with one another. This causes the Zn^{2+} ions to form complexes, leading to the direct and indirect crystallisation of ZnO, as described in Chapter 2. Supersaturated ZnO within the solution can homogeneously precipitate out of solution, or heterogeneously

nucleate at energetically favourable sites [190, 58, 40]. While the hydrothermal growth solution is preheating in the absence of a growth substrate, heterogeneous nucleation can only occur on the glass walls of the growth bottle. This ensures that the majority of ZnO nucleation during the preheat phases is homogeneous, with ZnO directly precipitating out of solution, with the rate of homogeneous precipitation declining over time as the precursors are exhausted.

In comparison, when the hydrothermal growth takes place with growth substrates present in the solution, ZnO can heterogeneously nucleate onto energetically favourite sites on the substrate. These sites are typically along the exposed ZnO sidewall of the patterned ZnO seed layers, but ZnO can also unintentionally nucleate directly on the surface of Ti capping layer or the SiO₂ dielectric. The likelihood of nucleation and growth of ZnO on the SiO₂ surface is related to the concentration of precursors in the growth solution This means that the longer the growth solution has been preheated for, and the more Zn²⁺ ions have been complexed or precipitated out of solution, the less overall ZnO nucleation there is on the substrate surface. Conversely, if the hydrothermal growth solution has been preheated for an inadequate amount of time, or hasn't been preheated at all, there is often a correspondingly high amount of ZnO precipitation on the SiO₂ surface.

Figure 7.6 shows optical images of primary nanowires grown after a 0 minute, 20 minute, and 60 minute preheat, all grown without PEI over 6 hours. The comparative difference in the amount of surface-nucleated ZnO (or 'debris') is clearly obvious, with a significantly larger amount of debris present on the substrate grown without a preheat. The amount of debris present on the 20 minute preheated substrate is substantially less, and in turn there is less on the substrate preheated for 60 minutes. The sample grown after a 60 minute preheat has the least amount of debris present on the SiO₂ surface, although is not totally devoid from it.

Debris are ordinarily undesirable when fabricating lateral ZnO nanowire devices as there is no way to control where they nucleate. This means they can nucleate and grow in and around inter-electrode gaps, potentially forming unintentional conduction pathways between transistor sources and drains. These conduction pathways can extend over long distances if the debris have nucleated at a sufficiently high density, which can even cause electrical shorts between source electrodes and the back gate of the transistor. These back gate shorts are presented in Section 7.4. In addition to potentially fouling the channels between source, drain, and gate electrodes, debris on the SiO₂ surface pose a unique problem for the twice-grown hierarchical nanowires presented in this chapter. When primary-grown substrates are placed in the second growth, secondary nanowires can preferentially nucleate from the ZnO nanocrystals



Figure 7.6: Optical images showing the relationship between the preheat of the primary growth and the density of surface-nucleated debris. The left-most sample is grown without a preheat, the middle sample is grown with a 20 minute preheat, and the right-most sample is grown with a 60 minute preheat. PEI was omitted for all of the growths.

on the SiO_2 surface, rather than from the primary nanowires themselves. As such, the surface-nucleated debris can compete with the primary nanowires for secondary nanowire growth, which can reduce the overall yield of hierarchical nanowires. These yields are presented in the next subsection.

7.3.3 Results and discussion

Systematically varying the preheat time of the primary growth, along with the inclusion/exclusion of PEI in the primary and secondary growths, can lead to drastic differences between the primary/secondary nanowires. In terms of reproducible hierarchical nanowire synthesis, the results of a given primary/secondary growth combination fall into one of three categories:

- No secondary nanowire growth No secondary nanowire growth from any of the nanowires/nanostructures grown during the primary growth
- Secondary nanowire growth from debris Secondary nanowire growth exclusively from ZnO nano/microcrystals which have nucleated on the SiO₂ surface of the substrate during the primary growth. Primary nanowires remain unchanged.
- Hierarchical nanowire growth Successful nucleation and growth of secondary nanowires from the primary nanowires, as in Section 7.2. Secondary nanowires may also have nucleated and grown from ZnO nano/microcrystals which have nucleated on the substrate surface during the primary growth

Hierarchical nanowires are ultimately synthesised with the intention of bridging source/drain electrodes, thereby fabricating electronic devices. To this end, any growth that results in either no secondary nanowires or secondary nanowires confined to debris is considered a failure, while hierarchical nanowire growth is considered a success.

The results of the varied growths are shown in Table 7.1. Samples which showed no secondary growth are left blank and coloured red. Samples which showed secondary nanowire growth exclusively on surface-nucleated debris are labelled 'Debris' and coloured yellow. Finally, samples which showed successful hierarchical growth are labelled 'Hier. NWs' and coloured green. The top row of the table indicates the preheat time of the primary growth, and whether or not PEI is included in the growth. The left-most column indexes the individual growth substrates. As described in Section 7.3.1, three separate primary growth bottles were prepared for each set of primary growth variables (namely the preheat time and the inclusion/exclusion of PEI.) The Bx label refers to which of these three bottles the individual substrate was first grown

in. This allows us to compare twice-grown substrates which were in the same primary growth bottle, or were grown in different bottles using ostensibly identical recipes. Each primary growth bottle contained four samples, which were then individually regrown in separate growth bottles during the second hydrothermal growth. The second number in the left-most column labels uniquely identifies these substrates. The label also indicates whether PEI was included in the second hydrothermal growth, and is omitted for those which did not. The overall success rate of hierarchical nanowire synthesis by primary growth is shown in the bottom row of the table. This success rate only considers hierarchical nanowire growth (i.e. secondary growth from the primary nanowires), and disregards secondary nanowire growth from surface-nucleated debris.

Table 7.1: Yield results from secondary nanowire growths. Primary growths are indexed by column while secondary growths are indexed by row. Entries left blank failed to grow any secondary nanowires. 'Debris' refers to secondary nanowire growth exclusively from surface-nucleated debris. 'Hier. NWs.' refers to successful secondary nanowire growth from the primary nanowires.

| | 0 min. | 0 min. | 20 min. | 20 min. | 60 min. | 60 min. |
|--------------|-----------|-----------|-----------|-----------|-----------|-----------|
| | (PEI) | (No PEI) | (PEI) | (No PEI) | (PEI) | (No PEI) |
| B1 - 1 (PEI) | Hier. NWs | Debris | Debris | Hier. NWs | Hier. NWs | Debris |
| B1 - 2 (PEI) | Hier. NWs | | Hier. NWs | Debris | Hier. NWs | Debris |
| B1 - 3 | | Debris | | Hier. NWs | Hier. NWs | Hier. NWs |
| B1 - 4 | | | | Hier. NWs | | Hier. NWs |
| B2 - 1 (PEI) | Hier. NWs | Debris | Debris | Debris | Debris | Debris |
| B2 - 2 (PEI) | Hier. NWs | | Debris | Debris | Hier. NWs | Debris |
| B2 - 3 | | Debris | Debris | Hier. NWs | | Hier. NWs |
| B2 - 4 | | | | Hier. NWs | | Hier. NWs |
| B3 - 1 (PEI) | Debris | Debris | Debris | Debris | Debris | Hier. NWs |
| B3 - 2 (PEI) | | Debris | Hier. NWs | Debris | Debris | Debris |
| B3 - 3 | | | | | | |
| B3 - 4 | | Hier. NWs | Debris | Hier. NWs | | Hier. NWs |
| Success rate | 33% | 8.3% | 16.6% | 50% | 33% | 50% |

There is a wide variety of overall success rates of hierarchical nanowire growth for the different combinations of primary and secondary growth conditions. The smallest yield of hierarchical nanowires occurs when the primary growth isn't preheated and PEI is omitted, with only one sample (B3- 4) successfully growing hierarchical nanowires. Six of the other substrates show secondary nanowire growth exclusively from surface-nucleated debris. This competition between the debris and primary nanowires for secondary nanowire nucleation seriously affects the overall yield of hierarchical nanowires, suggesting that preheated primary growths without PEI are unsuited for device fabrication.

When PEI is included in the unpreheated primary growth, the overall yield of hierarchical nanowires increases to 33%, with secondary nanowire growth exclusively from debris restricted to a single sample. Although the overall yield of secondary nanowires from either primary nanowires or debris is lower, with a total of 7 samples showing no secondary nanowire growth from either, the rate of successful secondary nanowire growth from the primary nanowires is improved. These results suggest that the presence of PEI in the primary growth increases the preferentiality of secondary nanowire nucleation from the primary nanowires.

Both 20 minute preheated and 60 minute preheated primary growths without PEI have the equal highest rate of hierarchical nanowire fabrication, with overall yields of 50%. Each primary growth recipe yields 6 samples with hierarchical nanowires and 5 samples with secondary nanowires from debris, leaving only one sample with no secondary nanowire synthesis. In both cases, 5 of the 6 hierarchical nanowire samples are fabricated without PEI in the second growth, with 1 sample grown using a PEI-mediated second growth. This clearly demonstrates that PEI isn't necessary for hierarchical nanowire synthesis when two separate hydrothermal growths are used.

When PEI is included with the 20 minute preheated primary growth, the frequency of samples showing any secondary nanowire synthesis decreases. The yield of successful hierarchical nanowires is drastically decreased from 50% to 16.6%, while the rate of secondary nanowire nucleation exclusively on the surface debris slightly increases from 41.6% to 50%. Four samples show no secondary nanowire growth when PEI is included with the growth solution, compared to only a single sample when PEI is omitted from the growth solution. Secondary nanowire synthesis is also reduced overall when PEI is included in the 60 minute preheated primary growth. The reduction of successful hierarchical nanowire synthesis is not as drastic, decreasing from 50% to 33.3%. However, secondary nucleation from surface debris decreases from 41.6% to 25%. Similarly, the number of samples showing no secondary nanowire growth increases to 5 samples when PEI is included in the primary growth, compared to only a single sample when PEI is omitted from the primary growth. These comparisons clearly indicate that at least as far as hierarchical nanowire synthesis is concerned, omitting PEI from the primary growth leads to increased yields and better reproducibility.

It is also interesting to note the correlation between the success rate of hierarchical nanowires and the inclusion/exclusion of PEI in both the primary and secondary growths. When PEI is included in only the primary or only the secondary growth, the yield of secondary nanowires drastically drops. However, the yield improves when PEI is included in *both* the primary and secondary growths. Of the 23 hierarchical nanowire samples grown in the preheat/PEI experiments, only 3 are grown with mismatched PEI inclusion in the primary/secondary growth. This may suggest a different nucleation and growth mechanism between primary/secondary nanowires grown with and without PEI.

Figure 7.7 shows SEM images of a hierarchical nanowire device fabricated using the optimised recipe of a 20 minute preheated primary growth, with the PEI omitted from the primary and secondary growths. The primary nanowires are noticeably shorter than the initial devices fabricated previously in Section 7.2 due to the shorter primary growth time of 6 hours, versus 19 hours. Dense networks of secondary nanowires lie in the channel gaps, bridging the source/drain electrode pairs. This allows us to perform electrical measurements on the secondary nanowires and to utilise them as field-effect transistors.



Figure 7.7: (a) A pair of electrodes bridged by hierarchical nanowires fabricated using a primary growth preheated for 20 minutes. PEI was omitted from both the primary and secondary growths. (b) A close-up of the device channel. A high density of secondary nanowires bridge the channel gap, allowing us to perform electrical measurements.

7.4 Electrical measurements

The architecture of the growth substrates used to grow hierarchical nanowires allows for electrical measurements of pairs of electrodes bridged by secondary nanowires. Each electrode consists of a thin film of ZnO capped with Ti. If the channel of the electrodes is bridged, voltage can be applied across the Ti capping layers, passing current through the bridging nanowires. The SiO₂ (100 nm)/p-Si substrate can then be used as a dielectric and back gate respectively, allowing the pair of electrodes to be used as a source and drain pair. The hierarchical nanowire substrates therefore double as device substrates. All electrical measurements presented in this section are performed using a Rucker and Kolls probe station, coupled with an Agilent 4156C parameter analyser. The experimental details of the electrical measurement set-up are described in detail in Chapter 4.

7.4.1 Unpreheated primary growths and leakage through debris

In addition to affecting the successful yields of hierarchical nanowires, the preheat of the primary growth has a strong effect on the density of debris which nucleate on the SiO₂ surface of the substrate (as discussed in Section 7.3.2.) The presence of surface debris can drastically affect the device performance by unintentionally establishing conduction paths between source/drain electrodes. In extreme cases, conduction paths can be introduced between source/gate electrodes by debris forming a continuous pathway across the SiO₂ substrate, which reach to the very edge and make contact to the back electrode. While unintentionally bridged source/drain pairs are relatively easy to observe using optical or electron microscopy, leaks between the source and gate electrodes are typically only revealed when the device is electrically measured.

As shown in Figure 7.6, samples grown with a primary growth preheated for 20 minutes or 60 minutes show debris on the substrate surface, although the overall density of the debris is too low to establish a conduction path between source electrodes and the transistor backgate electrode. However, this is not the case for samples grown using unpreheated primary growths. Figure 7.8 shows the transfer characteristics (I_{DS}-V_{DS} measurements) of the 30 μ m channel of sample 0m (PEI) - B2 -2 (PEI), indexed by the convention established in Table 7.1. The device is driven at V_{DS} = 0.1 V, while the gate is swept from V_G = -10 V to 10 V. The device shows a strong leakage current (I_G) of 1.4 ×10⁻⁷ A at V_G = +/- 10 V due to the surface-nucleated debris. This excessively high leakage current precludes this sample from being used as an FET, even if source and drain electrodes are bridged by hierarchical nanowires.



Figure 7.8: Transfer characteristics of a sample grown with an unpreheated primary growth, with PEI included in both the primary and secondary growths. The FET is driven at $V_{DS} = 0.1$ V. This particular device measured is the 30 μ m channel of B2 - 2. The very high leakage current is caused by the high density of debris which has nucleated on the SiO₂ surface.

7.4.2 Hierarchical nanowire device measurements

Hierarchical nanowires grown with a 20 minute or 60 minute preheated primary growth don't suffer from the surface-nucleated debris to the same extent as those grown with unpreheated primary growths. This allows for output characteristic measurements ($I_{DS} - V_{DS}$) and transfer characteristic measurements ($I_{DS}-V_G$) without unintentionally measuring debris which bridge the source/drain or source/gate electrodes.

All of the hierarchical nanowire FETs fabricated in Chapter 6 showed consistent field dependence, without the need for post-growth annealing or processing. In contrast, only some twice-grown hierarchical nanowire FETs in this chapter show consistent and reliable field dependency. One of these devices is shown in Figure 7.9. This device is grown using a 20 minute preheated primary growth, without any PEI in the primary or secondary growths. The transistor is driven at $V_{DS} = 1$ V. This FET shows obvious field dependency, with an on-off ratio of 10⁴, a threshold voltage of ≈ 0 V, a transconductance of 0.75 nS, and a subthreshold swing of 0.56 V/decade. The leakage current in this device is negligible, remaining below 10^{-10} A. These device characteristics are comparable to the hierarchical nanowire FETs fabricated in a single growth in Chapter 6, although the magnitude of the on-currents differ by 2 orders of magnitude.



Figure 7.9: Transfer characteristics of a field-dependent FET fabricated using twicegrown hierarchical nanowires. The primary growth was preheated for 20 minutes. Neither the primary or secondary growths included PEI.



Figure 7.10: Diode measurements of a device fabricated using a 60 minute preheated primary growth. Neither the primary or secondary growths contained PEI. The measurement is taken at $V_G = 0$ V. The low current at high drain/source voltages suggest a low concentration of charge carriers.

In contrast, other devices show limited field dependence. While a lack of field dependence is typically due to excessive charge carriers caused by unintentional doping, nanowires with low charge carriers can also show a lack of field dependence. Figure 7.10 shows a diode measurement (taken at $V_G = 0$ V) of one such device. This device is fabricated using a 60 minute preheated primary growth, with PEI excluded from both the primary and secondary growths. Although the device does show conduction, the drain/source current is below 1 nA at $V_{DS} = 10$ V, which suggests that the hierarchical nanowires are highly resistive.

The extent of the depletion in the hierarchical nanowires significantly affects the transfer characteristics of the device, shown in Figure 7.11. The device is driven at a high voltage of $V_{DS} = 10$ V, which is necessary due to the high resistivity of the FET. Despite the high resistivity of the FET, the device shows some measure of field dependence, albeit with poorly defined on and off regimes. The FET has an on-off ratio of $\approx 10^3$, and a threshold voltage of -5 V. This suggests that the FET is operating in depletion mode, despite the high resistance of the device. The relatively low charge carrier concentration of the hierarchical nanowires is therefore unlikely to be caused by the presence of depleting surface states or other electron traps, which would necessitate a positive threshold voltage to switch the FET from off to on. Although the device shows a well-defined subthreshold regime, the drain-source current continues to rise



Figure 7.11: Transfer characteristics of the FET also measured in Figure 7.10. The primary growth is preheated for 60 minutes. Neither the primary or secondary growths contained PEI. The measurement is taken at $V_{DS} = 10$ V.

approximately one order of magnitude between $V_G = -2.5$ V to 20 V. The maximum transconductance of the device is 34 nS, and the minimum subthreshold swing is 2.14 V/decade.

Despite the above devices showing some measure of field dependence, most of the hierarchical nanowire FETs do not respond to applied gate voltages as-is. However, their field dependence can be significantly improved by utilising a liquid dielectric instead of (or in combination with) the SiO_2 layer. These measurements are described in the next subsection.

7.4.3 Liquid gating measurements

Transfer characteristics of the modular hierarchical nanowires in the previous section utilised the 100 nm thick SiO_2 layer from the substrate as the gate dielectric, although this was largely ineffective. In this section, liquid gating measurements are performed using de-ionised water as a liquid dielectric. In this configuration, the dielectric liquid and nanowires are in direct contact with each other. This covers the entire surface area of the nanowires, which allows for much more effective gating.

The gate electrode is typically directly in contact with the liquid, although the regular SiO_2/Si dielectric/gate electrode can also be used in conjunction with the

liquid dielectric. When a voltage is applied to the liquid dielectric, an electric double layer forms, which brings induced charges very close to the surface of the nanowires [191, 192, 193, 194, 195, 196]. The close proximity of the induced charge to the semiconducting nanowires causes the much more effective gating for a given gate-source voltage than would be normally achieved using the SiO₂ layer as a gate dielectric. However, extra steps must be taken to separate the source electrode from the gate electrode when using liquid gates. Despite the fact that the liquid is a highly resistive dielectric, high leakage currents can result if both the gate electrode and source electrode are in direct contact with the liquid.

Photolithography is used to insulate the source and drain electrodes from the liquid dielectric. After the hierarchical nanowires have been synthesized via the primary and secondary hydrothermal growths, the entire substrate is covered in AZ1518 photoresist. The channel area of each electrode pair is exposed to UV light using the Karl Suss MJB3 mask aligner for 60 seconds. This particularly long exposure time is used to ensure that no photoresist remains on the nanowire surface, which would insulate the nanowires from the liquid dielectric. The substrates are then developed for 40 seconds in 1:2 AZ326:DI-H₂O, rinsed in DI-H₂O, and dried with N₂ at room temperature. This photolithography step also exposes the large electrode pads to allow for contact with the probe station. Photolithography and the use of the mask aligner is explained in detail in Chapter 4.

The liquid gate architecture is shown in Figure 7.12. Deionised H_2O ($\rho \ge 18.2M\Omega \cdot cm$) is used as the liquid dielectric. The liquid gate electrode is an Ag/AgCl reference electrode (sourced from In Vivo Metric.) The source/drain electrodes and primary nanowires are encapsulated by the AZ1518 photoresist layer, ensuring that only the secondary nanowires are in contact with the liquid dielectric. Gating is achieved by either applying a voltage between the reference electrode and the source electrode, or between the back gate electrode and the source electrode.



Figure 7.12: A schematic of a liquid gated hierarchical nanowire FET. The hierarchical nanowires lie in the channel between the source and drain electrodes, and are in contact with de-ionised water. The source and drain electrodes are insulated from the water by AZ1518 photoresist.

Hierarchical nanowire FETs which were otherwise unresponsive to applied fields can show marked field dependence when the liquid gate is used. Figure 7.13 shows the output characteristics of a device fabricated using a 20 minute preheated primary growth, with PEI omitted from both the primary and secondary growths. The device is clearly conductive, but shows only a minimal difference in current when V_{DS} is sweeped at $V_G = -10$ V and +10 V using the SiO₂/back gate. This suggests that the device is largely field-independent.



Figure 7.13: Output characteristics of a hierarchical nanowire FET taken at $V_G = -10$ V and 10 V. The device is fabricated using a 20 minute preheated primary growth without PEI. The output shows minimal change between the two gate voltages, suggesting the device is field independent.



Figure 7.14: Transfer characteristics of the hierarchical nanowire FET measured in Figure 7.13. These transfer characteristics employ a liquid gate in conjunction with a reference electrode, instead of the typical SiO_2 /backgate combination. Clear field dependence is achieved when the sample is measured using a liquid gate.

In contrast, the FET shows significant field dependence when a liquid gate is used. Figure 7.14 shows the transfer characteristics of the device using a liquid gate/reference electrode combination. The device is driven at $V_{DS} = 0.1$ V. The FET shows clear switching, with a threshold voltage of ≈ 0.45 V, an on-off ratio of 10^3 , a transconductance of 6.43 μ S, and a subthreshold swing of 88 mV/decade. The effectiveness of the liquid gating is apparent in the low threshold voltage, high transconductance, and low subthreshold swing. These device characteristics are better than the SiO₂-gated hierarchical nanowire FETs presented in Chapter 6.



Figure 7.15: Transfer characteristics of a hierarchical nanowire FET taken at $V_{DS} = 1$ V using the SiO₂/backgate electrode configuration. The device is fabricated using a primary growth preheated for 20 minutes, with PEI omitted from both primary and secondary growths. The device shows very little field dependence as-is.

This behaviour is consistent across other hierarchical nanowire FETs which don't show field dependency using the traditional SiO₂/backgate configuration. Figure 7.15 shows the transfer characteristics of a device fabricated using a 20 minute preheated primary, with PEI excluded from the primary and secondary growths. The device is driven at $V_{DS} = 1$ V. Although the FET is subjected to relatively extreme gate voltages (+/- 30 V), the drain-source current shows only shows an order of magnitude change over the 60 volt range. This device is clearly unsuitable for FET applications as-is.



Figure 7.16: Transfer characteristics of the same hierarchical nanowire FET measured in Figure 7.15 taken using a liquid gate and a reference electrode. The FET is driven at $V_{DS} = 0.1$ V. Unlike the previous measurement taken with a conventional SiO₂ dry backgate, the FET shows significant field dependence.

In comparison, the device shows strong field dependence when it is liquid gated. Transfer characteristics taken using a liquid gate are shown Figure 7.16. The FET is driven at $V_{DS} = 0.1$ V. The device shows clear on and off regimes, with a threshold voltage of ≈ 0.88 V, an on-off ratio of $10^4 - 10^5$, a transconductance of 0.8 μ S, and a subthreshold swing of 82 mV/decade. These good device characteristics are comparable to the liquid-gated device shown in Figure 7.13, although the comparatively low transconductance is due to the lower on current of the device.

Although the liquid gate measurements in Figures 7.14 and 7.16 both use a reference electrode directly in contact with the liquid dielectric, it is possible to use the conventional SiO_2 /backgate electrode with the liquid in contact with the nanowires. This is less effective than directly placing a reference electrode into the liquid droplet, as the electric field needs to propagate through the SiO_2 . However, the field effect is still drastically improved in comparison to using a "dry" SiO_2 /backgate combination, as the presence of the electric double layer ensures that the induced charges in the dielectric liquid are very close to the nanowires, and that the entire surface area of the nanowires is surrounded.

Figure 7.17 shows a wet backgated measurement of the FET measured in Fig-



Figure 7.17: Transfer characteristics of the hierarchical nanowire FET measured in Figures 7.15 and 7.16. The transistor is driven at $V_{DS} = 0.1$ V. This measurement is performed with DI-H₂O on the nanowire surface, but using the regular SiO₂/substrate as the gate electrode. The SiO₂ layer consequently couples to the DI-H₂O.

ures 7.15 and 7.16. The device is driven at $V_{DS} = 0.1$ V. Although the liquid gate is only used as a secondary dielectric in conjunction with the SiO₂ layer, the FET shows clear field dependence, in stark contrast to the "dry" back-gated transfer characteristic shown in Figure 7.15. The FET operates at a threshold voltage of ≈ 8 V, with an on-off ratio of 10⁵. The device has a maximum transconductance of 1.02 μ S, and a minimum subthreshold swing of 1 V/decade. The difference in device characteristics between the liquid back-gated measurement and liquid top-gated measurement (i.e. when the reference electrode is in direct contact with the liquid dielectric) is to be expected, due to the relatively inefficient coupling of the SiO₂ layer and the DI-H₂O.



Figure 7.18: Output characteristics of the previously measured wet backgated hierarchical nanowire FET. V_{DS} is swept from 0 to 2 V. The wet backgated voltage ranges from +20 V to -15 V in 5 V steps.

The comparatively high operating voltages of the wet backgated device allow for output characteristic measurements over a wide range of gate voltages. Figure 7.18 shows the output characteristics of the wet backgated device. V_{DS} is swept from 0 to 2 V, while V_G is stepped from $V_G = +20$ V to $V_G = -15$ V in 5 V steps. These measurements clearly show the device depleted at $V_G = -15$ V. Other measurements at higher gate voltages show well-defined Ohmic and saturation regimes, with the saturation voltage increasing concurrently with V_G .

7.4.4 Context within literature

Although liquid-gated carbon nanotube FETs have attracted significant attention for their potential in biosensing applications [197], there is very little evidence in the literature of liquid-gated ZnO nanowires. There is evidence of at least one liquid-gated ZnO nanowire transistor reported [112], and another device which operates in liquid environments, but not as an FET [198]. Similarly, there are at least two reports of liquid-gated transistors which are based on bulk ZnO [199, 200]. The single report which does use ZnO nanowires as liquid-gated transistors [112] is similar to the work here, in that

the devices are fabricated in pre-defined locations using ZnO seeds. However, the transistors which are fabricated require high-temperature annealing to achieve field dependence, whereas all of the modular hierarchical nanowires measured here show field dependence without the need for an anneal. This suggests that they are also compatible with fabrication on flexible device substrates, which is a great boon for biosensing devices. As such, there is a real opportunity for this work to be furthered in biosensing applications, and a gap in the literature exists which these devices could readily fill. These ideas are extrapolated on in the conclusion of this thesis.

7.5 Conclusion

In conclusion, hierarchical nanowires can be synthesised by modular growth of the primary and secondary nanowires in two separate hydrothermal growths. This significantly generalises the growth process from the PEI-mediated single growth presented in Chapter 6, and allows for control over the individual growth phases. As the secondary growth replaces the PEI-mediated gradual release of Zn^{2+} ions, PEI can be omitted from both the primary and secondary growths if required.

Yields of hierarchical nanowires were optimised by systematically adjusting the preheat of the primary nanowire growth, the inclusion of PEI in the primary growth, and the inclusion of PEI in the secondary growth. Primary growths were carried out using 25 mM of HMT/Zn(NO₃)₂, along with 3 mM of PEI ($M_W = 2000 \text{ g/mol}$) for the PEI-included growths. The primary growths were either unpreheated, preheated for 20 minutes, or preheated for 60 minutes. The primary growth time was fixed at 6 hours. The secondary hydrothermal growths were carried out at 10% of the precursors of the primary growths, namely 2.5 mM of HMT/Zn(NO₃)₂. PEI was added at a concentration of 0.3 mM ($M_W = 2000 \text{ g/mol}$) for the secondary growths which included PEI. The secondary growth preheat was fixed at 3 hours, and the growth time was fixed at 19 hours.

These experiments either resulted in no secondary nanowire growth, secondary nanowire growth confined to surface-nucleated debris, or secondary nanowire growth from the primary nanowires (i.e. hierarchical nanowire growth.) The optimum recipes were found to be a 20 or 60 minute preheated primary growth without PEI, followed by a secondary growth without PEI. These both had an equal overall hierarchical nanowire growth yield of 50%. The inclusion of PEI in either the primary or secondary nanowire growth significantly reduced the overall yield of hierarchical nanowires.

Electrical measurements of the twice-grown hierarchical nanowires show that

they are typically field independent when measured using a typical SiO₂/backgate, unlike the PEI-mediated hierarchical nanowires measured in Chapter 6. In contrast, the hierarchical nanowire FETs show field dependence when measured when using de-ionised H_2O as a liquid gate. These measurements can be performed using a gate electrode directly in contact with the liquid, or by using the standard back-gate to couple the DI-H₂O to the SiO₂. Using a gate electrode in direct contact with the DI-H₂O results in faster and easier switching, with subthreshold swings as low as 82 mV/decade, and threshold voltages as low as 0.45 V. In comparison, the hierarchical nanowire FET measured using a liquid gate in conjunction with the back gate electrode has a threshold voltage of 8 V, with a subthreshold swing of 1V/decade.

Chapter 8

Individually contacted ZnO nanowires

8.1 Introduction

In this chapter, large numbers of vertically-grown ZnO nanowires are individually contacted using electron beam lithography to fabricate FETs. The single-nanowire devices are fabricated in batches, and three separate batches are presented here. Each batch consisted of 3 - 4 samples (device chips), with multiple nanowires contacted on each sample. All of the single nanowire devices show non-Ohmic conduction, and are modelled using back-to-back Schottky barriers consistent with metal-semiconductor-metal devices which have not been annealed after metallisation [201]. The Schottky barrier heights of two contacts on a single nanowire are shown to be strongly correlated with one another, as are the ideality factors. It is also shown that these do not depend on the diameter of the nanowire. There is no obvious relationship between the physical characteristics of the ZnO nanowire or its growth conditions with the Schottky barrier heights/ideality factors of the contacts.

Despite the large number of devices fabricated, almost all of the FETs show no field dependence regardless of whether or not the nanowires are annealed, in contrast with the literature [113, 114, 118, 54, 11]. One single FET fabricated using a vertically-grown nanowire responds to gate voltages, with an on-off ratio of 10^4 and a threshold voltage of -7 V, a transconductance of 1.65 nS, and a minimum subthreshold swing of 937 mV/decade. The field-dependent mobility of the FET is approximately 0.106 cm² V⁻¹ s⁻¹.

Another functioning FET is fabricated using a field-dependent secondary nanowire which is isolated from a hierarchical lateral FET, as described in Chapter 6. Damage to the SiO_2 dielectric of the device substrate sustained during the fabrication process negatively impacts the off current of the FET, which is 3 to 4 orders of magnitude

higher than than a typical lateral device. The contacted nanowire displays field dependence, with a threshold voltage of -6.2 V, an on-off ratio of 10^2 , a transconductance of 30.1 nS, and a subthreshold swing of 845 mV/decade. Calculations yield a mobility of 2.964 cm² V⁻¹ s⁻¹.

The chapter concludes with some outstanding questions that the results have raised, which pave the way for future work in this field of research.

8.2 Motivation and background

While the ZnO nanowire field-effect transistors fabricated in Chapters 6 and 7 show consistent field dependence and device operation, the FET device channels are complicated due to the overlapping and intersecting nanowires. These nanowire-nanowire junctions and the interfaces between ZnO seed layers and Ti source/drain electrodes could negatively impact the performance of the FETs. The conduction pathway between source and drain electrodes is shown in red in Figure 8.1, with junctions between thin films, primary nanowires, and secondary nanowires highlighted by red circles. A further issue with the device geometry used in Chapters 6 and 7 is the indeterminate channel length of the FET. Accurately knowing the length of the FET channel is necessary to measure certain device parameters, such as the field dependent mobility [103, 104, 105, 54, 106, 107]. Although the gap between the source/drain electrodes is well defined, the length of the secondary nanowires which bridge the gap can be difficult to determine, as they often bridge the electrodes in very large numbers and at a variety of angles.



Figure 8.1: A schematic showing the conduction pathway (red) between the source and drain for a lateral hierarchical FET. Junctions in the pathway are highlighted by red circles.

As such, there is a clear motivation to contact individual ZnO nanowires using electron beam lithography (EBL). Restricting the channel to a single ZnO nanowire allows for accurate measurements of the device channel and nanowire radius, which allows us to calculate device parameters such as mobilities and current densities. Decoupling the channel material from complicated bilayer electrodes also allows for cleaner interpretations of electrical measurements and the characteristics of ZnO nanowires as an electronic material.

8.2.1 Nanowire-metal contacts and Schottky barrier modelling

Unlike the multiple-nanowire FETs fabricated in Chapters 6 and 7, single-nanowire FETs are directly in contact with the source and drain metal electrodes. The metalsemiconductor contact can influence the electrical behaviour of the FET [202], which can be broadly described as Ohmic or non-Ohmic, and can vary significantly from device to device due to impurities and surface states [202]. Although individually contacted nanowires are ostensibly a simple system, the exact cause for non-Ohmic conduction is difficult to conclusively determine. Ohmic contacts to ZnO nanowires are desirable, and have been previously achieved using Ti as a contact metal [203, 204, 205, 206, 207], often in conjunction with a high-temperature anneal. However, ZnO nanowires have also been reported to form non-Ohmic contacts with Ti [208, 209, 210, 211, 212, 213, 214, 215].

Non-Ohmic Ti contact to ZnO nanowires has not been studied in much depth, and most publications which report non-Ohmic conduction do not speculate much on possible causes behind the phenomenon. However, there are isolated reports which investigate the issue. Lin et al. [216] have fabricated both Ohmic and non-Ohmic Ti contacts on individual ZnO nanowires, and relate the quality of the metal to the electrical behaviour. Pure and high-quality Ti allows for Ohmic conduction, while contacts deposited under poorer conditions show non-Ohmic behaviour, which is attributed to nanocrystalline TiO_x granules in either the bulk of the contact or in the vicinity of the nanowire [216]. In addition to problems with the metal contacts, non-Ohmic behaviour has also been attributed to variation in the surface energy of the ZnO nanowire. Furthermore, because the surface energy of the nanowire is affected by the its radius, tapered or inconsistently-shaped nanowires have shown variable electrical behaviour when contacted with Ti. Bercu et al. [217] showed that Ti contacts to large ends of ZnO nanowires were Ohmic, while contacts to narrow ends were Schottky contacts, which was attributed to differences in the nanowire surface energy due to the difference in radius. It is possible that poor-quality metal contacts or the surface states/energy of the ZnO nanowires contributes to the non-Ohmic conduction reported in this chapter, or possibly a combination of the two factors.

The complexity of non-Ohmic contacts therefore makes comprehensive modelling of the ZnO nanowires difficult. One model which has been used successfully in the past to model non-Ohmic conduction treats both metal/semiconductor interfaces as Schottky diodes [218, 219, 220, 221, 222, 223, 224]. When metal comes into contact with a semiconductor, a Schottky barrier typically forms at the interface [202]. As a ZnO nanowire is contacted by metal at both ends, it forms a metal-semiconductor-metal (MSM) system with two Schottky barriers. For *n*-type semiconductors, the Schottky barrier height (SBH) is given as the difference between the conduction band edge of the semiconductor and the Fermi level of the metal, depicted in Figure 8.2. The height of this Schottky barrier can be approximated to a first order by considering the difference in the work function of the metal and the electron affinity of the semiconductor. However, the actuality of the situation is more complicated, and Schottky barrier heights can vary dramatically even when ostensibly using identical metals and semiconductors due to interface states between the materials [202]. Another figure of merit is the ideality factor, which qualitatively describes how rigorously the Schottky barrier adheres to the Shockley diode equation [202].



Figure 8.2: A figure showing the Schottky barrier ϕ_B which forms at the metal/semiconductor interface due to the mismatch of the Fermi level of the metal and the conduction band edge of the *n*-type semiconductor. Figure adapted from [225].

While there are several ways of modelling back-to-back Schottky diodes, several of these rely on temperature-dependent measurements. In this chapter, we make use of the current density model first developed by Chiquito et al. [218], and later used/adapted by others [219, 220, 221, 222, 223, 224]. Diode measurements of nanowire devices are taken with no gate voltage applied, and the cross-sectional area of the nanowire is then ascertained using SEM. When a source-drain voltage is applied, one of the Schottky diodes is forward-biased, while the other is reverse-biased. The current densities of the individual Schottky diodes are considered using a thermionic emission model [218], and the net current density through the nanowire is given as the sum of the current densities through each individual Schottky diode. For modelling purposes, the series resistance of the nanowire is assumed to be negligible, and barrier lowering from mirror images is neglected. Using these assumptions, the current density of the nanowire can then be expressed as [218]:

$$J = \frac{J_1 J_2 \sinh(\frac{qV}{2k_B T})}{(J_1 \exp(\frac{qV}{2n_1 k_B T}) + J_2 \exp(\frac{-qV}{2n_2 k_B T}))}$$
(8.1)

where k_B is Boltzmann's constant, q is the elementary charge, V is the drain-source potential, T is the temperature, and n_1 and n_2 are the ideality factors of the first and second Schottky contacts, respectively. J_1 and J_2 are defined as:

$$J_{1,2} = A^{**}T^2 \exp(-\frac{\phi_{1,2}}{k_B T})$$
(8.2)

Where A^{**} is the Richardson constant. In the calculations used throughout this chapter, the temperature is taken to be 300 K, and the Richardson constant is taken to be 32 A cm⁻² K⁻² [93].

Current densities were fitted using a custom non-linear curve fit build using OriginLab's Origin 8.5 Fitting Function Builder. The ideality factors and Schottky barrier heights were constrained to be above 0, and fits were achieved using a least squares approach. However, the ideality factor isn't restricted to any particular range. As the ideality factor describes how carriers recombine in thermionic diodes [202], a least-squares approach can in theory produce results which mathematically model the observed results, but do not accurately reflect the Schottky diode. It is important to keep this in mind when considering these results.

8.2.2 Batchwise fabrication of devices

The EBL work was carried out in collaboration with Assoc. Prof. Martin Allen through the MacDiarmid Institute at the University of Canterbury (UC), and Assoc. Prof. Adam Micolich at the University of New South Wales (UNSW). Single nanowire devices were fabricated in batches. The recipes used to grow the ZnO nanowires differed between these batches of devices, as well as EBL techniques, device substrate designs, and the corresponding results. As such, this chapter is separated into sections where each batch is analysed individually. The overall single nanowire yields are considered and the electrical measurements of the devices are presented.

Single nanowires are contacted via EBL on device substrates, which are referred to in this chapter as "samples". Each sample has multiple single-nanowire devices, which are referred to as writefields. The samples are grouped together batch-wise. There are three batches of samples in total which are presented in this chapter. They are:

- 1. UNSW1 Batch of samples fabricated at the University of New South Wales
- 2. UC1 First batch of samples fabricated at the University of Canterbury
- 3. UC2 Second batch of samples fabricated at the University of Canterbury

Samples within each batch are given a number, SN, and individual writefields on each sample are indexed as WFN. The batch number, sample number, and writefield are used to refer to specific single nanowire devices. For example, UC1 - S3 - WF4 refers to the fourth writefield of the third sample from the UC1 batch. The total number of writefields per sample differs between batches, as different device substrate designs were used for each batch. These are detailed in each relevant section, along with the particular ZnO nanowires used.

8.3 UNSW1

8.3.1 Single nanowire device fabrication and yields

UNSW1 consisted of 3 samples in total, referred to as UNSW1 - S1, S2, and S3. All three samples were fabricated using identical nanowires. The contacted nanowires were first hydrothermally synthesised using 25 mM HMT/Zn(NO₃)₂ and with 1.5 mM of PEI($M_W = 2000 \text{ g/mol}$) on a dedicated growth substrate, as per Chapter 5. The hydrothermal growth solution was preheated at 95°C for 20 minutes, and the growth lasted 19 hours. The vertical nanowires were then transferred to the samples using

the dry tex-wipe method described in Appendix A. All of the contacted nanowires originated from a single nanowire growth substrate, to ensure they were directly comparable. The nanowires were used as-is after transfer, with no annealing step. The UNSW1 samples are summarised in Table 8.1.

Table 8.1: The nanowires and anneals used in the fabrication of the UNSW1 batch of samples

| | PEI Concentration | Crowth substrate | Anneal |
|------------|---------------------|------------------|--------|
| | $(M_W = 2000g/mol)$ | Glowin Substrate | |
| UNSW1 - S1 | 1.5 mM | Vertical | None |
| UNSW1 - S2 | 1.5 mM | Vertical | None |
| UNSW1 - S3 | 1.5 mM | Vertical | None |

The device substrates used in all the UNSW1 samples are very similar to those described in Appendix A, although differ slightly. The device substrates have 24 pairs of electrodes, corresponding to 24 write fields in total. The photolithographically defined Ti/Au electrodes are deposited onto an $HfO_2(10 \text{ nm})/SiO_2(100 \text{ nm})/Si$ substrate, which is used as a bi-layer dielectric and back gate. The thin layer of HfO_2 has a higher dielectric constant than SiO_2 , which can improve device characteristics. The EBL system used in the fabrication of UNSW1 is a Raith150-TWO, which slightly differs from the Raith150 at the University of Canterbury, described in Appendix A, however the processes are similar.

Patterned areas of the UNSW1 samples were exposed to 470 μ C/cm², and were developed in 3:1 IPA:MIBK (2-propanol/methyl isobutyl ketone) for 60 seconds. The samples were cleaved in half before metallisation, with one half of each sample exposed to an O₂ plasma at 50W. This plasma ostensibly descummed the nanowire ends from residual PMMA, which can improve contact and the yield of fabricated devices. The other half of each sample was left untreated. Both halves of each sample were then pumped down in an evaporator and coated with a 85/15 Ti/Al layer via resistive evaporation. After metallisation, both halves of each were lifted off using an overnight acetone bath and agitation with a plastic pipette.

Figure 8.3 shows a collage of SEM images of single nanowires contacted from UNSW1 - S1, S2, and S3. The nanowires which have been broken off from the growth substrate during transfer to the device substrate, which has reduced their lengths considerably. Figure 8.3(a) to (e) show nanowires which have successfully been contacted using EBL, while Figure 8.3(f) shows a contact which is unreliable/broken due to discontinuous metal in the contact. This typically occurs from a combination of



factors, including the EBL process itself and the metallisation/liftoff.

Figure 8.3: SEM images taken of different contacted nanowires from the UNSW1 S1 (a & b), S2 (c & d), and S3 (e & f). Each image is labelled with the specific writefield imaged. The quality of the contacts is good overall, although some suffer from discontinuities in the deposited metal, as in (f).
8.3.2 Electrical measurements

Contacted nanowires were electrically measured to determine whether the nanowires conducted and whether they were field dependent. Significantly, the half samples which were exposed to an O_2 plasma prior to metallisation showed no conduction, and are omitted here. This is consistent with reports in the literature correlating increased nanowire resistivity with exposure to O_2 plasma, which is caused by surface states depleting charge carriers from the nanowire [177].

All of the nanowires contacted on the halves of S1, S2, and S3 which weren't plasma treated showed conduction. However, only one device showed field dependence, which was contacted on UC1 - S1. To contrast the field-dependent device from the field-independent devices, Figure 8.4 shows a collage of diode measurements and transfer measurements from various writefields of UNSW1 - S1 which weren't exposed to O_2 plasma.

The SBH/ideality factor of the contacts for each writefield are included on the diode measurements. The Schottky barriers can vary within individual writefields, with each individual contact on a single nanowire having a different barrier height and ideality factor. This can lead to asymmetric diode behaviour such as writefields 16 and 17, shown in Figure 8.4(a) and (e) respectively. Other devices show less variation between the two contacts and symmetry about $V_{DS} = 0$ V, as in writefield 19, shown in Figure 8.4(c). In addition to variations between the two contacts on an individual nanowire, the Schottky barrier height and ideality factors of the contacts can also vary strongly between different nanowires/writefields.

Figure 8.5 shows the current density of writefields 16 and 17 fitted using the back-to-back Schottky barrier model. This model accurately reproduces the electrical behaviour of the devices, and both fits have low residual errors.



Figure 8.4: Diode and transfer measurements of three different writefields from UNSW1 S1. Diode measurements of all three writefields (a, c, and e) show that the nanowires clearly conduct. Writefields 16 and 17 (a and e, respectively) are asymmetric, which is reflected in different SBH/ideality factors for each contact. Conversely, writefield 19 (c) is symmetric, and the SBH of the two contacts differ by $\approx 2\%$. Transfer characteristics of the writefields are shown in (b, d, and f). Writefields 16 and 19 show no field dependence, while writefield 17 clearly responds to applied gate voltages.



Figure 8.5: The current densities of (a) WF16 and (b) WF17 fitted using the back-to-back Schottky model (red).

Although the nanowires show conduction, only the nanowire contacted at writefield 17 shows field dependence. The transfer characteristics of writefields 16 and 19, shown in Figure 8.4(b) and (d), are more representative of the nanowires contacted on the UNSW1 samples. The drain-source currents shows no variation between $V_G =$ -10 to 10 V. In contrast to the field-independent nanowires, WF17 of S1 shows marked field dependence. The transfer characteristics of this device are shown in Figure 8.4(f). The nanowire operates with an on-off ratio of 10⁴ and a threshold voltage of -7 V. The FET has a maximum transconductance of 1.65 nS, and a minimum subthreshold swing of 937 mV/decade.

As the channel length of this device is well-defined, we can estimate the field-dependent mobility of the FET. The mobility of the device is typically defined as [103, 104, 105, 54, 106, 107]:

$$\mu = g_m \times \frac{L^2}{V_{\rm DS} \times C_{\rm NW}} \tag{8.3}$$

where μ is the mobility of the device, g_m is the transconductance of the device, L is the length of the nanowire, V_D is the drain-source voltage, and C_{NW} is the capacitance of the nanowire. The capacitance of a nanowire is typically approximated as the cylinder-on-infinite-plane model, given as [103, 104, 105, 54, 106, 107]:

$$C_{NW} \approx \frac{2\pi\varepsilon_0\varepsilon_r L}{\cosh^{-1}(\frac{r+d_{ox}}{r})}$$
(8.4)

where ε_0 is the permittivity of free space, ε_r is the dielectric constant, r is the radius of the nanowire, and d_{ox} is the thickness of the dielectric layer. It is worth noting that the thin HfO_2 dielectric layer deposited on top of the SiO₂ layer is fundamentally different from the infinite-plane model, which limits the accuracy of the capacitance calculation [226]. For the purposes of this calculation, the $HfO_2(10nm)/SiO_2(100nm)$ dielectric bilayer is treated as a homogeneous 110 nm thick dielectric with an effective dielectric constant of 4.97. This is arrived at by treating the dielectric as a bilayer with an effective dielectric constant of:

$$\varepsilon_{eff} = \frac{\varepsilon_{\mathrm{SiO}_2}\varepsilon_{\mathrm{HfO}_2}}{\frac{1}{1.1}\varepsilon_{\mathrm{SiO}_2} + \frac{0.1}{1.1}\varepsilon_{\mathrm{HfO}_2}}$$
(8.5)

The individual relative dielectric constants are taken to be $\varepsilon_{SiO_2} = 3.9$ and $\varepsilon_{HfO_2} = 25$, respectively [227].

Taking the nanowire radius to be 38 nm, the nanowire length to be 1.1 μ m, ε_0 to be $\approx 8.854 \times 10^{-12}$ F/m, the nanowire capacitance is calculated to be $C_{NW} = 3.76 \times 10^{-16}$ F. Using the transconductance of the device (1.65 nS), the capacitance of the nanowire, and $V_{DS} = 0.5$ V, the mobility of the FET is calculated to be 0.11 cm² V⁻¹ s⁻¹. This mobility is relatively low compared to other ZnO nanowire FETs, which are typically on the order of 0.1 - 1000 cm² V⁻¹ s⁻¹ (a table reviewing the device characteristics of published hydrothermal ZnO nanowire FETs can be found in 3.1 in Chapter 3.)

It isn't necessarily surprising that the majority of measured devices from UNSW1 do not respond to applied fields, as the contacted nanowires are unannealed and are too heavily doped to show field dependence prior to high-temperature annealing [113, 114, 118, 54, 11] (this phenomenon in general is reviewed in detail in the overview of ZnO nanowire FETs in Chapter 3.) However, it is unusual that WF17 of S1 specifically responds to applied gate voltages, while other contacted nanowires on the same sample show no response. It is not clear what exactly causes this particular nanowire to exhibit field dependence and clear switching. All of the the nanowires on the device substrate have been transferred from the same growth substrate, and are all metallised during the same Ti/Al deposition, which should minimise the difference between nanowires and contacts.

However, as we have seen in Figure 8.4, ostensibly similar nanowires contacted on the same chip can have different Schottky barrier heights and ideality factors. Furthermore, variations in the field dependence may be due to differences in the nanowires themselves, or in the contacts, or in a complex interplay between the two. Possible relationships between the nanowire properties, the contact properties, and the Schottky barrier heights/ideality factors are explored in detail in Section 8.6.

8.4 UC1

8.4.1 Single nanowire device fabrication and yields

UC1 consists of three samples in total, referred to as UC1 - S1, S2, and S3. UC1 - S1 and S2 were fabricated using nanowires grown with 25 mM HMT/Zn(NO₃)₂ and 8 mM of PEI($M_W = 800 \text{ g/mol}$). These nanowires originated from a single dedicated growth substrate. The nanowire growth solution was preheated for 1 hour before the growth substrate was introduced, and the hydrothermal growth lasted 19 hours. After the hydrothermal growth was complete, the dedicated growth substrate was annealed at 650°C for 1.5 hours in a tube furnace. The annealed nanowires were then transferred to the UC1 - S1 and S2 samples using the tex-wipe method described in Appendix A.

UC1 - S3 was fabricated using secondary nanowires which were isolated from hierarchical nanowires grown via a PEI-mediated one-pot hydrothermal growth, described in Chapter 6. These hierarchical nanowires were grown on a lateral substrate using 8 mM of PEI($M_W = 800 \text{ g/mol}$) and 25 mM of HMT/Zn(NO₃)₂. These secondary nanowires were specifically chosen as they had shown good field dependence without requiring high-temperature annealing, as evidenced by the devices measured in Chapter 6. The secondary nanowires were isolated from the lateral growth substrates using the same tex-wipe method as used for the vertical growth substrates, described in Appendix A. The secondary nanowires were used as-is, without any annealing.

All three of the UC1 samples used identical device substrates with 6 pairs of electrodes, corresponding to 6 writefields per sample. The device substrates used 100 nm of SiO₂ as the backgate dielectric. Electron beam lithography was carried out using the Raith150 at the University of Canterbury, which is described in detail in Appendix A. The samples were exposed using a dosage of 100 μ C/cm², and developed for 60 seconds in 3:1 IPA:MIBK. After development, the samples were rinsed in IPA, and were not plasma treated. All three of the samples were metallised using Ti/Au (20/90 nm) deposited via egun deposition and resistive evaporation, respectively. After metallisation, S1 and S2 were lifted off in acetone for 3 hours and 1 hour, respectively. S3 was lifted off in NMP (*N*-Methyl-2-pyrrolidone) at 85°C for 1 hour. The UC1 samples are summarised in Table 8.2

Table 8.2: Samples fabricated from the UC1 batch of samples. UC1 S1 and S2 are fabricated using vertically grown ZnO nanowires, while S3 is fabricated using secondary nanowires which have been isolated and transferred from their lateral hierarchical substrate.

| | PEI Concentration | Growth substrate | Anneal | |
|----------|--------------------|------------------------|---------------------------|--|
| | $(M_W = 800g/mol)$ | Growin Substrate | | |
| UC1 - S1 | 8 mM | Vertical | 600°C on growth substrate | |
| UC1 - S2 | 6 mM | Vertical | 600°C on growth substrate | |
| UC1 - S3 | 8 mM | Lateral (hierarchical) | None | |

Figure 8.6 shows a collage of SEM images of contacted single nanowires from UC1 - S1, S2, and S3. Some of the nanowires which were contacted on S1 have been partially or completely etched away during the liftoff process, as in Figure 8.6(c) and Figure 8.6(d). The nanowires which have been totally etched away were clearly present on the device substrate when the metal contacts were initially deposited, as the impression of the nanowires can be seen on the surface of the metal (as in Figure 8.6(c)). Other nanowires on S1 show partial etching or damage to the nanowire surface, as in Figure 8.6(a). In addition to the etching that can occur during the liftoff, all of the nanowires contacted on UC1 - S1 and S2 also show a significant degradation over time, completely disappearing from the substrate after 8 months. The nanowire etching/degradation is discussed in detail in Section 8.7.

In comparison to UC1 - S1, nanowires contacted on UC1 - S2 do not show any surface etching, as shown in Figure 8.6(e) and (f). This is due to a reduction in the liftoff time, from 3 hours in acetone to 1 hour in acetone. Although acetone has been reported to partially etch the surface of ZnO nanowires [228, 229], it is surprising to see complete etching/dissolution of the nanowires over a comparatively short timescale. Several more samples were fabricated and lifted off in acetone over long periods of time (3 to 14 hours) to confirm the phenomenon, which also showed complete etching of the nanowires (not shown here.)

Finally, secondary nanowires contacted on UC1 - S3 showed no obvious surface etching. This may be due to either a short liftoff time (1 hour) or the use of hot NMP as a liftoff solvent, or a combination of these two factors. The overall device yield is also improved by the comparative length of the secondary nanowires, as the tolerances of errors in the contact design/alignment are significantly higher than samples fabricated using shorter nanowires. As such, all of the writefields successfully contacted the nanowires, with a contact yield of 100%.

While all of the nanowires on S3 were successfully contacted, unfortunately the total yield of viable FETs on UC1 - S3 is greatly compromised by leaks through the SiO_2 dielectric, and only one FET can be successfully measured. These leaks are most likely to be small scratches or hairline fissures which have been introduced when the device substrate is cleaved from a wafer. These leaks cannot be seen by SEM, and are only apparent when the nanowires are electrically measured.



Figure 8.6: SEM images of UC1 - S1, S2, and S3. UC1 - S1 suffers from partial etching of the ZnO nanowires, shown in (a) and (b), and complete etching, as in (c) and (d). The nanowires contacted on UC1 - S2, shown in (e) and (f), are unetched due to the reduced liftoff time. Finally, (g) and (h) show contacted secondary nanowires from UC1 - S3.

8.4.2 Electrical measurements

Electrical measurements of UC1 - S1 and UC1 - S2 are shown in Figure 8.7. These measurements are taken in dark (ambient) conditions, shown in black (squares), and when exposed to UV light at 405 nm, shown in red (triangles). Exposing the nanowires to UV light photoexcites charge carriers through electron-hole pairs [96, 230, 185], which affects the electrical characteristics of the device and the SBH/ideality factor of the contacts. Diode measurements of UC1 - S1 and UC1 - S2 are shown in Figure 8.7(a - e). The devices show back-to-back Schottky barrier behaviour under both conditions. Different SBH/ideality factors between two contacts on a single nanowire lead to asymmetric measurements about $V_{DS} = 0$ V, as in Figure 8.7(b). These asymmetric devices represent \approx 85% of the contacted nanowires from UC1 - S1 and S2.

The remaining contacted nanowires show more symmetric behaviour, such as S2- WF4 (shown Figure 8.7(e)). Although the contacted nanowires on UC1 - S1 and S2 show conduction, none of the writefields show field dependence. An example is given in Figure 8.7(f), representative of all the contacted nanowires. It is clear that the device shows no response to the applied field due to the constant I_{DS} value ($\approx 1.5 \times 10^{-11}$ A).

All of the devices show an increase in current when measured under UV illumination. The increase in current can be symmetrical, with increased drain-source current in both positive and negative V_{DS} values (as in Figure 8.7(e)), or asymmetrical, with increased current in only one direction (as in Figure 8.7(b)). These differences can be modelled as changes to the SBH/ideality factor to one or both of the contacts under UV illumination, which are shown in Figure 8.7(a - e).



Figure 8.7: Electrical measurements of UC1 S1 (a,b) and S2 (c - d) under ambient (black, squares) and UV-illuminated (red, triangles) conditions. The fitted SBH/ideality factor of each contact is shown in the diode measurements of each device. All of the nanowires show conduction, although none show field dependence - S1 - WF3 is included in (f) as an example.

The current densities of some devices from UC1 - S1 are shown in Figure 8.8. Similar to the fits shown previously in Figure 8.5, the model accurately reproduces the current densities for diodes measured in both the dark and under UV illumination. The electrical behaviour of symmetric diodes is reproduced particularly well, although the model shows some inaccuracy with increasing asymmetry, as in Figure 8.8(d). However, the overall residual error remains small, with an R² value of 0.94.

There is one report on the lowering of ZnO nanowire Schottky barriers with UV light [231], which is mediated by the desorption of oxygen species on the nanowire surface through photoexcited holes. This reduces the surface depletion of the nanowire, which in turn lowers the SBH. This is identical to the mechanism which induces persistent photoconductivity in ZnO nanowires, described in detail in Chapter 6.

In this report, the UV light is at 365 nm, which falls within the bandgap of ZnO (\approx 368 nm, or 3.37 eV [10]). In the experiments presented here, the wavelength of light used is 405 nm, which is less energetic than the ZnO band gap. Despite this discrepancy, it is likely that the same oxygen desorption mechanism is taking place in the UV-illuminated measurements of UC1 - S1 and S2. As we have shown in Chapter 6, charge carriers and persistent photoconductivity can be induced in ZnO nanowires by using 405-nm UV light, with energy less than the bandgap of ZnO. There are also reports in the literature which show charge carrier photoexcitation when using UV light which falls below the bandgap [186, 187].



Figure 8.8: The current densities of UC1 writefields measured in dark and UV conditions, fitted using the back-to-back Schottky model (red).

In contrast to the vertical nanowires used in UC1 - S1 and S2, UC1 - S3 was fabricated using secondary nanowires which were known to be field dependent, as they were isolated from a lateral field-dependent FET fabricated in Chapter 6. However, 5 out of the 6 writefields were unusable due to leakage currents on the order of 10^{-6} A between V_G = -10 V and 10 V. These high leakage currents are caused by damage to the SiO₂ dielectric layer on the sample, introducing unintentional conduction pathways between the source/drain electrodes and the gate electrode. This damage was likely caused when the device substrate was cleaved from its larger wafer, which was inadequately protected.

Despite this damage, the nanowire contacted at writefield 3 was measurable as the leakage currents were considerably smaller. An SEM image of this nanowire is shown in Figure 8.6(g). Figure 8.9 shows the electrical measurements of UC1 - S3 - WF3. The device is measured as a diode in Figure 8.9(a) under dark and UV-illuminated conditions. The measurement is asymmetric about $V_{DS} = 0$ V under both dark and UV conditions due to the difference in Schottky barrier heights and ideality factors. Figure 8.9(b) shows the drain-source current (squares) of the transfer characteristics of the device, which is driven at $V_{DS} = 3$ V. Although the nanowire shows field dependence, the off-current of the device is relatively high (10^{-8} A), which is due to the leakage current of the device. A typical leakage current from a back-gated lateral FET fabricated in Chapter 6 is shown in Figure 8.9(c). In comparison, Figure 8.9(d) shows the leakage current of UC1 - S3 - WF3 (triangles.) The leakage current for UC1 - S3 - WF3 is \approx 3 to 4 orders of magnitude higher than the leakage current of the lateral FET due to the damaged SiO₂ dielectric.

The FET at UC1 - S3 - WF3 operates at a threshold voltage of -6.2 V, with an on-off ratio of 10². The maximum transconductance is 30.1 nS, and the minimum subthreshold swing is 845 mV/decade. Taking the length of the nanowire to be 1.56 μ m and the radius of the nanowire to be 17 nm, the capacitance of the nanowire is calculated to be 8.24 × 10⁻¹⁷ F, using the methods described in Section 8.3.2. Using the transconductance of the device along with the capacitance and length of the nanowire yields a mobility of 2.96 cm² V⁻¹ s⁻¹. This mobility is low yet comparable to other ZnO nanowire FETs which use hydrothermal nanowires, which are typically between 1 - 1000 cm² V⁻¹ s⁻¹ (a summary of different device characteristics in the literature can be found in Table 3.1 in Chapter 3.)



Figure 8.9: Electrical measurements of UC1 - S3 - WF3. (a) Diode measurements in dark (black, diamonds) and UV (red, circles) show an asymmetric device. (b) Drainsource current (I_{DS}) (black, squares) from the transfer measurement of the device. (c) The leakage of a typical lateral FET, where the secondary nanowire used in UC1 - S3 - WF3 is sourced from. (d) The leakage current (I_G) (red, triangles) from the transfer characteristics of UC1 - S3 - WF3.

8.5 UC2

8.5.1 Single nanowire device fabrication and yields

UC2 consists of 4 samples in total, referred to as UC2 - S1, S2, S3, and S4. These samples were fabricated using annealed and unannealed nanowires hydrothermally grown with 6 mM and 8 mM of $PEI(M_W = 800 \text{ g/mol})$ and 25 mM HMT/Zn(NO₃)₂. Each nanowire growth solution was preheated for one hour at 95°C before the growth substrates were introduced. Each hydrothermal growth lasted 19 hours.

These growth recipes were chosen as nanowires grown with 8 and 6 mM of $PEI(M_W = 800 \text{ g/mol})$ show a wide distribution of diameters below 100 nm (these results are summarised in Figure 5.14 in Chapter 5.) There was strong motivation to investigate the effects of the nanowire diameter on the nanowire field dependence, as the persistent photoconductivity measurements in Chapter 6 suggested that depleting surface states and small diameters may be responsible for field dependence. It was hoped that we could contact field-dependent nanowires which were grown vertically, summarise these results, and then contact isolated secondary nanowires *en masse*.

The device substrates used in UC2 are those described in Appendix A. Specifically, each device substrate is photolithographically patterned with 15 pairs of electrodes, corresponding to 15 writefields. The substrate itself is SiO₂ (100 nm)/Si, with a Cr/Au backgate. Samples from UC2 use a mix of unannealed and annealed nanowires. The nanowires were annealed by transferring them from the growth substrate to the device substrate as per Appendix A, and then annealing the device substrate at 450°C for 1 hour. These nanowires were consequently annealed on the device substrates rather than the growth substrates, as in the UC1 samples. All of the nanowires grown using 6 mM of PEI(M_W = 800 g/mol) were isolated from a single growth substrate. The specific nanowires used in each sample from UC2 are summarised in Table 8.3.

| Table 8.3: | The nanow | vires used in tl | ne fabrication | of UC2, | and their | correspondin | ig an- |
|-------------|-----------|------------------|----------------|---------|-----------|--------------|--------|
| nealing tre | eatments. | | | | | | |

| | PEI Concentration (M _W = 800g/mol) | Growth substrate | Anneal |
|----------|--|------------------|---------------------------|
| UC2 - S1 | 8 mM | Vertical | None |
| UC2 - S2 | 6 mM | Vertical | 450°C on growth substrate |
| UC2 - S3 | 6 mM | Vertical | None |
| UC2 - S4 | 6 mM | Vertical | 450°C on growth substrate |

Electron beam lithography was performed using the Raith150 at the University of Canterbury, as described in Appendix A. Electron beam dosages were kept at 100 μ C/cm², and substrates were developed for 60 seconds in 3:1 IPA:MIBK. After development, the device substrates were exposed to an argon plasma at 50 W for 30 seconds prior to metallisation. Argon plasmas have been reported to increase the conductivity of ZnO nanowires [232, 233, 113, 118], which can improve the quality of the nanowire/contact interface. Ti/Au (20/80 nm) contacts were deposited using egun and resistive evaporation respectively, after which the substrates were lifted off in acetone over 40 minutes. This short liftoff was intentionally chosen to avoid nanowire etching as we had observed in the UC1 samples.

Figure 8.10 shows a collection of SEM images of S1 and S3. All of the nanowires appear intact and unetched, which is attributed to the short liftoff time in acetone. As was the case with samples from UNSW1 and UC1, the nanowires have snapped and partially broken during transfer from the growth substrate to the device substrate. This has significantly reduced their length, which reduces the tolerance of misalignment during the EBL process. This can lead to some nanowires being missed due to EBL misalignment. Despite this, the samples have good overall yields of contacted nanowires, with yields ranging from 75% to 100%. Furthermore, as each sample has 15 writefields, many separate nanowires can be contacted on each chip.

Although the number of successfully contacted nanowires is relatively high, the nanowires suffer material degradation over time and completely dissolve over the course of at least 8 months, as is the case for the UC1 samples. This gradual degradation occurs despite the fact that the nanowires appear initially unetched due to the short liftoff in acetone. The material degradation is discussed in detail in Section 8.7.



Figure 8.10: SEM measurements of UC2 - S1 and S2, which are representative of the UC2 batch of samples overall. None of the nanowires show any signs of etching after liftoff, as they are kept in acetone for only 40 minutes.

8.5.2 Electrical measurements

Electrical measurements of UC2 - S2 and S3 are shown in Figure 8.11. UC2 - S2 and S3 use nanowires which have been synthesised using the same growth conditions, but are annealed and unannealed, respectively. These measurements are taken immediately after the samples have been lifted off, under ambient conditions. All of the nanowires show back-to-back Schottky contact behaviour. Similar to the UNSW1 and UC1 samples, some of the nanowires are symmetric about $V_{DS} = 0$ V, such as Figure 8.11(a), while others show extreme asymmetry, such as Figure 8.11(c), depending on the SBH/ideality factor of each contact.

The unannealed nanowires contacted on UC - S3 show higher currents than the annealed nanowires contacted on UC - S2. The highest currents of the unannealed nanowires are on the order of 5×10^{-6} A, shown in Figure 8.11(e), even when the drain-source voltage is kept below 1 V. In comparison, the drain-source currents for annealed nanowires are typically on the order of 5×10^{-8} A, as in Figure 8.11(a - c). Figure 8.12 shows the current densities of some UC2 devices which have been fitted using the back-to-back Schottky model described in Section 8.2.1. The model accurately fits strongly asymmetric devices, such as UC2 - S2 - WF11 (shown in Figure 8.12(b)). However, other devices show some deviation from their respective fits, such as S3 - WF12 (Figure 8.12(b)). This highlights some of the limitations of the model, which may be related to the assumptions made, such as a negligible nanowire

resistance.



Figure 8.11: Diode measurements of UC2 - S2 (a - c) and S3, (d - f) taken immediately after liftoff, under ambient conditions. Contacted nanowires on UC2 - S2 and S3 can show symmetrical IV characteristics, (a) and (e), or asymmetric characteristics, (c) and (d). The different SBH/ideality factors of each contact are displayed on the graph. The annealed nanowires shown in (a - c) have lower currents than the unannealed nanowires, shown (d - f).



Figure 8.12: The current densities of some UC2 writefields fitted using the back-to-back Schottky model (red).

Although the nanowires show conduction immediately after liftoff, the devices are unresponsive to gate voltages and show minimum field dependence. Transfer characteristics of an annealed nanowire (UC2 - S2 - WF5) are shown in Figure 8.13. The nanowire is driven at a V_{DS} of 0.1 V. The gate voltage is lowered from $V_G = 0$ to $V_G = -20$ V in an attempt to deplete the nanowire. Although the nanowire has been annealed, the drain-source current is only reduced by an order of magnitude at best, with no well-defined subthreshold regime.



Figure 8.13: Transfer characteristics of UC2 - S2 - WF5. Despite being annealed, the nanowire shows no field dependence.

Unannealed nanowires also show a lack of field dependence. Figure 8.14 shows an attempt to deplete an unannealed nanowire at much higher gate voltages. This particular nanowire (UC2 - S1 - WF11) shows a drain current of $\approx 1 \times 10^{-8}$ A at a drain-source voltage of 0.1 V. The transfer characteristics in Figure 8.14(a) show the gate voltage progressively driven from $V_G = 0$ V to $V_G = \approx$ -67 V. As the voltage is decreased, the leakage current spikes upwards at $V_G = -25$ V, until the dielectric layer breaks down catastrophically at $V_G = -67$ V. At this point the leakage current and drain-source current both drastically increase. Imaging the nanowire in SEM after the transfer measurement shows the extent of the damage, as shown in Figure 8.14(b). The nanowire itself has completely disappeared and the SiO₂ layer between the source and drain electrodes has been cratered. This clearly demonstrates the difficulty in using unannealed nanowires as FETs. The nanowire is shown prior to transfer measurements in Figure 8.10(b).



Figure 8.14: (a) Transfer characteristics of UC2 - S1 - WF11. The device is driven at a constant V_{DS} of 0.1 V, and its drain-source current is approximately 1×10^{-8} A. The gate voltage is driven to successively lower voltages in an attempt to deplete the nanowire. Eventually the SiO₂ dielectric catastrophically breaks down at \approx -67 V. (b) An SEM image of the device after the SiO₂ has completely broken down. The nanowire has been destroyed, and the SiO₂ has been damaged. An SEM image of the device prior to breakdown can be seen in Figure 8.10(b).

8.5.3 Conductivity decay and electron beam exposure

All of the contacted nanowires from UC2 - S1, S2, S3, and S4 initially show conduction immediately after lift-off. However, this conductivity rapidly decays over a 48 hour period, with Figure 8.15(a) and (b) showing diode measurements UC2 - S2 immediately after liftoff and 48 hours later. Although these nanowires all initially showed clear conduction, they behave as insulators only 2 days later. This insulating behaviour may be due to significant depletion within the nanowire channel, or extremely high Schottky barrier heights. Although this conducting-to-insulating transition is unusual, and does not have any particular precedent in the literature, it is reproducible and measurable for all of the nanowires measured in all of the samples from UC2.

After a nanowire has undergone a conductor-to-insulator transition, the conductivity of the device can be restored by exposing the nanowire to an electron beam source, such as an SEM. Figure 8.15(c) and (d) shows the same nanowires measured in (a) and (b) (respectively) after the current had decayed, and after the nanowires had been imaged in SEM. This clearly induces a drastic change in the nanowire, which has reverted from insulating to conducting. This ebeam-induced change in conductivity is temporary, and the nanowire once again transitions from a conductor to an insulator typically over 48 - 72 hours.

To ensure that the conductivity change was caused by exposure to the electron beam as opposed to some other coincidental mechanism, such as the low-pressure environment within the SEM chamber, WF8 to WF15 were selectively imaged on UC2 - S2, while the other writefields were left unimaged. As the entire sample is pumped down and subjected to the same conditions, the only difference between the imaged and unimaged nanowires is whether they are exposed to the electron beam of the SEM or not. The nanowires which were left unimaged showed no discernible sign of conduction, with the drain-source current effectively as low as the noise limit for the parameter analyser. Conversely, the nanowires which were imaged by the SEM show clear conduction, with the drain-source current several of orders of magnitude higher than the not imaged nanowires. This conclusively shows that the electron beam is responsible for inducing the temporary transition from insulating to conducting behaviour in the ZnO nanowires.

There are a very limited number of published reports in the literature which relate ZnO nanowire conductivities with exposure to electron beams [234, 235], although the increase in conductivity is relatively mild (1 to 2 orders of magnitude) when compared to the drastic transition from insulating to conducting states observed here. In both instances, the nanowires in question are contacted by back-to-back Schottky barriers.



Figure 8.15: Electrical measurements of UC2 - S2 demonstrate the conducting-toinsulating transition the nanowires experience over time after liftoff, and the reversibility of this transition with electron beam exposure. S2 - WF10 and WF9 are shown in (a) and (b), respectively. Both show clear conduction when measured immediately after liftoff (black), but are effectively insulating after 48 hours (red). The short range of voltages measured in (b) is due to the device reaching a compliance of 5×10^{-8} A. WF10 and WF9 are also shown in (c) and (d). The red line shows their IV characteristics several days after liftoff, and they are clearly insulating. After being exposed to the electron beam (shown in black), their conductivity is restored.

Although no conclusive mechanism behind the conductivity increase is given, the proposed theories involve the desorption of oxygen surface states from the nanowires, similar to the influence of UV illumination on the UC1 samples reported in Section 8.4.2 and the hierarchical nanowire FETs in Chapter 6. This would also explain the initial transition from conductive to insulating behaviour we see after the nanowires are lifted off, as the nanowire surface of nanowire may be passivated by the PMMA layer used in EBL until it is removed by the acetone. In theory, this would allow us to re-apply a PMMA layer after the device has been exposed to the electron beam in SEM.

To test this theory, ZnO nanowires from UC2 - S4 were imaged in the SEM several days after being initially lifted off, to allow the nanowires to naturally transition from conducting to insulating behaviour. Once the nanowires were behaving as insulators, the sample was placed into the SEM and the contacted nanowires exposed to the electron beam. Figure 8.16(a) shows electrical measurements of UC2 - S4 - WF5 immediately after liftoff (black), and after 48 hours (red). The conduction of the nanowire has clearly decayed significantly, from an initial maximum current of 10^{-8} A to $\approx 10^{-13}$ A. After being imaged in SEM (blue), the nanowire shows a drastic increase in conductivity, as shown in Figure 8.16(b), with a maximum current of $\approx 10^{-6}$ A. This post-SEM current is actually higher than when the nanowire is immediately measured after liftoff - the two measurements are directly compared in Figure 8.16(c). Although the post-SEM measurement shows lower currents within a drain-source voltage of $\approx +/-2.5$ V, the maximum currents of post-SEM are 2 to 3 orders of magnitude higher than the post-liftoff measurements.

Once the nanowire was measured post-SEM to ensure that it was conductive, a thin layer of PMMA (on the order of hundreds of nm) was deposited on the surface using a spin coater. Figure 8.16(d) shows measurements of UC2 - S4 - WF5 immediately after the PMMA has been deposited and soft-baked onto the device substrate. Although the maximum current of the nanowire remains unchanged in the negative drain-source voltage direction, the drain-source current at a drain-source voltage of 10 V is approximately 3 orders of magnitude lower than the post-SEM measurements. These currents are still both higher than the immediate post-liftoff measurements of the device. A consequent measurement 192 hours (8 days) after the PMMA was applied (Figure 8.16(d), red) shows that the conductivity has decayed significantly from its initial PMMA-encapsulated state, but the nanowire has not undergone an insulating transition. This differs from unencapsulated nanowires which become insulating within 48 hours of liftoff.



Figure 8.16: (a) Electrical measurements of UC2 - S4 - WF5 after the nanowire was initially lifted off (black) and after 48 hours (red). (b) Electrical measurements of the same device 48 hours after liftoff (red) and immediately after SEM (blue). (c) Measurements after initial liftoff contrasted with measurements immediately after SEM. (d) Electrical measurements immediately after PMMA was deposited on the device (black) and 192 hours after the PMMA was deposited (red).

This significant decay over 192 hours may due to surface species adsorping to the nanowire during the brief period of time between the nanowires being removed from the SEM and the application of PMMA. The fact that the nanowires are conductive immediately after liftoff and turn insulating over 48 hours (with currents on the order of 10^{-13} A within a source-drain voltage of +/- 10 V) suggests an interaction between the nanowires and the environment, which is almost certainly mediated through surface states. This is corroborated by the fact that the conductivity can be temporarily restored to within an order of magnitude or even increased from its original value, by subjecting the body of the nanowire to an electron beam. Furthermore, the conductivity ity slowly decays again once the nanowire is reintroduced into the atmosphere after SEM measurements have taken place.

8.6 Schottky barrier heights and ideality factors

*red*All of the contacted nanowires which are measured in this chapter show non-Ohmic behaviour, and have been fitted using a back-to-back Schottky barrier model. As we have seen, the Schottky barrier heights and ideality factors derived from this model can vary between two different samples fabricated using the same nanowires, two different nanowires contacted on the same sample, or even two different contacts on the same nanowire. Here, the different SBH (Schottky barrier height) values and ideality factors observed across the devices are summarised.

8.6.1 Relation between contacts on a single nanowire

Schottky barrier heights and ideality factors can be asymmetric or symmetric for a single nanowire. Symmetric devices imply that there are similarities between the two metal/semiconductor contacts on the same nanowire. Despite this, there is a wide variety of Schottky barrier heights between symmetric devices, which may suggest that some variable local to each nanowire determines the barrier height, and that this variable differs between nanowires. Figure 8.17(a) shows the SBH of contact 2 plotted against the SBH of contact 1 for Schottky barriers within 0.75 eV, restricting the dataset to the more symmetric devices. The linear intercept which runs through (0, 0) represents the symmetry line - the closer a nanowire is to this line, the more symmetrical the SBH is. Although there are a large number of symmetric devices, there is a wide variety in the barrier heights between different devices. Similarly, Figure 8.17(b) shows the ideality factor of contact 2 plotted against the ideality factor of contact 1, with devices with contact ideality factors within 0.9 - 1.2. The ideality factors of the contacts show slightly more asymmetry than the barrier heights.

8.6.2 Relation between the SBH, ideality factor, and nanowire diameter

The Schottky barrier heights and ideality factors of both contacts on symmetric devices are highly correlated with one another. Because each pair of contacts has a nanowire in common, it may be expected that there is a relationship between the physical properties of the nanowires and the SBH/ideality factors of the contacts. Furthermore, as there is a well-established relationship between the diameter of a nanowire and its conductivity [179], it would not be surprising to see a relationship between the nanowire diameter and the SBH. Figure 8.18 shows the SBH (a) and ideality factors (b) of both individual contacts plotted against the diameter of the nanowire. However, no clear relationship emerges when the barrier heights and ideality factors are plotted against the nanowire diameter. This suggests that the the SBH and ideality factor may be more



Figure 8.17: (a) The Schottky barriers of pairs of electrodes, restricted to Schottky contacts with barrier heights within 0.75 eV. The linear intercept which runs through (0, 0)represents a line of symmetry. (b)The ideality factors of pairs of electrodes, restricted to contacts with ideality factors within 0.9 - 1.2.



Figure 8.18: The Schottky barrier height (a) and ideality factor (b) of contact 1 and plotted against the radius of the contacted nanowire. No clear relationship emerges between the two.

or less fixed by the surface of the nanowire, or by some other variable which does not depend on the bulk or surface-to-volume ratio of the nanowire.

8.6.3 Nanowire growth conditions and Schottky contacts

The majority of the nanowires contacted here are grown using 6 mM of 8 mM of PEI($M_W = 800$), referred to here as PEI(800), along with the standard precursors of 25 mM HMT/Zn(NO₃)₂. The growth conditions of the ZnO nanowires have an obvious impact on their diameter, as detailed in Chapter 5. These growth conditions also

may affect the electrical properties of the nanowires, which in turn could influence the Schottky contacts. Furthermore, some of the contacted nanowires have been annealed, while others have been left unannealed. Annealing certainly affects the charge carrier concentration and field dependence of ZnO nanowires, which may in turn systematically influence the Schottky contacts.

Figure 8.19 shows the SBH (a) and ideality factors (b) of both contacts on each nanowire, grouped by their hydrothermal precursors and their annealing treatments. The data has been restricted from 0 - 0.75 eV and 0.9 - 1.2, respectively. Although there is no decisive trend, there may be a tendency for annealed nanowires grown with 6 mM of PEI(800) to show lower SBHs. This trend remains even when accounting for the diameter of the nanowire, which is shown in Figure 8.19(c). The contact ideality factors of annealed nanowires grown with 6 mM of PEI(800) also show a larger degree of symmetry than the other nanowires used here, with only 3 devices deviating significantly from the symmetry line in Figure 8.19(b). These results are also consistent when accounting for nanowire diameter, as shown in Figure 8.19(d), with the bulk of the ideality factors of contact 1 having a value close to unity. These results may suggest that nanowires grown using 6 mM of PEI(800) are more suitable for nanowire devices in general, given the reproducibility of the Schottky contacts.



Figure 8.19: (a, b) The Schottky barrier heights and ideality factors of pairs of electrodes, grouped by the nanowire growth conditions. (c, d) The Schottky barrier heights and ideality factors of contact 1 plotted against nanowire diameter, grouped by the nanowire growth conditions.

8.7 Material degradation

Partial etching and complete dissolution of ZnO nanowires occurs when the liftoff time in acetone is 3 or more hours, as reported in Section 8.4. When the samples are imaged in SEM after liftoff, the majority of the nanowires have disappeared, and all that remains is an amorphous residue on the surface of the substrate. Figure 8.20 shows a UC1 sample which has degraded during a 14 hour liftoff in acetone. This particular sample is omitted from Section 8.4, as all of the nanowires have completely degraded and no electrical measurements can be performed.

Figure 8.20(a) shows the amorphous residue left by the etching process on the substrate. Some heavily etched nanowires remain on the substrate. The white portion of right-hand side of the image is a metal electrode on the device substrate, which appears overexposed in order to highlight the residue on the substrate surface. The residue isn't found elsewhere on the sample, and only occurs where the nanowires were prior to liftoff. Figure 8.20(b) shows a composite image of the residue, with some remaining nanowires which haven't been completely etched away. The image consists of a typical secondary electron image (bottom-left), and a backscatter image (upper-right.) Backscatter images give information about the elemental make-up of the material being imaged, as there is a relationship between the contrast of the material and its atomic number. The nanowires are clearly higher in contrast than the residue, which suggests they are composed of different elements.



Figure 8.20: Images of nanowires on the substrate of a sample from UC1. The sample was lifted off in acetone over 14 hours, which dissolved the nanowires which had been transferred to the device substrate. The remaining nanowires are covered in a residue on which sits on the substrate surface, which is highlighted in (a). (b) shows a composite SEM image of the nanowires/residue, which is split between SEI and backscatter modes.

After this relationship between the liftoff time and damage to the nanowires was discovered, the liftoff time was reduced to 40 minutes in an effort to prevent damage to the nanowires. This was successful, as the nanowires lifted off in a shorter time showed no obvious damage. However, even the undamaged ZnO nanowires from UC1 and UC2 show complete degradation/dissolution over time when the samples are kept in regular plastic sample containers maintained in atmosphere.

Figure 8.21 shows SEM images of nanowires contacted on UC2 - S2, which were obviously intact immediately after liftoff and at the time of imaging. When they were reimaged after 8 months, it became clear that they had disappeared. Similar to Figure 8.20, all that remains is an amorphous residue on the SiO_2 surface. This behaviour is reproducible and consistent across all of the samples contacted in UC1 and UC2, although the exact timeframe over which the disappearance occurs isn't precisely known at the time of writing.



Figure 8.21: SEM images of UC2 - S2 - WF8 and WF15 initially after they were lifted off, and once they were reimaged 8 months later. The nanowires have clearly disappeared some time before the second images were taken. The post-liftoff images of WF8 and 15 are shown in (a) and (c) respectively, while the images taken after an 8 month interlude are shown in (b) and (d).

We don't currently know the mechanism behind the degradation of the nanowires over time. ZnO is an amphoteric material [10], which means that it can dissolve in both basic and acidic conditions. As the nanowires are initially intact when first contacted, the dissolution over time suggests than an interaction between the nanowires and the atmosphere is what drives the material degradation. Furthermore, we have seen ZnO nanowires remain intact when kept under the initial PMMA layer used for electron beam lithography for up to one year, and only degrade over time once the PMMA layer has been lifted off. This reinforces the idea that an atmospheric interaction causes the dissolution of the ZnO nanowires. ZnO nanowires have been reported to dissolve in water over time [236], which may condense from the atmosphere and degrade the nanowires over time.

Another possible mechanism for the degradation of the nanowires over time is carbon capture from the environment. It is well established that PEI captures CO_2 from the atmosphere [237, 238, 239, 240, 241], which may be present on the surface of the nanowires, and that the captured CO_2 molecules are acidic [242, 243, 244]. PEI has a complex interaction with different precursors during the hydrothermal growth, as detailed in Chapters 2 and 5, but it is also thought to bind to the non-polar facets of ZnO while the nanowires are nucleating and growing.

It is not known how much PEI remains on the surface of the nanowires after the hydrothermal growth. Thorough rinsing [174, 74] and high-temperature annealing [245, 246, 73] are typically attributed to removing any residual PEI after nanowire growth, although this is not often directly investigated. Both annealed and unannealed nanowires contacted in this chapter show material degradation over time; if PEI is removed from nanowire surfaces during the annealing process, and CO_2 capture by the PEI is responsible for the material degradation, we would expect the unannealed nanowires to degrade, while the annealed nanowires remain intact. As this is not the case, it is unlikely that there is enough residual PEI on the surface to contribute to material degradation.

However, surfaces of ZnO interact with carbon molecules in the atmosphere [182, 247], and can capture CO_2 without the presence of PEI [248]. Carbon in the form of CO or CO_2 can also generate formate species (HCHO₂⁻) [249, 250], which are also acidic. Atmospheric interactions with carbon and carbon complexes may still play a role in the degradation of the nanowires, even if the PEI has been completely removed from the nanowire surface.

Nanowires which have degraded eventually over time also leave a residue on

the substrate in their vicinity, even when they are initially intact when lifted off. Figure 8.22 shows an SEM image of partially dissolved ZnO nanowires from UC2 -S2 taken under backscatter conditions. The backscatter measurements are coupled with energy dispersive x-ray spectroscopy (EDS) measurements, which is a technique used to determine the elemental make-up of a material by measuring the energy of x-rays released when the material is bombarded with electrons. Figure 8.22(a) shows the backscatter image of the nanowires, with a clear contrast difference between the nanowires and the residue. Figure 8.22(b) shows an EDS measurement of carbon, which has been overlaid with the backscatter image. The highest concentration of carbon is located in the residue material. Figure 8.22(c) shows an EDS map of Zn (green) and O (blue), also overlaid the backscatter image. Although there are trace amounts of Zn within the residue, the majority of the Zn is confined to the nanowires. Finally, Figure 8.22(d) shows C, Zn, and O overlaid with the backscatter image. This clearly suggests that the residue is rich in C compared to other parts of the image, which supports the idea of carbon accumulation from the atmosphere.



Figure 8.22: (a) A backscatter image of etched ZnO nanowires which are surrounded by amorphous residue. (b) An EDS map of C (red) which is superimposed over the backscatter image shown in (a). (c) An EDS map of Zn (green) and O (blue) which is superimposed over the backscatter image in (a). (d) A composite EDS map showing C, Zn, and O. The residue clearly has a high concentration of C compared to the surrounding substrate/nanowires.

8.8 Summary

In summary, two functioning FETs were fabricated in total. One of these FETs (UNSW1 - S1 - WF17) used an unannealed nanowire grown vertically with 1.5 mM of PEI(M_W = 2000 g/mol.) The device operated with an on-off ratio of 10^4 and a threshold voltage of -7 V, a transconductance of 1.65 nS, and a minimum subthreshold swing of 937 mV/decade. The field-dependent mobility of the FET is approximately 0.11 cm² V⁻¹ s⁻¹. The second FET (UC1 - S3 - WF3) used a lateral secondary nanowire which was isolated from a field-dependent hierarchical FET. The individual secondary nanowire operated with a threshold voltage of -6.2 V, an on-off ratio of 10^2 , a transconductance 30.1 nS, and a subthreshold swing of 845 mV/decade. The mobility was calculated to be 2.96 cm² V⁻¹ s⁻¹.

Other contacted nanowires showed consistent back-to-back Schottky behaviour, but were field independent. The measurements on single ZnO nanowires presented in this chapter raise several outstanding questions about the material properties of ZnO nanowires, their reactions with the environment, and the fabrication of ZnO nanowire devices. These can be summarised as follows:

• What determines the Schottky barrier height and ideality factor of single nanowire contacts?

While there are reports on the various factors that influence the Schottky barrier height [251], the results presented here show a variety of barrier heights despite using ostensibly identical or very similar nanowires and contact metals. While there is a clear relationship between the Schottky barrier heights and ideality factors of two contacts on a single nanowire, there isn't any obvious relationship between the barrier heights/ideality factors of the contacts and the physical properties of the contacted nanowire. Determining the variables which influence the Schottky barrier height/ideality factor of a contact would allow for more reliable and reproducible fabrication of a number of single nanowire devices, which utilise both Schottky and Ohmic contacts.

• What determines the field dependence of ZnO nanowires?

We have seen one example of a vertically-grown ZnO nanowire from the UNSW batch which shows reliable field dependence (specifically UNSW1 - S1 - WF17 in Section 8.3.2). Other nanowires contacted on the same chip do not show field dependence, even though they originate from the same growth substrate and are contacted the same way. Why is this nanowire in particular field dependent, while the other nanowires are not? Similarly, a field dependent secondary nanowire is contacted in Section 8.4, and shows response to gate voltages de-

spite the relatively high leakage currents. This is unsurprising, as the secondary nanowires are field-dependent when used in lateral FETs, but what are the critical differences between the secondary nanowires and the vertical nanowires which determine field dependence?

• What causes the conducting-to-insulating transition of the UC2 nanowires contacted in Section 8.5?

All of the nanowires contacted from UC2 in Section 8.5 show a dramatic transition after liftoff between conductive and insulating states over 48 hours. This transition can be temporarily reversed by exposing the samples to an electron beam, such as an SEM. What causes this initial transition, and what mechanism allows it to be reversed? If the transition occurs due to interactions with the nanowire and the atmosphere, why doesn't encapsulation in PMMA retain the conductive state after the device is exposed to the electron beam? And why do other single nanowire devices not show the same transitions over time?

• What causes the ZnO nanowires to dissolve over time after liftoff?

The nanowires contacted from the UC1 and UC2 batches of nanowires (in Sections 8.4 and 8.5 respectively) show dramatic degradation over time when stored in plastic containers in ambient air, eventually dissolving entirely over the course of at least 8 months. Interactions between atmospheric carbon and the ZnO nanowire surface is one possible mechanism for this dissolution, as are interactions with atmospheric water. However, if either of these mechanisms destroy the ZnO nanowires over time, why do other nanowires grown under identical conditions - including the untransferred nanowires which remain on the dedicated growth substrates - remain intact even after several years?

The unanswered questions presented in this chapter motivate a substantial amount of future work, which is presented in Chapter 9.
Chapter 9

Conclusion and future work

9.1 Conclusion

This thesis has explored the hydrothermal synthesis of ZnO nanowires and their application in field-effect transistors. Field-effect transistors have been successfully fabricated by bridging electrodes with hierarchical ZnO nanowires, grown via both PEImediated one-pot and modular two-pot hydrothermal growths. Significantly, these devices show field dependence without requiring high-temperature annealing. One-pot devices can be used dry, with on-off ratios of 10³ - 10⁵ and threshold voltages between -7.5 V to 5 V. Two-pot devices are measured in a wet environment using back and top gates, with on-off ratios and threshold voltages of 10⁵ and 8 V, and 10³ - 10⁴ and 0.4 - 0.9 V, respectively. Vertically-grown nanowires have been individually contacted using electron-beam lithography, although only one nanowire showed field dependence. All of these nanowires showed complete dissolution over an 8 month period, and a conducting-to-insulating transition over 48 hours. This transition can be temporarily reversed by exposure to an electron beam.

9.1.1 Hydrothermal synthesis of vertical ZnO nanowires

The influence of polyethylenimine (PEI) on the hydrothermal growth of vertical ZnO nanowires from seeded substrates was thoroughly investigated. Hydrothermal growths were carried out using PEI at a range concentrations between 2 mM to 8 mM, with PEI molecular weights of 800, 1300, and 2000 g/mol. These hydrothermal growths resulted in either homogeneous nanowires, hierarchical nanowires, thin ZnO films, growth suppression, or substrate etching, depending on the PEI concentration used.

The homogeneous ZnO nanowires were compared to control ZnO nanowires, which were grown without PEI. All of these homogeneous nanowires had improved

lengths over the control nanowires, which had average lengths of 1.6 μ m. The longest homogeneous nanowires were grown with 8 mM of PEI(M_W = 800 g/mol), with an average length of 10.5 μ m. Similarly, all of the homogeneous nanowires had improved aspect ratios over the control nanowires, which had average aspect ratios of 32. The nanowires with the highest aspect ratios were grown using 6 mM of PEI(M_W = 800 g/mol), which had average aspect ratios of 153.

The synthesis of hierarchical ZnO nanowires occurs at 4 mM of $PEI(M_W = 1300 \text{ g/mol})$. These nanowires consist of a large primary nanowire which abruptly terminates in a thin, long secondary nanowire. These secondary nanowires are up to 10 μ m long and below 50 nm in diameter. The synthesis of the hierarchical nanowires is attributed to two separate growth phases within the single PEI-mediated hydrothermal growth.

9.1.2 PEI-mediated growth of hierarchical ZnO nanowire FETs

ZnO nanowires were grown laterally from ZnO/Ti films that were patterned on SiO_2/Si device substrates. These were grown using identical concentrations and molecular weights of PEI used in the vertical ZnO nanowire synthesis. Hierarchical nanowires were synthesised using 8 mM of PEI(M_W = 800 g/mol), 3 & 4 mM of PEI(M_W = 1300 g/mol), and 2 & 3 mM of PEI(M_W = 2000 g/mol). FETs were fabricated by allowing the secondary nanowire portion of the hierarchical nanowires to intersect between pairs of ZnO/Ti films, which were used as source and drain electrodes. All of these hierarchical nanowire FETs showed significant field dependence without requiring any annealing step, with on-off ratios between $10^3 - 10^5$ and threshold voltages between -7.5 V to 5 V.

Persistent photoconductivity measurements of the hierarchical nanowire FETs show that depleting oxygen/hydroxyl surface states on the nanowire sufficiently reduce the charge carrier concentration to allow for as-grown field dependence. These conclusions are supported by persistent photoconductivity measurements of field-independent FETs which are bridged with primary nanowires. The ZnO nanowire surface and the surface-to-volume ratio therefore play a critical role in determining their suitability in FET applications.

9.1.3 Modular growth of hierarchical ZnO nanowire FETs

The previous hydrothermal growth of hierarchical ZnO nanowires relies on two growth phases within a PEI-mediated hydrothermal growth. Lateral hierarchical

nanowires were grown on SiO₂/Si device substrates in a modular fashion by separating the two growth phases into individual primary and secondary hydrothermal growths. The preheat time of the primary growth and presence of PEI in both growths were investigated to optimise the reproducibility of the synthesis route. The most reproducible combination was found to be a primary growth preheated for 20 minutes, with neither the primary or secondary growth containing PEI. This yielded a reproducibility of 50%.

FETs were fabricated by bridging separated ZnO/Ti films on the SiO₂/Si substrate. These hierarchical nanowire FETs showed unreliable field dependence when measured in air, with only some devices responding to applied fields. This is in contrast with the PEI-mediated hierarchical nanowires FETs, which consistently show field dependence. The best modular-grown FET operates with an on-off ratio of 10^4 and a threshold voltage of ≈ 0 V. Devices which are field-independent in air can reliably be used as FETs by back-gating and top-gating the devices in a wet environment, using de-ionised H₂O as the dielectric. A device which is field-independent in air is wet back-gated, and operates with an on-off ratio of 10^5 and a threshold voltage of ≈ 8 V. Wet top-gated devices operate with on-off ratios of $10^3 - 10^4$, and threshold voltages within 0.4 - 0.9 V. These FETs have significantly low subthreshold swings on the order of 80 mV/decade.

9.1.4 Individually contacted ZnO nanowires

Vertical ZnO nanowires were hydrothermally synthesised on dedicated growth substrates and then transferred to SiO_2/Si and $HfO_2/SiO_2/Si$ device substrates. After being transferred, the nanowires were individually electrically contacted using electron beam lithography. The nanowires were grown using 1.5 mM of PEI(M_W = 2000 g/mol) and 8 & 6 mM of PEI(M_W = 800 g/mol). All of the contacted nanowires show back-to-back Schottky contact behaviour. Only one of these contacted nanowires showed field dependence, which was grown using 1.5 mM of PEI(M_W = 2000 g/mol). This FET operates with an on-off ratio of 10⁴ and a threshold voltage of -7 V. The transconductance of the FET is 1.65 nS, and the mobility is calculated to be 0.106 cm² V⁻¹ s⁻¹. Other nanowires contacted on the same device substrate do not show field dependence.

Secondary nanowires were also isolated from lateral PEI-mediated hierarchical nanowires and contacted on an SiO₂/Si device substrate. These hierarchical nanowires were grown from lateral substrates using 8 mM of PEI($M_W = 800$ g/mol). The SiO₂/Si substrate allows excessive leakage currents due to a damaged dielectric,

although one contacted nanowire can be measured. This FET shows field dependence, with an on-off ratio of 10^2 and a threshold voltage of -6.2 V. The transconductance of this FET is 30.1 nS, and the mobility is calculated to be 2.964 cm² V⁻¹ s⁻¹.

All of the vertical nanowires grown with $PEI(M_W = 800 \text{ g/mol})$ completely disappear from the device substrate over at most 8 months. This may be caused by carbon accumulation or by exposure to atmospheric water. Additionally, a portion of these nanowires undergo a conducting-to-insulating transition over 48 hours. This transition can be temporarily reversed by exposure to an electron beam. The conducting-to-insulating transition and its reversibility is likely caused by an accumulation of depleting surface states, which can be temporarily removed by an electron beam. This may be related to the mechanism which causes the disappearance of the nanowires over time.

9.2 Future work

This research topic could continue in several different directions if more time was devoted to it. These individual research projects could specifically continue the work of each results chapter; namely, the vertical hydrothermal growth of ZnO nanowires, the PEI-mediated and modular growth of hierarchical nanowire FETs, and the individually contacted ZnO nanowires. These further projects could improve the applications of ZnO nanowires in FETs from a practical point of view, and also shed light on fundamental questions regarding the material properties of ZnO nanowires which remain unanswered.

9.2.1 PEI-mediated hierarchical nanowire growth over time

The synthesis of hierarchical nanowires in a single PEI-mediated hydrothermal growth is attributed to two distinct growth phases, where the second growth phase is driven by a gradual release of Zn^{2+} ions which have complexed with PEI molecules. Strong empirical evidence has been presented in this thesis to argue that this is the case, but quantitative measurements of the growth solution would further clarify the exact growth mechanism, and the timeframe over which the secondary nanowires are grown. The concentration of zinc over time in the hydrothermal growth solution is able to be measured on the parts-per-million level by using atomic absorption spectroscopy [252], and the hydrothermal growth of ZnO nanowires can even be imaged in situ using transmission electron microscopy [253]. Measurements such as these could potentially identify when the secondary growth phase specifically begins during the hydrothermal growth condi-

tions could then be tailored to emulate the secondary growth phase over the duration of the entire hydrothermal growth, omitting the primary growth phase entirely. This could lead to a more efficient and reliable secondary nanowire synthesis, which may not require the use of PEI.

9.2.2 Hierarchical nanowire FETs as device platforms for biosensors

Modular hierarchical-nanowire FETs are measured in wet environments using deionised water as a gate dielectric in Chapter 7. This is used to achieve field dependence of the devices, allowing for operation in back-gate and top-gate configurations. Developing field-effect transistors which operate in wet environments paves the way for the fabrication of biosensors, which have been identified as an important emerging application of nanoelectronics [254]. Although field-effect transistors are important devices for semiconducting nanowire biosensors [255], biosensors are rarely based on ZnO nanowire field-effect transistors [256]. Biosensors which do use ZnO nanowires as field-effect transistors use individually contacted nanowires which are fabricated using high-temperature methods [257, 258]. The long-term applicability of these devices remains uncertain, as ZnO nanowires rapidly dissolve in many liquids used for biosensing applications [236], and are ideally disposable in the first place. This discourages devices which are fabricated using expensive or laborious processes, such as electron beam lithography.

In contrast, the fabrication of PEI-mediated and modular hierarchical nanowires in Chapters 6 and 7 is comparatively inexpensive and scalable, as the growth substrates also double as the device substrates. Furthermore, the hydrothermal growth is perfectly suited for industrial throughputs, as many nanowire devices and FETs can be grown in parallel. As such, there is a real opportunity to use both the PEI-mediated and modular hierarchical nanowire FETs presented in this thesis as device platforms for biosensing. This would require further research into the functionalisation of the surface of the hierarchical nanowires, and more research into maximising the reproducibility of electrical characteristics between devices. This may favour a modular approach to the hierarchical nanowire synthesis, as there is ostensibly more control over the growth conditions.

9.2.3 ZnO nanowire sensitivity to electron beam exposure

Some of the nanowires contacted in Chapter 8 show a drastic transition from an initially conductive state into an insulating state over 48 hours. This transition can be temporarily reversed and the conductivity restored by exposing the nanowires to an electron beam, although the nanowires return to an insulating state after another

48 hours. This is attributed to depleting oxygen and hydroxyl states on the surface of the nanowire which trap electrons. If the conducting-to-insulating transition is mediated by surface states which accumulate from the ambient environment, this may be exploited to serve as the basis for an environmental detector or monitor based on the resistance of a nanowire. Although it is impractical to refresh the conductance of the nanowire using an electron beam, a similar effect might possibly be achieved by using ultraviolet light.

In any case, experiments should be performed to conclusively determine what surface species are adsorping to affect the nanowire resistance, or whether the conductance transition is driven by a different mechanism. One possible technique that could be used is x-ray photoelectron spectroscopy (XPS), which is useful for determining the elemental makeup of the surface of materials. These experiments would most likely have to be conducted on separate nanowires with similar diameters which were contacted at the same time, rather than repeated measurements of a single nanowire, as the XPS measurements may intrinsically change the surface of the nanowire.

The electron-beam induced insulating-to-conducting transition of the material has wide implications for the characterisation of ZnO nanowires and ZnO nanowire devices. ZnO nanowire devices are typically imaged using techniques which involve exposure to electron beams, such as scanning electron microscopy, and techniques which use electron beams may be employed in their fabrication, such as electron beam lithography. The sensitivity of regular ZnO nanowires to these experimental methods should be determined in order to assess whether they may affect fabricated devices in unintended ways. The electron beam doses that the ZnO nanowires were exposed to in this thesis were relatively uncontrolled, as the aperture of an SEM was used as the electron beam source. A systematic study could be performed by using an electron beam lithography system, where the exact electron beam dose can be specified. Specified electron beam dosages could be followed up by XPS measurements, which would paint a comprehensive picture of the ZnO nanowire surface states, and how the surface may unintentionally change during regular experimental procedures such as SEM imaging.

9.2.4 ZnO nanowire dissolution over time

Nanowires which are contacted from UC1 and UC2 in Chapter 8 dissolve over an 8 month period. If there were sufficient time to explore this further, a series of experiments would be dedicated to tracking this dissolution more carefully over time at regular intervals, using further SEM measurements. These experiments would ideally

be performed in conjunction with further XPS measurements, to track how the surfaces of the nanowires change over time. Furthermore, this set of experiments would tie in to the transitions between conducting and insulating states of the nanowire, as the characterisations to determine how the surface states evolve over time may influence the results. These experiments may have wide-reaching implications regarding the ageing of ZnO nanowires and ZnO nanowire devices in the atmosphere.

Appendix A

Electron beam lithography procedure

A.1 Electron beam lithography

As discussed in Section 4.2.1 in Chapter 4, photolithography is a method of patterning areas of a substrate using ultraviolet light. The smallest feature size is therefore theoretically limited to the wavelength of the UV light, due to the diffraction limit. In practice, the maximum resolution of photolithography is usually much larger than the theoretical limit, and is typically on the order of microns. While this isn't an issue when defining relatively large electrodes or other features, it poses a problem if one wishes to make contact to individual nanowires, which requires resolutions on the order of hundreds or even tens of nanometres.

To this end, electron beam lithography (EBL) is a relatively commonly technique used to fabricate electronic devices using single nanowires. EBL is analogous to photolithography, although the lithographic resist is exposed using a beam of collimated electrons rather than ultraviolet light. This offers much greater resolution than photolithography, in the same way that electron microscopy offers much greater resolution than optical microscopy. It also affords a greater flexibility over the design of the features to be patterned, as the patterns are not constrained to a fixed mask, unlike photolithography. This allows for custom-made patterns designed on-the-fly to suit specific nanowires.

EBL has been used throughout this thesis to fabricate the FETs and other electronic devices which make use of single ZnO nanowires, as in Chapter 8. This appendix aims to describe the steps used to prepare single nanowires for patterning, and the EBL process itself.

A.1.1 Device substrate fabrication

The lateral ZnO nanowire FETs employed in Chapters 7 and 6 use a single substrate for nanowire growth and FET fabrication. In comparison, when contacting individual nanowires using electron beam lithography (EBL), nanowires are first grown on a dedicated growth substrate, and then transferred to a separate device substrate. The growth of vertical nanowires is described in Chapter 5. The fabrication of device substrates for EBL is very similar to the fabrication of lateral growth substrates (described in Section 4.2 of Chapter 4), although regular metals are used for electrodes instead of ZnO/Ti.

As the single-nanowire FETs fabricated in this work use a global back gate, we begin with an SiO₂ (100 nm) / p⁺-Si wafer, with the SiO₂ layer ultimately serving as a gate dielectric. The wafers are cleaned and protected with AZ1518/PDMS before depositing a 5/50 nm Cr/Au layer on the wafer back to serve as a gate electrode. After the wafers have been protected and their backs metallised, they are cleaved into 1.5×1.1 cm substrates and the protective layers removed in acetone.

Electron beam lithography is designed to pattern and expose minute areas of the substrate, which makes it unsuitable for defining the comparatively large metal pads used to electrically measure the contacted nanowires via probing. As such, we use a photolithography step to define the bulk area of the electrode structure, which is then in turn contacted using EBL. To this end, after the substrates have been cleaved and cleaned, they are prepared for photolithography as per Section 4.2.1 of Chapter 4. The AZ1518-coated substrates are exposed for 10 seconds using the Karl Suss MJB3, and developed in $3:1 \text{ AZ326:DI H}_2\text{O}$ for 40 seconds before being rinsed twice.

The mask used to define the electrodes is shown schematically in Figure A.1. Each chip features 15 pairs of electrodes, which will eventually be used to individually contact 15 separate nanowires. The gaps between electrodes are either 20 μ m or 40 μ m, depending on the exact mask used. The electrode gaps are intentionally grouped at the centre of the chip, which reduces any astigmatism or other imperfections in the electron beam lithography alignment. Because the dimensions and tolerances used in EBL are on the order of tens to hundreds of nanometres, small misalignments can completely ruin the EBL process. The mask also features a right-angle cross-bar. This cross-bar is used to align the EBL system, and also serves as a reference point during the EBL process.

After photolithography, Ti/Au (\approx 80/20 nm) is deposited on the substrates using the Angstrom Engineering NexDep evaporator. The Ti layer is deposited using a elec-



Figure A.1: (a) The photolithographic mask design used to define the large electrodes for EBL. The black portions of the schematic appear as holes in the mask, while the white portions are chrome. The cross-bar (shown on the far left of the schematic) is used for alignment purposes. (b) A close-up of one of the electrode gaps (not to scale.) The electrodes are spaced 20 μ m apart, and are 100 μ m in width. We have also made use of a 40 μ m gap variant of the mask.

tron gun powered by a Telemark TT6 power supply. The Ti used is 99.995% pure Ti wire sourced from Kurt J. Lesker, held in a graphite crucible. The Au (Regal Castings Ltd.) is resistively evaporated using a tungsten boat. The Ti layer serves as an adhesion layer, while the Au layer prevents oxidisation. Once the Ti/Au electrodes have been deposited, the device substrates are lifted off in acetone in cleaned using IPA.

A.1.2 Nanowire transfer and imaging

Once the device substrates have been fabricated, the vertical nanowires we wish to contact are transferred from their growth substrate to the device substrates. We have tried several methods of transfer, although the best method (and the method utilised in this thesis) is the tex-wipe transfer method, adopted from the nanoelectronics group at the University of New South Wales. A small triangle is cut from a cleanroom-grade tex-wipe using a dedicated pair of scissors, with particular care taken to obtain sharp and crisp corners free of fibers. The tex-wipe triangle is held using a pair of tweezers and the tip of a corner gently drawn across the surface of the nanowire growth substrate several times, which accumulates nanowires on the tex-wipe tip.

The tip of the corner is then carefully "dotted" between the electrode gaps on the device substrate, akin to pointillism – indeed, this particular transfer method is perhaps more of an art than a science, requiring a steady hand not only to place the tex-wipe tip roughly in between the electrodes, but also to transfer the right amount of nanomaterials. If too few nanowires are transferred, there won't be enough nanowires suitable for contact via EBL. Conversely, if too many nanowires are transferred, the electrode gaps may be unintentionally bridged with nanowires, or the nanowires clumped in too dense aggregates, which makes it impossible to contact single nanowires individually. It is best to err on the side of transferring too few nanowires, and then assess using optical microscopy. If more nanowires are required, the tex-wipe tip can be re-applied to the device substrate (or growth substrate) as many times as necessary. As with everything else in life, practice makes perfect.

The device substrate is imaged using scanning electron microscopy (SEM) once a suitable number of nanowires have been transferred from the growth substrate. These images serve as a record of the locations of the nanowires. During the EBL process, the nanowires will be very difficult to see underneath the EBL resist, so it is critical to know where the nanowires lie between the electrodes. The images also have to contain suitable reference points, such as small irregularities in the photolithographically-defined electrode edges or comparatively large nanostructures, which will be visible underneath the EBL resist layer. The most useful images are taken at a magnification roughly equivalent to the images acquired during the EBL process. This magnification is fixed by the width of the writefield (explained in Section A.1.3). This corresponds to 2,400x magnification for the EBL performed in this thesis. The bulk of the device substrates in this thesis have been imaged using Victoria University's JEOL 6500F FESEM, which is only capable of acquiring images at quantised magnification values. The closest magnification for this SEM is 2,500x.

It is also useful to map the entire electrode gap and nearby regions of the device substrate using SEM, which requires 9 to 12 separate SEM images for each pair of electrodes, depending on the electrode gap length (20 μ m or 40 μ m.) Figure A.2 shows a single SEM image of a device substrate - nanowires are clearly visible between the photolithographically-defined electrodes, the edges of which are also in the frame of the image. The bottom-left inset of Figure A.2 shows a pair of electrodes roughly divided into $25 \times 25 \ \mu$ m areas, which need to be individually imaged to map the entire electrode area. This ensures that the location of any nanowire suitable for contact via EBL has been captured, and can be inferred from reference points during the EBL process. The main body of Figure A.2 is taken from position 5.

A.1.3 Electron beam lithography

The first step in the EBL process is to apply an e-beam resist, which is insoluble in developer unless exposed to an electron beam. We use a bi-layer stack of polymethylmetharcylite (PMMA) as our resist, which is a very common polymer used widely as an e-beam resist throughout the literature. The first layer of PMMA has a low molecular weight (LMW) ($M_W = 120,000 \text{ g/mol}$), which is dissolved in chlorobenzene at 4% by weight. This is spincoated onto the substrate surface at 3000 RPM for 1 minute using a spin coater. After the first layer of PMMA has been applied, the device substrate is placed into a convection oven for 30 minutes at 80°C to bake off excess solvent. After this first soft bake, a high molecular weight (HMW) ($M_W = 996,000$ g/mol) layer of PMMA is applied to the device substrate via spin coating at 3000 RPM for 1 minute. This HMW PMMA is dissolved in O-xylene at 2.5% by weight. After the HMW layer has been applied, the substrate is returned into the convection oven for another 30 minutes to ensure there is no residual solvent. Both of the LMW and HMW PMMA solutions have been prepared at the University of Canterbury.

The bi-layer stack of PMMA is used in favour of a single PMMA layer as it creates an undercut in the resist after exposure and development. The LMW PMMA is



Figure A.2: An SEM image taken between a pair of electrodes from a device substrate. Nanowires are clearly visible, and can be located later on by comparing their position with a point on the electrode edge. Inset: A map showing a pair of electrodes separated into $25 \times 25 \ \mu$ m areas, which must be individually imaged to completely map the electrode pair.

more susceptible to e-beam exposure, and a comparatively larger portion is exposed compared to the HMW layer. When the exposed portions of the HMW and LMW layers are removed through development, the LMW layer slightly undercuts the HMW layer, as shown schematically in Figure A.3. After the samples are eventually metallised, the LMW undercut significantly improves the liftoff, as the deposited metal is not in contact with the sidewall of the LMW PMMA. This is critical for high-resolution lithography with small patterned features. The height of the bi-layer is typically 300 nm thick, which has been measured using a Veeco Dektak 150 profilometer.



Figure A.3: (a) The PMMA bi-layer prior to exposure. (b) After exposure and development, the LMW layer undercuts the HMW layer. (c) Liftoff is made easier after metallisation as the metal on the sample doesn't touch the PMMA sidewall.

After the device substrates have spun with PMMA, the nanowires are ready to be contacted via EBL. The bulk of the single nanowire devices in this thesis were contacted using the Raith 150 at the University of Canterbury. The Raith 150 doubles as an SEM and as an EBL system. Samples are attached to the EBL chuck by a hinged clip in the loadlock. After the loading procedure, a robot arm moves the chuck into the EBL chamber. The chuck is positioned within the chamber using a joystick, and an internal infrared camera is used to gauge the position of the sample. All of the EBL work using the Raith 150 was carried out with an accelerating voltage of 10 kV and an aperture size of 10 μ m. It should be noted that there are several steps taken to ensure the alignment of the electron beam is correct, along with measurements of the beam current. The description of these steps is omitted for brevity.

Imaging the device substrates during EBL poses an obvious problem. Other than the use of the infrared camera, the only way to image the substrates is using the Raith 150 as an SEM. However, this unintentionally exposes vast areas of the PMMA to the electron beam, meaning the portions of the PMMA imaged via SEM will be dissolved during development. Therefore most of the EBL work is done semi-blind, with the electron beam blanked for most of the time. To do this, the sample is first positioned slightly outside of the SEM column using the joystick controller and infrared camera. Once the sample is just out of view of the column, the beamline is unblanked and the corner of the substrate brought into the field of view. Once the corner of the substrate can be seen, the chuck is rapidly moved so that the beam traverses to the edge of the cross-bar. While this does expose the PMMA between the substrate corner and the cross-bar, the nanowires and electrode gaps are far in the centre of the substrate, ensuring they aren't accidentally exposed to the electron beam. This procedure is shown schematically in Figure A.4.



Figure A.4: A schematic showing the (u, v) = (0, 0) origin definition. The SEM field of view (red dot) starts (1) outside of the sample, before (2) moving to the corner of the sample, and from there moving (3) to the edge of the cross-bar, where the (u, v) origin is defined.

The Raith 150 uses a set of coordinates, (u, v), that are separate from the lab coordinates (x, y). The (u, v) coordinate system allows for navigation of the device substrate without directly imaging the sample using SEM. First, a (u, v) = (0, 0) origin is established at the edge of the cross-bar on the substrate. The (u, v) coordinate system is initially parallel to the (x, y) coordinate system, and usually needs to be rotated to account for the angle θ between the two coordinate systems. Two points on the crossbar(P1 and P2) are used to define the u axis, and the (u, v) coordinate system is then rotated by θ , schematically shown in Figure A.5. This aligns the (u, v) coordinate system with the layout of the chip.

Once a suitable (u, v) = (0, 0) origin has been established and the coordinate system has been rotated, the sample can be navigated by moving the chuck using stepper motors. These stepper motors allow movement in both the u or v directions while the electron beam is blanked. If the (u, v) = (0, 0) origin is a known point on



Figure A.5: A schematic showing the angle correction of the *u* axis. The (u, v) and (x, y) axes are initially parallel, but as the sample is usually loaded at an angle, the axis needs to be slightly rotated. Two points P1 and P2, shown in (a), are chosen from the crossbar, which is used to define the *u* axis. After the coordinate system has been rotated by θ , it is aligned with the substrate, which makes navigation of the sample easier.

the photolithographically-defined Ti/Au pattern, and the photolithographic pattern is known, the electrode gaps can be reliably navigated to without exposing the sample to the electron beam. This process is shown in Figure A.6.



Figure A.6: The path taken from the (u, v) origin to the first writefield via stepper motors is shown by the red line. This ensures the PMMA remains unexposed to the electron beam.

When the electrode gaps are underneath the SEM column, the actual EBL process can begin. A snapshot of the sample is acquired with a "slowscan", which is a low-resolution SEM image captured in a single raster. The dimensions of the slowscan are determined by the size of the writefield (WF), which is the maximum area the Raith 150 can pattern for a fixed stage position. We consistently use a writefield size of 25×25 microns for all of the EBL performed using the Raith 150. Although the slowscan does expose the entire writefield to the electron beam, the exposure from a single slowscan isn't significant enough to affect the PMMA. However, if 3 or 4 consecutive slowscans are taken of the same writefield, the PMMA will be exposed and removed during the development process.

The majority of the nanowires underneath the PMMA are unable to be seen in the slowscan image. This is why SEM images of the nanowires are taken before the PMMA bi-layer is applied. Figure A.7 shows an SEM image of part of a channel, taken before the sample has been spincoated with PMMA, and a slowscan of the same area. The nanowire to be contacted is highlighted with a dotted box, and shown in higher magnification in the upper-right inset. While this nanowire is invisible underneath the PMMA, a few large nanowires can be seen in both SEM and slowscan images. These large nanowires (numbered 1, 2, and 3) are used as common reference points to triangulate the position of the nanowire to be contacted in the slowscan image. This is most easily done by measuring the distance of the nanowire ends from the reference points in the SEM image using ImageJ, which is a free piece of software designed for scientific micrograph analysis.



Figure A.7: (a) An SEM image of nanowires deposited on a channel, before PMMA has been spincoated. A box highlights the nanowire to be contacted, which is shown in the upper-right inset. (b) A corresponding slowscan image of the same location. Large nanowires (labelled 1, 2, and 3 in both images) can be used to triangulate the position of the nanowire.

Once the nanowire can be reliably located in the slowscan image, the EBL pattern is designed by dictating which portions of the writefield are to be exposed using the GDS editor, which is part of the Raith 150 software. These designs are superimposed over the slowscan image. The pattern is positioned in relation to the reference points (measured from the SEM images) by comparing the distance between points on the slowscan image using the coordinate system. To make this process easier, the Raith 150 introduces a third set of coordinates (U, V) local to the writefield, with the centre of the writefield designated (U, V) = (0, 0). Figure A.8 shows the pattern designed in the GDS editor and its overlay with the slowscan image.

After the pattern has been designed and positioned accordingly, the Raith 150



Figure A.8: The GDS editor (left), used to design custom patterns, and the slowscan overlay (right) relating the pattern to the writefield.

relates the (U, V) coordinate system back to the lab (x, y) coordinate system and the writefield is finally exposed. It is often necessary to connect the exposed regions of the nanowire/writefield with the large photolithographically defined electrodes, which is typically more than a writefield's distance away. The stepper motor is moved 20 microns towards the electrode, and a new slowscan image is taken. Because the size of the writefield is 25 microns, the new and old writefields partially overlap. A large square (usually 22 × 22 microns) is written in the new writefield, which connects the initial patterns. This process is repeated until the electrode is thoroughly exposed, and then the stage is stepped an equal number of times in the opposite direction to return it back to the initial writefield. The process is then repeated in the opposite direction until the second electrode has been exposed, ensuring that the EBL-exposed portions of the PMMA will connect to both electrodes.

This process is schematically shown in Figure A.9. The nanowire contact design is shown in blue. The stepper motor has moved 20 μ m to the right, and the edge of the new writefield is shown by the black dotted line that runs vertically down the centre of the figure. The 22 × 22 μ m square is shown by the yellow dotted box in the new writefield. The part where the two writefields overlap is shown by the blue diagonally-shaded area.



Figure A.9: Overlapping writefields to create a continuous exposure using the GDS editor.

A.1.4 Development and metal deposition

Once the samples have been patterned, the exposed PMMA is dissolved through a development step, akin to photolithographic development. Samples are gently agitated in a 1:3 mixture of methyl isobutyl ketone (MIBK):IPA for 40 seconds, which is a very common developer used in conjunction with PMMA throughout the literature. After the samples have been developed, they are rinsed in pure IPA to remove any residue.

Although the nanowires are very small, the EBL results can usually be assessed using optical microscopy, provided that the PMMA channel isn't too thin to resolve. Figure A.10 shows an optical microscope image of a sample after development. The photolithographically defined electrodes (referred to as the PL electrodes) can be seen in the top-right and bottom-left of the image. The bulk of the electrodes are still underneath the PMMA, although windows have been opened on each electrode through EBL and development. The channel between the two PL electrodes runs diagonally through the centre of the image. This too is underneath PMMA, although leads have been defined through EBL to make contact to the nanowire. The nanowire itself is highlighted by the dashed black box, and is shown in high-contrast in the upper-right corner of the image. Although this image has been taken with an optical microscope, the two individual ends of the nanowire can be seen, as well as the PMMA channel separating the nanowire contacts. The PMMA channel is particularly large in this image, and can be much more difficult to resolve if the nanowire contacts are spaced closer together.



Figure A.10: Optical image of an EBL sample post-development.

Bibliography

- [1] M. Riordan and L. Hoddeson, *Crystal fire: the invention of the transistor and the birth of the information age*. Norton, 1998.
- [2] J. F. Wager, "Transparent electronics," *Science*, vol. 300, no. 5623, pp. 1245–1246, 2003.
- [3] P. J. Pauzauskie and P. Yang, "Nanowire photonics," *Materials Today*, vol. 9, no. 10, pp. 36 45, 2006.
- [4] P. G. Neudeck, R. S. Okojie, and L.-Y. Chen, "High-temperature electronics - a role for wide bandgap semiconductors?," *Proceedings of the IEEE*, vol. 90, pp. 1065–1076, Jun 2002.
- [5] "Nanomarkets: LCD to dominate the transparent display market in the next few years, but OLED adoption will rapidly pick up." URL: http://www.oledinfo.com/nanomarkets-lcd-dominate-transparent-display-market-next-fewyears-oled-adoption-will-rapidly-pick. [Online; accessed 05-April-2017].
- [6] "International technology roadmap for semiconductors 2.0 2015 edition," Available at: http://www.itrs2.net/.
- [7] R. Courtland, "Transistors could stop shrinking in 2021," *IEEE Spectrum: Technology, Engineering, and Science News*, Jul 2016.
- [8] P. McLellan, "Imec's An Steegen Talks Future Process Technology." https://www.semiwiki.com/forum/content/ 4757-imecs-steegen-talks-future-process-technology.html, Jun 2015.
- [9] Z. Y. Fan and J. G. Lu, "Zinc oxide nanostructures: Synthesis and properties," *Journal of Nanoscience and Nanotechnology*, vol. 5, no. 10, pp. 1561–1573, 2005.
- [10] S. Xu and Z. L. Wang, "One-dimensional ZnO nanostructures: Solution growth and functional properties," *Nano Research*, vol. 4, pp. 1013–1098, Aug 2011.

- [11] C. P. Burke-Govey and N. O. V. Plank, "Review of hydrothermal ZnO nanowires: Toward FET applications," *Journal of Vacuum Science & Technology B: Microelectronics and Nanometer Structures*, vol. 31, no. 6, p. 06F101, 2013.
- [12] U. Ozgur, Y. I. Alivov, C. Liu, A. Teke, M. A. Reshchikov, S. Dogan, V. Avrutin, S.-J. Cho, and H. Morkoc, "A comprehensive review of ZnO materials and devices," *Journal of Applied Physics*, vol. 98, no. 4, p. 041301, 2005.
- [13] D. Look, "Recent advances in ZnO materials and devices," *Materials Science and Engineering: B*, vol. 80, pp. 383–387, Mar 2001.
- [14] "Wurtzite polyhedra." https://en.wikipedia.org/wiki/File: Wurtzite_polyhedra.png. [Online; accessed 06-April-2017].
- [15] C. Klingshirn, J. Fallert, H. Zhou, J. Sartor, C. Thiele, F. Maier-Flaig, D. Schneider, and H. Kalt, "65 years of ZnO research - old and very recent results," *Physica Status Solidi B*, vol. 247, pp. 1424–1447, Apr 2010.
- [16] Y. Qin, X. Wang, and Z. L. Wang, "Microfibre-nanowire hybrid structure for energy scavenging.," *Nature*, vol. 451, pp. 809–13, Feb 2008.
- [17] Z. L. Wang, "Piezopotential gated nanowire devices: Piezotronics and piezophototronics," *Nano Today*, vol. 5, no. 6, pp. 540 – 552, 2010.
- [18] L. Schmidt-Mende and J. MacManus-Driscoll, "ZnO nanostructures, defects, and devices," *Materials Today*, vol. 10, no. 5, pp. 40–48, 2007.
- [19] C. Jagadish and S. Pearton, *Zinc Oxide Bulk, Thin Films and Nanostructures: Processing, Properties, and Applications.* Elsevier Science, 2011.
- [20] U. Ozgur, D. Hofstetter, and H. Morkoç, "ZnO devices and applications: a review of current status and future prospects," *Proceedings of the IEEE*, 2010.
- [21] D. Look, D. Reynolds, J. Sizelove, R. Jones, C. Litton, G. Cantwell, and W. Harsch, "Electrical properties of bulk ZnO," *Solid State Communications*, vol. 105, no. 6, pp. 399 – 401, 1998.
- [22] A. F. Kohan, G. Ceder, D. Morgan, and C. G. Van de Walle, "First-principles study of native point defects in ZnO," *Phys. Rev. B*, vol. 61, pp. 15019–15027, Jun 2000.
- [23] S. B. Zhang, S.-H. Wei, and A. Zunger, "Intrinsic n-type versus p-type doping asymmetry and the defect physics of ZnO," *Phys. Rev. B*, vol. 63, p. 075205, Jan 2001.

- [24] F. Oba, S. R. Nishitani, S. Isotani, H. Adachi, and I. Tanaka, "Energetics of native defects in ZnO," *Journal of Applied Physics*, vol. 90, no. 2, pp. 824–828, 2001.
- [25] Van De Walle CG, "Hydrogen as a cause of doping in zinc oxide," *Physical Review Letters*, vol. 85, pp. 1012–5, Jul 2000.
- [26] S. F. J. Cox, E. A. Davis, S. P. Cottrell, P. J. C. King, J. S. Lord, J. M. Gil, H. V. Alberto, R. C. Vilão, J. Piroto Duarte, N. Ayres de Campos, A. Weidinger, R. L. Lichti, and S. J. C. Irvine, "Experimental Confirmation of the Predicted Shallow Donor Hydrogen State in Zinc Oxide," *Physical Review Letters*, vol. 86, pp. 2601–2604, Mar 2001.
- [27] O. Schmidt, P. Kiesel, D. Ehrentraut, T. Fukuda, and N. Johnson, "Electrical characterization of ZnO, including analysis of surface conductivity," *Applied Physics A*, vol. 88, pp. 71–75, Jul 2007.
- [28] S. Akasaka, K. Nakahara, A. Tsukazaki, A. Ohtomo, and M. Kawasaki, "MgZnO Films with a Low Residual Donor Concentration (≤ 10¹⁵ cm⁻³) Grown by Molecular Beam Epitaxy," *Applied Physics Express*, vol. 3, no. 7, p. 071101, 2010.
- [29] D. Tanoff, M. Al-Khalfioui, C. Deparis, B. Vinter, M. Teisseire, C. Morhain, and J.-M. Chauveau, "Residual and nitrogen doping of homoepitaxial nonpolar mplane ZnO films grown by molecular beam epitaxy," *Applied Physics Letters*, vol. 98, no. 13, p. 131915, 2011.
- [30] A. D. L. Bugallo, F. Donatini, C. Sartel, V. Sallet, and J. Pernot, "Metallic core conduction in unintentionally doped ZnO nanowire," *Applied Physics Express*, vol. 8, no. 2, p. 025001, 2015.
- [31] D. Kaelblein, *Field-effect Transistors Based on ZnO Nanowires*. PhD thesis, ECOLE POLYTECHNIQUE FEDERALE DE LAUSANNE, 2011.
- [32] J. D. Albrecht, P. P. Ruden, S. Limpijumnong, W. R. L. Lambrecht, and K. F. Brennan, "High field electron transport properties of bulk ZnO," *Journal of Applied Physics*, vol. 86, no. 12, pp. 6864–6867, 1999.
- [33] F. Bertazzi, M. Goano, and E. Bellotti, "Electron and Hole Transport in Bulk ZnO: A Full Band Monte Carlo Study," *Journal of Electronic Materials*, vol. 36, pp. 857– 863, Aug 2007.
- [34] B. Guo, U. Ravaioli, and M. Staedele, "Full band Monte Carlo calculations of velocity-field characteristics of wurtzite ZnO," *Computer Physics Communications*, vol. 175, no. 7, pp. 482 – 486, 2006.

- [35] S. N. Cha, J. E. Jang, Y. Choi, G. a. J. Amaratunga, G. W. Ho, M. E. Welland, D. G. Hasko, D.-J. Kang, and J. M. Kim, "High performance ZnO nanowire field effect transistor using self-aligned nanogap gate electrodes," *Applied Physics Letters*, vol. 89, no. 26, p. 263102, 2006.
- [36] S. Ju, K. Lee, M.-H. Yoon, A. Facchetti, T. J. Marks, and D. B. Janes, "High performance ZnO nanowire field effect transistors with organic gate nanodielectrics: effects of metal contacts and ozone treatment," *Nanotechnology*, vol. 18, no. 15, p. 155201, 2007.
- [37] W. I. Park, J. S. Kim, G.-C. Yi, M. H. Bae, and H.-J. Lee, "Fabrication and electrical characteristics of high-performance ZnO nanorod field-effect transistors," *Applied Physics Letters*, vol. 85, no. 21, p. 5052, 2004.
- [38] C. Klingshirn, "ZnO: material, physics and applications," *ChemPhysChem*, vol. 8, pp. 782–803, Apr 2007.
- [39] S. Sasa, T. Hayafuji, M. Kawasaki, A. Nakashima, K. Koike, M. Yano, and M. Inoue, "High-field characteristics of ZnO and ZnO/ZnMgO heterostructures," *Physica Status Solidi* (c), vol. 5, no. 1, pp. 115–118, 2008.
- [40] S. Baruah and J. Dutta, "Hydrothermal growth of ZnO nanostructures," *Science and Technology of Advanced Materials*, vol. 10, p. 013001, Jan 2009.
- [41] Z. Dai, Z. Pan, and Z. Wang, "Novel nanostructures of functional oxides synthesized by thermal evaporation," *Advanced Functional Materials*, vol. 13, no. 1, pp. 9–24, 2003.
- [42] W. Lee, M.-C. Jeong, and J.-M. Myoung, "Catalyst-free growth of ZnO nanowires by metal-organic chemical vapour deposition (MOCVD) and thermal evaporation," *Acta Materialia*, vol. 52, pp. 3949–3957, Aug 2004.
- [43] B. D. Yao, Y. F. Chan, and N. Wang, "Formation of ZnO nanostructures by a simple way of thermal evaporation," *Applied Physics Letters*, vol. 81, no. 4, p. 757, 2002.
- [44] J. B. Baxter and E. S. Aydil, "Epitaxial growth of ZnO nanowires on a- and c-plane sapphire," *Journal of Crystal Growth*, vol. 274, pp. 407–411, Feb 2005.
- [45] C. Geng, Y. Jiang, Y. Yao, X. Meng, J. A. Zapien, C. S. Lee, Y. Lifshitz, and S. T. Lee, "Well-Aligned ZnO Nanowire Arrays Fabricated on Silicon Substrates," *Advanced Functional Materials*, vol. 14, pp. 589–594, Jun 2004.

- [46] W. Mai, P. Gao, C. Lao, Z. L. Wang, A. K. Sood, D. L. Polla, and M. B. Soprano, "Vertically aligned ZnO nanowire arrays on GaN and SiC substrates," *Chemical Physics Letters*, vol. 460, pp. 253–256, Jul 2008.
- [47] M. A. Verges, A. Mifsud, and C. J. Serna, "Formation of rod-like zinc oxide microcrystals in homogeneous solutions," *Journal of the Chemical Society, Faraday Transactions*, vol. 86, no. 6, p. 959, 1990.
- [48] L. Vayssieres, "Growth of Arrayed Nanorods and Nanowires of ZnO from Aqueous Solutions," *Advanced Materials*, vol. 15, pp. 464–466, Mar 2003.
- [49] S. Xu, Y. Wei, M. Kirkham, J. Liu, W. Mai, D. Davidovic, R. L. Snyder, and Z. L. Wang, "Patterned growth of vertically aligned ZnO nanowire arrays on inorganic substrates at low temperature without catalyst.," *Journal of the American Chemical Society*, vol. 130, no. 45, pp. 14958–14959, 2008.
- [50] H. Zhang, D. Yang, Y. Ji, X. Ma, J. Xu, and D. Que, "Low Temperature Synthesis of Flowerlike ZnO Nanostructures by Cetyltrimethylammonium Bromide-Assisted Hydrothermal Process," *Journal of Physical Chemistry B*, vol. 108, no. 13, pp. 3955– 3958, 2004.
- [51] D. Polsongkram, P. Chamninok, S. Pukird, L. Chow, O. Lupan, G. Chai, H. Khallaf, S. Park, and A. Schulte, "Effect of synthesis conditions on the growth of ZnO nanorods via hydrothermal method," *Physica B: Condensed Matter*, vol. 403, no. 19, pp. 3713 – 3717, 2008.
- [52] A. Kawska, P. Duchstein, O. Hochrein, and D. Zahn, "Atomistic mechanisms of ZnO aggregation from ethanolic solution: Ion association, proton transfer, and self-organization," *Nano Letters*, vol. 8, no. 8, pp. 2336–2340, 2008.
- [53] M. N. R. Ashfold, R. P. Doherty, N. G. Ndifor-Angwafor, D. J. Riley, and Y. Sun, "The kinetics of the hydrothermal growth of ZnO nanostructures," *Thin Solid Films*, vol. 515, no. 24 SPEC. ISS., pp. 8679–8683, 2007.
- [54] C. Opoku, A. S. Dahiya, C. Oshman, C. Daumont, F. Cayrel, G. Poulin-Vittrant, D. Alquier, and N. Camara, "Fabrication of high performance field-effect transistors and practical Schottky contacts using hydrothermal ZnO nanowires," *Nanotechnology*, vol. 26, no. 35, p. 355704, 2015.
- [55] R. Kozhummal, *Structural investigations on semiconductor nanostructures: wet chemical approaches for the synthesis of novel functional structures.* PhD thesis, University of Twente, 2014.

- [56] A. Sugunan, H. C. Warad, M. Boman, and J. Dutta, "Zinc oxide nanowires in chemical bath on seeded substrates: Role of hexamine," *Journal of Sol-Gel Science and Technology*, vol. 39, no. 1 SPEC. ISS., pp. 49–56, 2006.
- [57] K. M. McPeak, T. P. Le, N. G. Britton, Z. S. Nickolov, Y. A. Elabd, and J. B. Baxter, "Chemical bath deposition of ZnO nanowires at near-neutral pH conditions without hexamethylenetetramine (HMTA): Understanding the role of HMTA in ZnO nanowire growth," *Langmuir*, vol. 27, no. 7, pp. 3672–3677, 2011.
- [58] R. Parize, J. Garnier, O. Chaix-Pluchery, C. Verrier, E. Appert, and V. Consonni, "Effects of Hexamethylenetetramine on the Nucleation and Radial Growth of ZnO Nanowires by Chemical Bath Deposition," *Journal of Physical Chemistry C*, vol. 120, no. 9, pp. 5242–5250, 2016.
- [59] F. Zhang, Y. Ding, Y. Zhang, X. Zhang, and Z. L. Wang, "Piezo-phototronic effect enhanced visible and ultraviolet photodetection using a ZnO-CdS core-shell micro/nanowire," ACS Nano, vol. 6, no. 10, pp. 9229–9236, 2012.
- [60] B. Weintraub, Z. Zhou, Y. Li, and Y. Deng, "Solution synthesis of onedimensional ZnO nanomaterials and their applications," *Nanoscale*, vol. 2, pp. 1573–1587, 2010.
- [61] M. Ladanov, M. K. Ram, G. Matthews, and A. Kumar, "Structure and Optoelectrochemical Properties of ZnO Nanowires Grown on n-Si Substrate," *Langmuir*, vol. 27, no. 14, pp. 9012–9017, 2011. PMID: 21688806.
- [62] G. N. Narayanan and K. Annamalai, "Role of hexamethylenetetramine concentration on structural, morphological, optical and electrical properties of hydrothermally grown zinc oxide nanorods," *Journal of Materials Science: Materials in Electronics*, vol. 27, pp. 12209–12215, Nov 2016.
- [63] K. Govender, D. S. Boyle, P. B. Kenway, and P. O'Brien, "Understanding the factors that govern the deposition and morphology of thin films of zno from aqueous solution," *Journal of Materials Chemistry*, vol. 14, pp. 2575–2591, 2004.
- [64] S. Xu, N. Adiga, S. Ba, T. Dasgupta, C. F. J. Wu, and Z. L. Wang, "Optimizing and Improving the Growth Quality of ZnO Nanowire Arrays Guided by Statistical Design of Experiments.," ACS Nano, vol. 3, pp. 1803–12, Jul 2009.
- [65] J. Kim, D. Andeen, and F. Lange, "Hydrothermal Growth of Periodic, Single-Crystal ZnO Microrods and Microtunnels," *Advanced Materials*, vol. 18, no. 18, pp. 2453–2457, 2006.

- [66] Z. R. Tian, J. A. Voigt, J. Liu, B. Mckenzie, and M. J. Mcdermott, "Biomimetic Arrays of Oriented Helical ZnO Nanorods and Columns," *Journal of the American Chemical Society*, vol. 124, no. 44, pp. 12954–12955, 2002. PMID: 12405815.
- [67] Z. Yang, Q. Liu, H. Yu, B. Zou, Y. Wang, and T. H. Wang, "Substrate-free growth, characterization and growth mechanism of ZnO nanorod close-packed arrays," *Nanotechnology*, vol. 19, no. 3, p. 035704, 2008.
- [68] Y. Zhou, W. Wu, G. Hu, H. Wu, and S. Cui, "Hydrothermal synthesis of ZnO nanorod arrays with the addition of polyethyleneimine," *Materials Research Bulletin*, vol. 43, pp. 2113–2118, Aug 2008.
- [69] J. Qiu, X. Li, F. Zhuge, X. Gan, X. Gao, W. He, S.-J. Park, H.-K. Kim, and Y.-H. Hwang, "Solution-derived 40 microm vertically aligned ZnO nanowire arrays as photoelectrodes in dye-sensitized solar cells.," *Nanotechnology*, vol. 21, p. 195602, May 2010.
- [70] X. Liu, P. Lin, X. Yan, Z. Kang, Y. Zhao, Y. Lei, C. Li, H. Du, and Y. Zhang, "Enzyme-coated single ZnO nanowire FET biosensor for detection of uric acid," *Sensors and Actuators, B: Chemical*, vol. 176, pp. 22–27, 2013.
- [71] J. Yeo, S. Hong, G. Kim, H. Lee, Y. D. Suh, I. Park, C. P. Grigoropoulos, and S. H. Ko, "Laser-Induced Hydrothermal Growth of Heterogeneous Metal-Oxide Nanowire on Flexible Substrate by Laser Absorption Layer Design," ACS Nano, vol. 9, no. 6, p. 150602121228005, 2015.
- [72] P. Soman, M. Darnell, M. D. Feldman, and S. Chen, "Growth of High-Aspect Ratio Horizontally-Aligned ZnO Nanowire Arrays," *Journal of Nanoscience and Nanotechnology*, vol. 11, pp. 6880–6885, Aug 2011.
- [73] C. Xu, P. Shin, L. Cao, and D. Gao, "Preferential Growth of Long ZnO Nanowire Array and Its Application in Dye-Sensitized Solar Cells," *Journal of Physical Chemistry C*, vol. 114, pp. 125–129, Jan 2010.
- [74] Z. Han, S. Li, J. Chu, and Y. Chen, "Controlled growth of well-aligned ZnO nanowire arrays using the improved hydrothermal method," *Journal of Semiconductors*, vol. 34, p. 063002, Jun 2013.
- [75] K. Liu, W. Wu, B. Chen, X. Chen, and N. Zhang, "Continuous growth and improved PL property of ZnO nanoarrays with assistance of polyethylenimine," *Nanoscale*, vol. 5, no. 13, pp. 5986–5993, 2013.
- [76] "Branched PEI." https://commons.wikimedia.org/wiki/File: Branched_PEI.png. [Online; accessed 17-July-2017].

- [77] K. Liu, W. Wu, B. Chen, X. Chen, and N. Zhang, "Supplementary Information: Continuous growth and improved PL property of ZnO nanoarrays with assistance of polyethylenimine," *Nanoscale*, vol. 5, no. 13, pp. 5986–5993, 2013.
- [78] A. Ratkovich and R. Penn, "Zinc oxide nanoparticle growth from homogenous solution: Influence of Zn:OH, water concentration, and surfactant additives," *Materials Research Bulletin*, vol. 44, pp. 993–998, 5 2009.
- [79] A. S. Ratkovich and R. L. Penn, "Controlling Nanosized ZnO Growth Kinetics Using Various Zn:OH Concentration Ratios," *Journal of Physical Chemistry C*, vol. 111, no. 38, pp. 14098–14104, 2007.
- [80] S. Xu, Y. Qin, C. Xu, Y. Wei, R. Yang, and Z. L. Wang, "Self-powered nanowire devices.," *Nature Nanotechnology*, vol. 5, pp. 366–73, May 2010.
- [81] L. E. Greene, M. Law, D. H. Tan, M. Montano, J. Goldberger, G. Somorjai, and P. Yang, "General Route to Vertical ZnO Nanowire Arrays Using Textured ZnO Seeds," *Nano Letters*, vol. 5, no. 7, pp. 1231–1236, 2005. PMID: 16178216.
- [82] S. Baruah and J. Dutta, "Effect of seeded substrates on hydrothermally grown ZnO nanorods," *Journal of Sol-Gel Science and Technology*, vol. 50, p. 456, Feb 2009.
- [83] H. Ghayour, H. Rezaie, S. Mirdamadi, and A. Nourbakhsh, "The effect of seed layer thickness on alignment and morphology of ZnO nanorods," *Vacuum*, vol. 86, no. 1, pp. 101 – 105, 2011.
- [84] W.-Y. Wu, C.-C. Yeh, and J.-M. Ting, "Effects of Seed Layer Characteristics on the Synthesis of ZnO Nanowires," *Journal of the American Ceramic Society*, vol. 92, no. 11, pp. 2718–2723, 2009.
- [85] G. Kenanakis, D. Vernardou, E. Koudoumas, and N. Katsarakis, "Growth of c-axis oriented zno nanowires from aqueous solution: The decisive role of a seed layer for controlling the wires diameter," *Journal of Crystal Growth*, vol. 311, no. 23, pp. 4799 – 4804, 2009.
- [86] L.-W. Ji, S.-M. Peng, J.-S. Wu, W.-S. Shih, C.-Z. Wu, and I.-T. Tang, "Effect of seed layer on the growth of well-aligned ZnO nanowires," *Journal of Physics and Chemistry of Solids*, vol. 70, no. 10, pp. 1359–1362, 2009.
- [87] J.-H. Tian, J. Hu, S.-S. Li, F. Zhang, J. Liu, J. Shi, X. Li, Z.-Q. Tian, and Y. Chen, "Improved seedless hydrothermal synthesis of dense and ultralong ZnO nanowires," *Nanotechnology*, vol. 22, no. 24, p. 245601, 2011.

- [88] X. Li, Y.-H. Wang, A. Lu, and X. Liu, "Controllable Hydrothermal Growth of ZnO Nanowires on Cellulose Paper for Flexible Sensors and Electronics," *IEEE Sensors Journal*, vol. 15, pp. 6100–6107, Nov 2015.
- [89] O. Lupan and T. Pauport, "Hydrothermal treatment for the marked structural and optical quality improvement of ZnO nanowire arrays deposited on lightweight flexible substrates," *Journal of Crystal Growth*, vol. 312, no. 16-17, pp. 2454–2458, 2010.
- [90] Y. Lin, G. Ehlert, and H. A. Sodano, "Increased interface strength in carbon fiber composites through a ZnO nanowire interphase," *Advanced Functional Materials*, vol. 19, no. 16, pp. 2654–2660, 2009.
- [91] B. Xu and Z. Cai, "Fabrication of a superhydrophobic ZnO nanorod array film on cotton fabrics via a wet chemical route and hydrophobic modification," *Applied Surface Science*, vol. 254, no. 18, pp. 5899–5904, 2008.
- [92] W. I. Park, C.-H. Lee, J. H. Chae, D. H. Lee, and G.-C. Yi, "Ultrafine ZnO nanowire electronic device arrays fabricated by selective metal-organic chemical vapor deposition.," *Small*, vol. 5, pp. 181–4, Feb 2009.
- [93] Y. K. Park, H. S. Choi, J.-H. Kim, J.-H. Kim, and Y.-B. Hahn, "High performance field-effect transistors fabricated with laterally grown ZnO nanorods in solution.," *Nanotechnology*, vol. 22, p. 185310, May 2011.
- [94] P.-Y. Yang, J.-L. Wang, W.-C. Tsai, S.-J. Wang, J.-C. Lin, I.-C. Lee, C.-T. Chang, and H.-C. Cheng, "Photoresponse of hydrothermally grown lateral ZnO nanowires," *Thin Solid Films*, vol. 518, pp. 7328–7332, Oct 2010.
- [95] Y. Qin, R. Yang, and Z. L. Wang, "Growth of Horizonatal ZnO Nanowire Arrays on Any Substrate," *Journal of Physical Chemistry C*, vol. 112, pp. 18734–18736, nov 2008.
- [96] N. Liu, G. Fang, W. Zeng, H. Zhou, F. Cheng, Q. Zheng, L. Yuan, X. Zou, and X. Zhao, "Direct Growth of Lateral ZnO Nanorod UV Photodetectors with Schottky Contact by a Single-Step Hydrothermal Reaction," ACS Applied Materials & Interfaces, vol. 2, no. 7, pp. 1973–1979, 2010.
- [97] W.-C. Lin, Y.-C. Lin, M. Esashi, and A. Seshia, "In-situ hydrothermal synthesis of zinc oxide nanostructures using microheaters," *IEEE Transactions on Nanotechnology*, vol. 1, no. d, pp. 1–1, 2015.
- [98] P. Shi-Ming, S. Yan-Kuin, J. Liang-Wen, Y. Sheng-Joue, W. Cheng-Zhi, C. Wei-Bin, C. Wan-Chun, and T. Chi-Nan, "Photosensitivity of Field-Effect Transistors

Based on ZnO Nanowire Networks," *IEEE Electron Device Letters*, vol. 32, no. 3, pp. 339–341, 2011.

- [99] S. Xu, Y. Shen, Y. Ding, and Z. L. Wang, "Growth and Transfer of Monolithic Horizontal ZnO Nanowire Superstructures onto Flexible Substrates," *Advanced Functional Materials*, vol. 20, pp. 1493–1497, Apr 2010.
- [100] S. Sze and K. Ng, *Physics of Semiconductor Devices*. Wiley, 2006.
- [101] "Band bending of semiconductor." https://commons.wikimedia.org/ wiki/File:TFT_DIA_BIAS.png. [Online; accessed 16-July-2017].
- [102] A. Ortiz-Conde, F. G. Snchez, J. Liou, A. Cerdeira, M. Estrada, and Y. Yue, "A review of recent MOSFET threshold voltage extraction methods," *Microelectronics Reliability*, vol. 42, no. 4, pp. 583 596, 2002.
- [103] S.-S. Kwon, W.-K. Hong, G. Jo, J. Maeng, T.-W. Kim, S. Song, and T. Lee, "Piezoelectric Effect on the Electronic Transport Characteristics of ZnO Nanowire Field-Effect Transistors on Bent Flexible Substrates," *Advanced Materials*, vol. 20, pp. 4557–4562, dec 2008.
- [104] S. Jeong, M. Choe, J. W. Kang, M. W. Kim, W. G. Jung, Y. C. Leem, J. Chun, B. J. Kim, and S. J. Park, "High-performance photoconductivity and electrical transport of ZnO/ZnS core/shell nanowires for multifunctional nanodevice applications," ACS Applied Materials and Interfaces, vol. 6, no. 9, pp. 6170–6176, 2014.
- [105] J. I. Sohn, W.-K. Hong, M. J. Lee, T. Lee, H. Sirringhaus, D. J. Kang, and M. E. Welland, "The influence of surface chemical dynamics on electrical and optical properties of ZnO nanowire field effect transistors.," *Nanotechnology*, vol. 20, no. 50, p. 505202, 2009.
- [106] C. Opoku, A. S. Dahiya, F. Cayrel, G. Poulin-Vittrant, D. Alquier, and N. Camara, "Fabrication of field-effect transistors and functional nanogenerators using hydrothermally grown ZnO nanowires," *RSC Advances*, vol. 5, no. 86, pp. 69925– 69931, 2015.
- [107] D. Wang, Q. Wang, A. Javey, R. Tu, H. Dai, H. Kim, P. C. McIntyre, T. Krishnamohan, and K. C. Saraswat, "Germanium nanowire field-effect transistors with SiO₂ and high-k HfO₂ gate dielectrics," *Applied Physics Letters*, vol. 83, no. 12, pp. 2432–2434, 2003.
- [108] M. S. Arnold, P. Avouris, Z. W. Pan, and Z. L. Wang, "Field-Effect Transistors Based on Single Semiconducting Oxide Nanobelts," *Journal of Physical Chemistry B*, vol. 107, pp. 659–663, Jan 2003.

- [109] M. Law, L. E. Greene, J. C. Johnson, R. Saykally, and P. Yang, "Nanowire dyesensitized solar cells," *Nature Materials*, vol. 4, pp. 455–9, Jun 2005.
- [110] P. Jin Jeon, Y. Tack Lee, R. Ha, H.-J. Choi, K. Hyuck Yoon, M. M. Sung, and S. Im, "Annealing-induced conductivity transition in ZnO nanowires for fieldeffect devices," *Applied Physics Letters*, vol. 101, no. 4, p. 043504, 2012.
- [111] S. H. Ko, I. Park, H. Pan, N. Misra, M. S. Rogers, C. P. Grigoropoulos, and A. P. Pisano, "ZnO nanowire network transistor fabrication on a polymer substrate by low-temperature, all-inorganic nanoparticle solution process," *Applied Physics Letters*, vol. 92, no. 15, p. 154102, 2008.
- [112] V. Pachauri, A. Vlandas, K. Kern, and K. Balasubramanian, "Site-specific selfassembled liquid-gated ZnO nanowire transistors for sensing applications," *Small*, vol. 6, pp. 589–94, Feb 2010.
- [113] D. Kälblein, R. T. Weitz, H. J. Böttcher, F. Ante, U. Zschieschang, K. Kern, and H. Klauk, "Top-gate ZnO nanowire transistors and integrated circuits with ultrathin self-assembled monolayer gate dielectric.," *Nano letters*, vol. 11, pp. 5309–15, dec 2011.
- [114] J. Bang and K. J. Chang, "Diffusion and thermal stability of hydrogen in ZnO," *Applied Physics Letters*, vol. 92, no. 13, p. 132109, 2008.
- [115] P.-C. Chang, Z. Fan, C.-J. Chien, D. Stichtenoth, C. Ronning, and J. G. Lu, "High-performance ZnO nanowire field effect transistors," *Applied Physics Letters*, vol. 89, no. 13, p. 133113, 2006.
- [116] P.-C. Chang and J. G. Lu, "ZnO Nanowire Field-Effect Transistors," IEEE Transactions on Electron Devices, vol. 55, pp. 2977–2987, Nov 2008.
- [117] H. Klauk, U. Zschieschang, J. Pflaum, and M. Halik, "Ultralow-power organic complementary circuits.," *Nature*, vol. 445, pp. 745–8, Feb 2007.
- [118] D. Kälblein, H. Ryu, F. Ante, B. Fenk, K. Hahn, K. Kern, and H. Klauk, "Highperformance ZnO nanowire transistors with aluminum top-gate electrodes and naturally formed hybrid self-assembled monolayer/AlOx gate dielectric," ACS Nano, vol. 8, no. 7, pp. 6840–6848, 2014.
- [119] Y. Wang, X. W. Sun, J. Zhao, G. K. L. Goh, L. Chen, L.-L. Liew, J. Qiu, and Y.-H. Hwang, "Comparison of the hydrothermal and VPT grown ZnO nanowire field effect transistors," *International Journal of Nanoscience*, vol. 09, pp. 317–320, Aug 2010.

- [120] Y. Gao, M. Nagai, T.-C. Chang, and J.-J. Shyue, "Solution-Derived ZnO Nanowire Array Film as Photoelectrode in Dye-Sensitized Solar Cells," *Crystal Growth & Design*, vol. 7, no. 12, pp. 2467–2471, 2007.
- [121] R. Yang, Y. Qin, L. Dai, and Z. Wang, "Power generation with laterally packaged piezoelectric fine wires," *Nature Nanotechnology*, vol. 4, no. November 2008, pp. 34 – 39, 2008.
- [122] C. Florica, A. Costas, A. Kuncser, N. Preda, and I. Enculescu, "High performance FETs based on ZnO nanowires synthesized by low cost methods," *Nanotechnol*ogy, vol. 27, no. 47, p. 475303, 2016.
- [123] S.-W. Lee, M.-H. Ham, J. P. Kar, W. Lee, and J.-M. Myoung, "Selective alignment of a ZnO nanowire in a magnetic field for the fabrication of an air-gap field-effect transistor," *Microelectronic Engineering*, vol. 87, no. 1, pp. 10 – 14, 2010.
- [124] Y.-K. Chang and F. C.-N. Hong, "The fabrication of ZnO nanowire fieldeffect transistors combining dielectrophoresis and hot-pressing," *Nanotechnology*, vol. 20, no. 23, p. 235202, 2009.
- [125] D. Look, J. Hemsky, and J. Sizelove, "Residual Native Shallow Donor in ZnO," *Physical Review Letters*, vol. 82, pp. 2552–2555, Mar 1999.
- [126] A. Janotti and C. G. Van de Walle, "Oxygen vacancies in ZnO," *Applied Physics Letters*, vol. 87, no. 12, p. 122102, 2005.
- [127] A. Janotti and C. G. Van de Walle, "Hydrogen multicentre bonds," *Nature Materials*, vol. 6, pp. 44–7, Jan 2007.
- [128] M. Wardle, J. Goss, and P. Briddon, "First-Principles Study of the Diffusion of Hydrogen in ZnO," *Physical Review Letters*, vol. 96, p. 205504, May 2006.
- [129] Y.-S. Kim and C. Park, "Rich Variety of Defects in ZnO via an Attractive Interaction between O Vacancies and Zn Interstitials: Origin of n-Type Doping," *Physical Review Letters*, vol. 102, p. 086403, Feb 2009.
- [130] A. Janotti and C. G. Van de Walle, "Fundamentals of zinc oxide as a semiconductor," *Reports on Progress in Physics*, vol. 72, p. 126501, Dec 2009.
- [131] J. I. Sohn, S. S. Choi, S. M. Morris, J. S. Bendall, H. J. Coles, W.-K. Hong, G. Jo, T. Lee, and M. E. Welland, "Novel nonvolatile memory with multibit storage based on a ZnO nanowire transistor.," *Nano letters*, vol. 10, pp. 4316–20, Nov 2010.
- [132] Z. Fan, D. Wang, P.-C. Chang, W.-Y. Tseng, and J. G. Lu, "ZnO nanowire fieldeffect transistor and oxygen sensing property," *Applied Physics Letters*, vol. 85, no. 24, p. 5923, 2004.
- [133] W. Kim, A. Javey, O. Vermesh, Q. Wang, Y. Li, and H. Dai, "Hysteresis caused by water molecules in carbon nanotube field-effect transistors," *Nano Letters*, vol. 3, no. 2, pp. 193–198, 2003.
- [134] D. Wang, Y.-L. Chang, Q. Wang, J. Cao, D. B. Farmer, R. G. Gordon, and H. Dai, "Surface chemistry and electrical properties of germanium nanowires," *Journal* of the American Chemical Society, vol. 126, no. 37, pp. 11602–11611, 2004. PMID: 15366907.
- [135] H. Wang, Y. Wu, C. Cong, J. Shang, and T. Yu, "Hysteresis of electronic transport in graphene transistors," ACS Nano, vol. 4, no. 12, pp. 7221–7228, 2010. PMID: 21047068.
- [136] D. J. Late, B. Liu, H. S. S. R. Matte, V. P. Dravid, and C. N. R. Rao, "Hysteresis in Single-Layer MoS₂ Field Effect Transistors," ACS Nano, vol. 6, no. 6, pp. 5635– 5641, 2012. PMID: 22577885.
- [137] K. Keem, J. Kang, C. Yoon, D. Yeom, D.-Y. Jeong, B. Park, J. Park, and S. Kim, "ZnO Nanowire-Based Nonvolatile Memory Devices with Al₂O₃ Layers as Storage Nodes," *Journal of Nanoscience and Nanotechnology*, vol. 9, no. 7, 2009.
- [138] W.-K. Hong, J. I. Sohn, S. Cha, J. M. Kim, J.-B. Park, S. S. Choi, H. J. Coles, and M. E. Welland, "Programmable ZnO nanowire transistors using switchable polarization of ferroelectric liquid crystal," *Applied Physics Letters*, vol. 102, no. 5, p. 053504, 2013.
- [139] S. Nedic, Y. T. Chun, W.-K. Hong, D. Chu, and M. Welland, "High performance non-volatile ferroelectric copolymer memory based on a ZnO nanowire transistor fabricated on a transparent substrate," *Applied Physics Letters*, vol. 104, no. 3, p. 033101, 2014.
- [140] J. Yoon, W.-K. Hong, M. Jo, G. Jo, M. Choe, W. Park, J. I. Sohn, S. Nedic, H. Hwang, M. E. Welland, and T. Lee, "Nonvolatile Memory Functionality of ZnO Nanowire Transistors Controlled by Mobile Protons," ACS Nano, vol. 5, no. 1, pp. 558–564, 2011. PMID: 21155534.
- [141] C.-H. Lee, J. Yoo, Y.-J. Doh, and G.-C. Yi, "ZnO/Mg_{0.2}Zn_{0.8}O coaxial nanorod heterostructures for high-performance electronic nanodevice applications," *Applied Physics Letters*, vol. 94, no. 4, p. 043504, 2009.

- [142] J. Maeng, G. Jo, S.-S. Kwon, S. Song, J. Seo, S.-J. Kang, D.-Y. Kim, and T. Lee, "Effect of gate bias sweep rate on the electronic properties of ZnO nanowire fieldeffect transistors under different environments," *Applied Physics Letters*, vol. 92, no. 23, p. 233120, 2008.
- [143] T.-H. Moon, M.-C. Jeong, B.-Y. Oh, M.-H. Ham, M.-H. Jeun, W.-Y. Lee, and J.-M. Myoung, "Chemical surface passivation of HfO₂ films in a ZnO nanowire transistor," *Nanotechnology*, vol. 17, no. 9, p. 2116, 2006.
- [144] J. Nishii, A. Ohtomo, K. Ohtani, H. Ohno, and M. Kawasaki, "High-Mobility Field-Effect Transistors Based on Single-Crystalline ZnO Channels," *Japanese Journal of Applied Physics*, vol. 44, no. 9L, p. L1193, 2005.
- [145] J. Maeng, W. Park, M. Choe, G. Jo, Y. H. Kahng, and T. Lee, "Transient drain current characteristics of ZnO nanowire field effect transistors," *Applied Physics Letters*, vol. 95, no. 12, p. 123101, 2009.
- [146] S. Ju, D. B. Janes, G. Lu, A. Facchetti, and T. J. Marks, "Effects of bias stress on ZnO nanowire field-effect transistors fabricated with organic gate nanodielectrics," *Applied Physics Letters*, vol. 89, no. 19, p. 193506, 2006.
- [147] W.-K. Hong, G. Jo, J. I. Sohn, W. Park, M. Choe, G. Wang, Y. H. Kahng, M. E. Welland, and T. Lee, "Tuning of the Electronic Characteristics of ZnO Nanowire Field Effect Transistors by Proton Irradiation," ACS Nano, vol. 4, no. 2, pp. 811–818, 2010. PMID: 20112950.
- [148] S. J. Pearton, D. P. Norton, L. C. Tien, and J. Guo, "Modeling and Fabrication of ZnO Nanowire Transistors," *IEEE Transactions on Electron Devices*, vol. 55, pp. 3012–3019, Nov 2008.
- [149] "Agilent Technologies 4155C/4156C Semiconductor Parameter Analyzer User's Guide." http://eecs.oregonstate.edu/matdev/man/Agilent4155C_ 4156C_Users_Guide_Vol2.pdf. [Online; accessed 15-October-2017].
- [150] Y. F. Hsu, Y. Y. Xi, K. H. Tam, A. B. Djurišić, J. Luo, C. C. Ling, C. K. Cheung, A. M. C. Ng, W. K. Chan, X. Deng, C. D. Beling, S. Fung, K. W. Cheah, P. W. K. Fong, and C. C. Surya, "Undoped p-Type ZnO Nanorods Synthesized by a Hydrothermal Method," *Advanced Functional Materials*, vol. 18, pp. 1020–1030, Apr 2008.
- [151] B. Pradhan, S. K. Batabyal, and A. J. Pal, "Vertically aligned ZnO nanowire arrays in Rose Bengal-based dye-sensitized solar cells," *Solar Energy Materials and Solar Cells*, vol. 91, pp. 769–773, May 2007.

- [152] G. Zhu, R. Yang, S. Wang, and Z. L. Wang, "Flexible high-output nanogenerator based on lateral ZnO nanowire array.," *Nano letters*, vol. 10, pp. 3151–5, Aug 2010.
- [153] S. Y. Liu, T. Chen, J. Wan, G. P. Ru, B. Z. Li, and X. P. Qu, "The effect of preannealing of sputtered ZnO seed layers on growth of ZnO nanorods through a hydrothermal method," *Applied Physics A: Materials Science and Processing*, vol. 94, no. 4, pp. 775–780, 2009.
- [154] A. R. Kim, J.-Y. Lee, B. R. Jang, H. S. Kim, H. K. Park, Y. J. Cho, and N. W. Jang, "Effect of post annealing of zno buffer layer on the properties of hydrothermally grown zno nanorods," *Japanese Journal of Applied Physics*, vol. 49, no. 6S, p. 06GH10, 2010.
- [155] J. Song and S. Lim, "Effect of Seed Layer on the Growth of ZnO Nanorods," *Journal of Physical Chemistry C*, vol. 111, no. 2, pp. 596–600, 2007.
- [156] Y. Tao, M. Fu, A. Zhao, D. He, and Y. Wang, "The effect of seed layer on morphology of ZnO nanorod arrays grown by hydrothermal method," *Journal of Alloys and Compounds*, vol. 489, no. 1, pp. 99 – 102, 2010.
- [157] T. Sahoo, E. S. Kang, M. Kim, V. Kannan, Y. T. Yu, D. C. Shin, T. G. Kim, and I. H. Lee, "Growth of ZnO thin film on p-GaN/sapphire (0001) by simple hydrothermal technique," *Journal of Crystal Growth*, vol. 310, no. 3, pp. 570–574, 2008.
- [158] Q. Chen, Y. Qian, Z. Chen, G. Zhou, and Y. Zhang, "Hydrothermal preparation of highly oriented polycrystalline ZnO thin films," *Materials Letters*, vol. 22, no. 1-2, pp. 93–95, 1995.
- [159] K. Govender, D. S. Boyle, P. B. Kenway, and P. O'Brien, "Understanding the factors that govern the deposition and morphology of thin films of ZnO from aqueous solution," *J. Mater. Chem.*, vol. 14, no. 16, pp. 2575–2591, 2004.
- [160] J. H. Kim, E. M. Kim, D. Andeen, D. Thomson, S. P. DenBaars, and F. F. Lange, "Growth of heteroepitaxial ZnO thin films on GaN-buffered Al₂O₃ (0001) substrates by low-temperature hydrothermal synthesis at 90°C," *Advanced Functional Materials*, vol. 17, no. 3, pp. 463–471, 2007.
- [161] C. Y. Lee, M. Y. Lin, W. H. Wu, J. Y. Wang, Y. Chou, W. F. Su, Y. F. Chen, and C. F. Lin, "Flexible ZnO transparent thin-film transistors by a solution-based process at various solution concentrations," *Semiconductor Science and Technology*, vol. 25, no. 10, p. 105008, 2010.

- [162] C. Li, Y. Li, Y. Wu, B. S. Ong, and R. O. Loutfy, "ZnO field-effect transistors prepared by aqueous solution-growth ZnO crystal thin film," *Journal of Applied Physics*, vol. 102, no. 7, pp. 1–4, 2007.
- [163] H. Nishizawa and K. Yuasa, "Preparation of highly oriented ZnO thin film under hydrothermal conditions," *Journal of Materials Science Letters*, vol. 17, pp. 985–987, 1998.
- [164] K. Sun, W. Wei, Y. Ding, Y. Jing, Z. L. Wang, and D. Wang, "Crystalline ZnO thin film by hydrothermal growth," *Chemical Communications*, vol. 47, no. 27, p. 7776, 2011.
- [165] Z. Yuan, X. Jiaqiang, X. Qun, L. Hui, P. Qingyi, and X. Pengcheng, "Brush-like hierarchical ZnO nanostructures: synthesis, photoluminescence and gas sensor properties," *Journal of Physical Chemistry C*, vol. 113, no. 9, pp. 3430–3435, 2009.
- [166] C. Cheng, B. Liu, H. Yang, W. Zhou, L. Sun, R. Chen, S. F. Yu, J. Zhang, H. Gong, and H. Sun, "Hierarchical assembly of ZnO nanostructures on SnO₂ backbone nanowires: Low-temperature hydrothermal preparation and optical properties," *ACS Nano*, vol. 3, no. 10, pp. 3069–3076, 2009.
- [167] P. Jiang, J. J. Zhou, H. F. Fang, C. Y. Wang, Z. L. Wang, and S. S. Xie, "Hierarchical shelled ZnO structures made of bunched nanowire arrays," *Advanced Functional Materials*, vol. 17, no. 8, pp. 1303–1310, 2007.
- [168] J. Y. Lao, J. G. Wen, and Z. F. Ren, "Hierarchical ZnO Nanostructures," Nano Letters, vol. 2, no. 11, pp. 1287–1291, 2002.
- [169] J. Qiu, M. Guo, Y. Feng, and X. Wang, "Electrochemical deposition of branched hierarchical ZnO nanowire arrays and its photoelectrochemical properties," *Electrochimica Acta*, vol. 56, no. 16, pp. 5776–5782, 2011.
- [170] P. Sudhagar, T. Song, D. H. Lee, I. Mora-Seró, J. Bisquert, M. Laudenslager, W. M. Sigmund, W. I. Park, U. Paik, and Y. S. Kang, "High open circuit voltage quantum dot sensitized solar cells manufactured with ZnO nanowire arrays and Si/ZnO branched hierarchical structures," *Journal of Physical Chemistry Letters*, vol. 2, no. 16, pp. 1984–1990, 2011.
- [171] C.-T. Wu and J.-J. Wu, "Room-temperature synthesis of hierarchical nanostructures on ZnO nanowire anodes for dye-sensitized solar cells," *Journal of Materials Chemistry*, vol. 21, no. 35, p. 13605, 2011.

- [172] F. Xu, M. Dai, Y. Lu, and L. Sun, "Hierarchical ZnO Nanowire Nanosheet Architectures for High Power Conversion Efficiency in Dye-Sensitized Solar Cells," *Journal of Physical Chemistry*, vol. 114, pp. 2776–2782, 2010.
- [173] H. Yan, R. He, J. Johnson, M. Law, R. J. Saykally, and P. Yang, "Dendritic nanowire ultraviolet laser array," *Journal of the American Chemical Society*, vol. 125, no. 16, pp. 4728–4729, 2003.
- [174] S. Ko, D. Lee, H. Kang, K. Nam, and J. Yeo, "Nanoforest of Hydrothermally Grown Hierarchical ZnO Nanowires for a High Efficiency Dye-Sensitized Solar Cell," *Nano Letters*, vol. 11, no. 2, pp. 666–671, 2011.
- [175] A. Nathan, A. Ahnood, M. T. Cole, Y. Suzuki, P. Hiralal, F. Bonaccorso, T. Hasan, L. Garcia-Gancedo, A. Dyadyusha, S. Haque, P. Andrew, S. Hofmann, J. Moultrie, A. J. Flewitt, A. C. Ferrari, M. J. Kelly, J. Robertson, G. A. J. Amaratunga, and W. I. Milne, "Flexible Electronics: The Next Ubiquitous Platform," *Proceedings of the IEEE*, vol. 100, no. Special Centennial Issue, pp. 1486–1517, 2012.
- [176] L. Liao, H. Lu, J. Li, H. He, D. Wang, D. Fu, C. Liu, and W. Zhang, "Size Dependence of Gas Sensitivity of ZnO Nanorods," *Journal of Physical Chemistry C*, vol. 111, pp. 1900–1903, Feb 2007.
- [177] H.-W. Ra, R. Khan, J. Kim, B. Kang, K. Bai, and Y. Im, "Effects of surface modification of the individual ZnO nanowire with oxygen plasma treatment," *Materials Letters*, vol. 63, pp. 2516–2519, Nov 2009.
- [178] Q. H. Li, Y. X. Liang, Q. Wan, and T. H. Wang, "Oxygen sensing characteristics of individual ZnO nanowire transistors," *Applied Physics Letters*, vol. 85, no. 26, p. 6389, 2004.
- [179] A. M. Lord, T. G. Maffeis, A. S. Walton, D. M. Kepaptsoglou, Q. M. Ramasse, M. B. Ward, J. Köble, and S. P. Wilks, "Factors that determine and limit the resistivity of high-quality individual ZnO nanowires.," *Nanotechnology*, vol. 24, p. 435706, nov 2013.
- [180] J. Goldberger, D. J. Sirbuly, M. Law, and P. Yang, "ZnO nanowire transistors.," *Journal of Physical Chemistry B*, vol. 109, no. 1, pp. 9–14, 2005.
- [181] W.-K. Hong, G. Jo, S.-S. Kwon, S. Song, and T. Lee, "Electrical Properties of Surface-Tailored ZnO Nanowire Field-Effect Transistors," *IEEE Transactions on Electron Devices*, vol. 55, pp. 3020–3029, Nov 2008.

- [182] J. Bao, I. Shalish, Z. Su, R. Gurwitz, F. Capasso, X. Wang, and Z. Ren, "Photoinduced oxygen release and persistent photoconductivity in ZnO nanowires," *Nanoscale Research Letters*, vol. 6, no. 1, p. 404, 2011.
- [183] Z.-M. Liao, Y. Lu, J. Xu, J.-M. Zhang, and D.-P. Yu, "Temperature dependence of photoconductivity and persistent photoconductivity of single ZnO nanowires," *Applied Physics A*, vol. 95, no. 2, pp. 363–366, 2009.
- [184] Q. H. Li, T. Gao, Y. G. Wang, and T. H. Wang, "Adsorption and desorption of oxygen probed from ZnO nanowire films by photocurrent measurements," *Applied Physics Letters*, vol. 86, no. 12, pp. 1–3, 2005.
- [185] C. Soci, A. Zhang, B. Xiang, S. A. Dayeh, D. P. R. Aplin, J. Park, X. Y. Bao, Y. H. Lo, and D. Wang, "ZnO nanowire UV photodetectors with high internal gain," *Nano Letters*, vol. 7, no. 4, pp. 1003–1009, 2007.
- [186] M. E. Swanwick, S. M.-L. Pfaendler, A. I. Akinwande, and A. J. Flewitt, "Nearultraviolet zinc oxide nanowire sensor using low temperature hydrothermal growth.," *Nanotechnology*, vol. 23, p. 344009, Aug 2012.
- [187] S. Hullavarad, N. Hullavarad, D. Look, and B. Claflin, "Persistent Photoconductivity Studies in Nanostructured ZnO UV Sensors," *Nanoscale Research Letters*, vol. 4, pp. 1421–1427, Dec 2009.
- [188] W. Wu, G. Hu, S. Cui, Y. Zhou, and H. Wu, "Epitaxy of vertical ZnO nanorod arrays on highly (001)-oriented ZnO seed monolayer by a hydrothermal route," *Crystal Growth and Design*, no. 001, pp. 4014 – 4020, 2008.
- [189] Y. S. Bae, B. H. Kong, H. Cho, and J. H. Lee, "Ultralong ZnO Nanowire Arrays Synthesized by Hydrothermal Method Using Repetitive Refresh," *Journal* of Nanoscience and Nanotechnology, vol. 11, no. 8, pp. 7180–7184, 2011.
- [190] Z. R. Tian, J. A. Voigt, J. Liu, B. McKenzie, M. J. McDermott, M. A. Rodriguez, H. Konishi, and H. Xu, "Complex and oriented ZnO nanostructures," *Nature Materials*, vol. 2, pp. 821–6, Dec 2003.
- [191] Z. Fan and J. G. Lu, "Gate-refreshable nanowire chemical sensors," *Applied Physics Letters*, vol. 86, no. 12, p. 123510, 2005.
- [192] K. Yanagi, S. Kanda, Y. Oshima, Y. Kitamura, H. Kawai, T. Yamamoto, T. Takenobu, Y. Nakai, and Y. Maniwa, "Tuning of the thermoelectric properties of one-dimensional material networks by electric double layer techniques using ionic liquids," *Nano Letters*, vol. 14, no. 11, pp. 6437–6442, 2014. PMID: 25302572.

- [193] K. Besteman, J.-O. Lee, F. G. M. Wiertz, H. A. Heering, and C. Dekker, "Enzymecoated carbon nanotubes as single-molecule biosensors," *Nano Letters*, vol. 3, no. 6, pp. 727–730, 2003.
- [194] O. Knopfmacher, A. Tarasov, W. Fu, M. Wipf, B. Niesen, M. Calame, and C. Schnenberger, "Nernst limit in dual-gated si-nanowire fet sensors," *Nano Letters*, vol. 10, no. 6, pp. 2268–2274, 2010. PMID: 20499926.
- [195] H. Y. Zheng, O. A. Alsager, C. S. Wood, J. M. Hodgkiss, and N. O. V. Plank, "Carbon nanotube field effect transistor aptasensors for estrogen detection in liquids," *Journal of Vacuum Science & Technology B, Nanotechnology and Microelectronics: Materials, Processing, Measurement, and Phenomena*, vol. 33, no. 6, p. 06F904, 2015.
- [196] H. Y. Zheng, O. A. Alsager, B. Zhu, J. Travas-Sejdic, J. M. Hodgkiss, and N. O. V. Plank, "Electrostatic gating in carbon nanotube aptasensors," *Nanoscale*, vol. 8, pp. 13659–13668, 2016.
- [197] D. R. Kauffman and A. Star, "Electronically monitoring biological interactions with carbon nanotube field-effect transistors," *Chem. Soc. Rev.*, vol. 37, pp. 1197– 1206, 2008.
- [198] A. Menzel, K. Subannajui, F. Gder, D. Moser, O. Paul, and M. Zacharias, "Multifunctional ZnO-Nanowire-Based Sensor," *Advanced Functional Materials*, vol. 21, no. 22, pp. 4342–4348, 2011.
- [199] H. Yuan, H. Shimotani, A. Tsukazaki, A. Ohtomo, M. Kawasaki, and Y. Iwasa, "High-Density Carrier Accumulation in ZnO Field-Effect Transistors Gated by Electric Double Layers of Ionic Liquids," *Advanced Functional Materials*, vol. 19, no. 7, pp. 1046–1053, 2009.
- [200] S. Thiemann, S. Sachnov, S. Porscha, P. Wasserscheid, and J. Zaumseil, "Ionic Liquids for Electrolyte-Gating of ZnO Field-Effect Transistors," *Journal of Physical Chemistry C*, vol. 116, no. 25, pp. 13536–13544, 2012.
- [201] E. Schlenker, A. Bakin, T. Weimann, P. Hinze, D. H. Weber, A. Glzhuser, H.-H. Wehmann, and A. Waag, "On the difficulties in characterizing ZnO nanowires," *Nanotechnology*, vol. 19, no. 36, p. 365707, 2008.
- [202] S. Sze and K. K. Ng, *Metal-Semiconductor Contacts*, pp. 134–196. John Wiley & Sons, Inc., 2006.

- [203] T. P. Chen, S. J. Young, S. J. Chang, C. H. Hsiao, L. W. Ji, Y. J. Hsu, and S. L. Wu, "Low-frequency noise characteristics of zno nanorods schottky barrier photodetectors," *IEEE Sensors Journal*, vol. 13, pp. 2115–2119, June 2013.
- [204] J. He, C. Hsin, J. Liu, L. Chen, and Z. Wang, "Piezoelectric gated diode of a single zno nanowire," *Advanced Materials*, vol. 19, no. 6, pp. 781–784, 2007.
- [205] F. Donatini, A. de Luna Bugallo, P. Tchoulfian, G. Chicot, C. Sartel, V. Sallet, and J. Pernot, "Comparison of three e-beam techniques for electric field imaging and carrier diffusion length measurement on the same nanowires," *Nano Letters*, vol. 16, no. 5, pp. 2938–2944, 2016. PMID: 27105083.
- [206] S. R. Ali Raza, S. H. Hosseini Shokouh, Y. T. Lee, R. Ha, H.-J. Choi, and S. Im, "Niox schottky-gated zno nanowire metal-semiconductor field effect transistor: fast logic inverter and photo-detector," *J. Mater. Chem. C*, vol. 2, pp. 4428–4435, 2014.
- [207] S. H. Hosseini Shokouh, S. R. A. Raza, H. S. Lee, and S. Im, "Non-classical logic inverter coupling a zno nanowire-based schottky barrier transistor and adjacent schottky diode," *Phys. Chem. Chem. Phys.*, vol. 16, pp. 16367–16372, 2014.
- [208] C. Florica, E. Matei, A. Costas, M. E. T. Molares, and I. Enculescu, "Field effect transistor with electrodeposited zno nanowire channel," *Electrochimica Acta*, vol. 137, pp. 290 – 297, 2014.
- [209] K. Kim, H. Kang, H. Kim, J. S. Lee, S. Kim, W. Kang, and G.-T. Kim, "Contact barriers in a single zno nanowire device," *Applied Physics A*, vol. 94, pp. 253–256, Feb 2009.
- [210] Z.-M. Liao, K.-J. Liu, J.-M. Zhang, J. Xu, and D.-P. Yu, "Effect of surface states on electron transport in individual zno nanowires," *Physics Letters A*, vol. 367, no. 3, pp. 207 – 210, 2007.
- [211] Z.-M. Liao, C. Hou, Y.-B. Zhou, J. Xu, J.-M. Zhang, and D.-P. Yu, "Influence of temperature and illumination on surface barrier of individual zno nanowires," *The Journal of Chemical Physics*, vol. 130, no. 8, p. 084708, 2009.
- [212] C. Park, J. Lee, H.-M. So, and W. S. Chang, "An ultrafast response grating structural zno photodetector with back-to-back schottky barriers produced by hydrothermal growth," J. Mater. Chem. C, vol. 3, pp. 2737–2743, 2015.
- [213] Y. Shao, J. Yoon, H. Kim, T. Lee, and W. Lu, "Temperature dependence of electron transport in zno nanowire field effect transistors," *IEEE Transactions on Electron Devices*, vol. 61, pp. 625–630, Feb 2014.

- [214] R. Zhang, W. Pang, Q. Zhang, Y. Chen, X. Chen, Z. Feng, J. Yang, and D. Zhang, "Enhanced non-volatile resistive switching in suspended single-crystalline zno nanowire with controllable multiple states," *Nanotechnology*, vol. 27, no. 31, p. 315203, 2016.
- [215] R. Zhu and R. Yang, "Separation of the piezotronic and piezoresistive effects in a zinc oxide nanowire," *Nanotechnology*, vol. 25, no. 34, p. 345702, 2014.
- [216] Y.-F. Lin and W.-B. Jian, "The impact of nanocontact on nanowire based nanoelectronics," *Nano Letters*, vol. 8, no. 10, pp. 3146–3150, 2008. PMID: 18729520.
- [217] B. Bercu, W. Geng, O. Simonetti, S. Kostcheev, C. Sartel, V. Sallet, G. Lrondel, M. Molinari, L. Giraudet, and C. Couteau, "Characterizations of ohmic and schottky-behaving contacts of a single zno nanowire," *Nanotechnology*, vol. 24, no. 41, p. 415202, 2013.
- [218] A. J. Chiquito, C. A. Amorim, O. M. Berengue, L. S. Araujo, E. P. Bernardo, and E. R. Leite, "Back-to-back Schottky diodes: the generalization of the diode theory in analysis and extraction of electrical parameters of nanodevices," *Journal of Physics: Condensed Matter*, vol. 24, no. 22, p. 225303, 2012.
- [219] S. Sinha, S. K. Chatterjee, J. Ghosh, and A. K. Meikap, "Electrical transport properties of polyvinyl alcohol–selenium nanocomposite films at and above room temperature," *Journal of Materials Science*, vol. 50, pp. 1632–1645, Feb 2015.
- [220] S. Sinha, S. K. Chatterjee, J. Ghosh, and A. K. Meikap, "Semiconducting selenium nanoparticles: Structural, electrical characterization, and formation of a back-toback Schottky diode device," *Journal of Applied Physics*, vol. 113, no. 12, p. 123704, 2013.
- [221] R. A. Simon, H. Kamimura, O. M. Berengue, E. R. Leite, and A. J. Chiquito, "Disorder induced interface states and their influence on the Al/Ge nanowires Schottky devices," *Journal of Applied Physics*, vol. 114, no. 24, p. 243705, 2013.
- [222] S. Qiao, Q. Xu, R. K. Dutta, M. Le Thai, X. Li, and R. M. Penner, "Electrodeposited, transverse nanowire electroluminescent junctions," ACS Nano, vol. 10, no. 9, pp. 8233–8242, 2016. PMID: 27564397.
- [223] H. Kasani, R. Khodabakhsh, M. Taghi Ahmadi, D. Rezaei Ochbelagh, and R. Ismail, "Electrical Properties of MWCNT/HDPE Composite-Based MSM Structure Under Neutron Irradiation," *Journal of Electronic Materials*, vol. 46, pp. 2548– 2555, Apr 2017.

- [224] J. Osvald, "Back-to-back connected asymmetric Schottky diodes with series resistance as a single diode," *Physica Status Solidi* (a), vol. 212, no. 12, pp. 2754–2758, 2015.
- [225] "Schottky barrier at zero bias." https://commons.wikimedia.org/w/ index.php?title=File:Schottky_barrier_zero_bias.svg&oldid= 218923524.
- [226] K. Storm, G. Nylund, L. Samuelson, and A. P. Micolich, "Realizing Lateral Wrap-Gated Nanowire FETs: Controlling Gate Length with Chemistry Rather than Lithography," *Nano Letters*, vol. 12, no. 1, pp. 1–6, 2012. PMID: 21322605.
- [227] J. Robertson, "High dielectric constant oxides," *The European Physical Journal -Applied Physics*, vol. 28, no. 3, p. 265291, 2004.
- [228] W. Park, W.-K. Hong, G. Jo, G. Wang, M. Choe, J. Maeng, Y. H. Kahng, and T. Lee, "Tuning of operation mode of ZnO nanowire field effect transistors by solventdriven surface treatment.," *Nanotechnology*, vol. 20, p. 475702, Nov 2009.
- [229] W. Park, G. Jo, W.-K. Hong, J. Yoon, M. Choe, S. Lee, Y. Ji, G. Kim, Y. H. Kahng, K. Lee, D. Wang, and T. Lee, "Enhancement in the photodetection of ZnO nanowires by introducing surface-roughness-induced traps.," *Nanotechnology*, vol. 22, no. 20, p. 205204, 2011.
- [230] W. Kim and K. S. Chu, "ZnO nanowire field-effect transistor as a UV photodetector; optimization for maximum sensitivity," *Physica Status Solidi (A) Applications and Materials Science*, vol. 206, no. 1, pp. 179–182, 2009.
- [231] M.-Y. Lu, M.-P. Lu, S.-J. You, C.-W. Chen, and Y.-J. Wang, "Quantifying the barrier lowering of ZnO Schottky nanodevices under UV light," vol. 5, pp. 15123 EP –, Oct 2015. Article.
- [232] J.-M. Lee, K.-K. Kim, S.-J. Park, and W.-K. Choi, "Low-resistance and nonalloyed ohmic contacts to plasma treated ZnO," *Applied Physics Letters*, vol. 78, no. 24, pp. 3842–3844, 2001.
- [233] H.-W. Ra, K. S. Choi, C. W. Ok, S. Y. Jo, K. H. Bai, and Y. H. Im, "Ion bombardment effects on ZnO nanowires during plasma treatment," *Applied Physics Letters*, vol. 93, no. 3, p. 033112, 2008.
- [234] X. Li, J. Qi, Q. Zhang, Z. Wang, S. Lu, and Y. Zhang, "Investigation of electron beam detection properties of ZnO nanowire based back-to-back double Schottky diode," RSC Adv., vol. 4, pp. 12743–12747, 2014.

- [235] F. Donatini, C. Sartel, V. Sallet, and J. Pernot, "Electron beam dose dependence of surface recombination velocity and surface space charge in semiconductor nanowires," *Nanotechnology*, vol. 28, no. 23, p. 235701, 2017.
- [236] J. Zhou, N. Xu, and Z. Wang, "Dissolving Behavior and Stability of ZnO Wires in Biofluids: A Study on Biodegradability and Biocompatibility of ZnO Nanostructures," *Advanced Materials*, vol. 18, pp. 2432–2435, Sep 2006.
- [237] X. Xu, C. Song, J. M. Andresen, B. G. Miller, and A. W. Scaroni, "Novel Polyethylenimine-Modified Mesoporous Molecular Sieve of MCM-41 Type as High-Capacity Adsorbent for CO₂ Capture," *Energy & Fuels*, vol. 16, no. 6, pp. 1463–1469, 2002.
- [238] P. Li, B. Ge, S. Zhang, S. Chen, Q. Zhang, and Y. Zhao, "CO₂ Capture by Polyethylenimine-Modified Fibrous Adsorbent," *Langmuir*, vol. 24, no. 13, pp. 6567–6574, 2008. PMID: 18507414.
- [239] X. Xu, C. Song, B. G. Miller, and A. W. Scaroni, "Influence of Moisture on CO₂ Separation from Gas Mixture by a Nanoporous Adsorbent Based on Polyethylenimine-Modified Molecular Sieve MCM-41," *Industrial & Engineering Chemistry Research*, vol. 44, no. 21, pp. 8113–8119, 2005.
- [240] Z. Chen, S. Deng, H. Wei, B. Wang, J. Huang, and G. Yu, "Polyethylenimine-Impregnated Resin for High CO2 Adsorption: An Efficient Adsorbent for CO2 Capture from Simulated Flue Gas and Ambient Air," ACS Applied Materials & Interfaces, vol. 5, no. 15, pp. 6937–6945, 2013. PMID: 23808655.
- [241] A. Heydari-Gorji and A. Sayari, "Thermal, Oxidative, and CO₂-Induced Degradation of Supported Polyethylenimine Adsorbents," *Industrial & Engineering Chemistry Research*, vol. 51, no. 19, pp. 6887–6894, 2012.
- [242] J.-N. Shen, C.-C. Yu, G.-N. Zeng, and B. van der Bruggen, "Preparation of a Facilitated Transport Membrane Composed of Carboxymethyl Chitosan and Polyethylenimine for CO₂/N₂ Separation," *International Journal of Molecular Sciences*, vol. 14, no. 2, pp. 3621–3638, 2013.
- [243] T. Zhu, S. Yang, D. K. Choi, and K. H. Row, "Adsorption of carbon dioxide using polyethyleneimine modified silica gel," *Korean Journal of Chemical Engineering*, vol. 27, pp. 1910–1915, Nov 2010.
- [244] P. Kumar, S. Kim, J. Ida, and V. V. Guliants, "Polyethyleneimine-Modified MCM-48 Membranes: Effect of Water Vapor and Feed Concentration on N₂/CO₂ Selectivity," *Industrial & Engineering Chemistry Research*, vol. 47, no. 1, pp. 201–208, 2008.

- [245] S. Zhu, X. Chen, F. Zuo, M. Jiang, Z. Zhou, and D. Hui, "Controllable synthesis of ZnO nanograss with different morphologies and enhanced performance in dye-sensitized solar cells," *Journal of Solid State Chemistry*, vol. 197, pp. 69 – 74, 2013.
- [246] M. R. Alenezi, S. J. Henley, and S. R. P. Silva, "On-chip Fabrication of High Performance Nanostructured ZnO UV Detectors," *Scientific Reports*.
- [247] R. Gurwitz, R. Cohen, and I. Shalish, "Interaction of light with the ZnO surface: Photon induced oxygen breathing, oxygen vacancies, persistent photoconductivity, and persistent photovoltage," *Journal of Applied Physics*, vol. 115, no. 3, p. 033701, 2014.
- [248] X. Xia, J. Strunk, W. Busser, R. Naumann dAlnoncourt, and M. Muhler, "Probing the Surface Heterogeneity of Polycrystalline Zinc Oxide by Static Adsorption Microcalorimetry. 1. The Influence of the Thermal Pretreatment on the Adsorption of Carbon Dioxide," *Journal of Physical Chemistry C*, vol. 112, no. 29, pp. 10938– 10942, 2008.
- [249] S. K. Youn, N. Ramgir, C. Wang, K. Subannajui, V. Cimalla, and M. Zacharias, "Catalyst-free growth of zno nanowires based on topographical confinement and preferential chemisorption and their use for room temperature co detection," *Journal of Physical Chemistry C*, vol. 114, no. 22, pp. 10092–10100, 2010.
- [250] N. H. Moreira, A. L. da Rosa, and T. Frauenheim, "Covalent functionalization of ZnO surfaces: A density functional tight binding study," *Applied Physics Letters*, vol. 94, no. 19, p. 193109, 2009.
- [251] Z. Zhang, K. Yao, Y. Liu, C. Jin, X. Liang, Q. Chen, and L.-M. Peng, "Quantitative analysis of currentvoltage characteristics of semiconducting nanowires: Decoupling of contact effects," *Advanced Functional Materials*, vol. 17, no. 14, pp. 2478– 2489, 2007.
- [252] R. Kozhummal, Y. Yang, F. Gder, A. Hartel, X. Lu, U. M. Kkbayrak, A. Mateo-Alonso, M. Elwenspoek, and M. Zacharias, "Homoepitaxial Branching: An Unusual Polymorph of Zinc Oxide Derived from Seeded Solution Growth," ACS Nano, vol. 6, no. 8, pp. 7133–7141, 2012. PMID: 22849328.
- [253] T.-H. Hsieh, J.-Y. Chen, C.-W. Huang, and W.-W. Wu, "Observing Growth of Nanostructured ZnO in Liquid," *Chemistry of Materials*, vol. 28, no. 12, pp. 4507– 4511, 2016.
- [254] F. Patolsky, G. Zheng, and C. M. Lieber, "Nanowire sensors for medicine and the life sciences," *Nanomedicine*, vol. 1, no. 1, pp. 51–65, 2006. PMID: 17716209.

- [255] A. Wanekaya, W. Chen, N. Myung, and A. Mulchandani, "Nanowire-based electrochemical biosensors," *Electroanalysis*, vol. 18, no. 6, pp. 533–550, 2006.
- [256] Y. Zhang, K. Yu, D. Jiang, Z. Zhu, H. Geng, and L. Luo, "Zinc oxide nanorod and nanowire for humidity sensor," *Applied Surface Science*, vol. 242, no. 1-2, pp. 212– 217, 2005.
- [257] X. Liu, P. Lin, X. Yan, Z. Kang, Y. Zhao, Y. Lei, C. Li, H. Du, and Y. Zhang, "Enzyme-coated single zno nanowire fet biosensor for detection of uric acid," *Sensors and Actuators B: Chemical*, vol. 176, pp. 22 – 27, 2013.
- [258] A. Choi, K. Kim, H.-I. Jung, and S. Y. Lee, "ZnO nanowire biosensors for detection of biomolecular interactions in enhancement mode," *Sensors and Actuators B: Chemical*, vol. 148, no. 2, pp. 577 – 582, 2010.